UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 8-K	

CURRENT REPORT

Pursuant to Section 13 or 15(d) of The Securities Exchange Act of 1934

Date of report (Date of earliest event reported): November 3, 2009

MASTEC, INC.

(Exact Name of Registrant as Specified in Its Charter)

Florida

(State or Other Jurisdiction of Incorporation)

Florida (State or other jurisdiction of incorporation) 0-08106 (Commission File Number) 65-0829355 (IRS Employer Identification No.)

800 S. Douglas Road, 12th Floor, Coral Gables, Florida 33134 (Address of Principal Executive Offices) (Zip Code)

> (305) 599-1800 (Registrant's Telephone Number, Including Area Code)

> > N/A

(Former Name or Former Address, if Changed Since Last Report)

follo	Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions:			
	Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)			
	Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)			
	Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))			
П	Pre-commencement communications pursuant to Rule 13e- $4(c)$ under the Exchange Act (17 CFR 240 13e- $4(c)$)			

Item 1.01 Entry into a Material Definitive Agreement.

On November 3, 2009, MasTec, Inc., a Florida corporation ("MasTec", "we", "us" or "our"); Precision Acquisition, LLC, a Wisconsin limited liability company and wholly-owned subsidiary of MasTec ("Buyer"); Precision Pipeline LLC, a Wisconsin limited liability company ("Precision Pipeline"); Precision Transport Company, LLC, a Wisconsin limited liability company ("Transport" and, together with Precision Pipeline, "Precision" or the "Companies"); PPL Management, Inc., a Wisconsin corporation ("PPL"); Michael Daniel Murphy ("Murphy"); Steven R. Rooney ("Rooney" and, together with PPL and Murphy, "Sellers"); Angela D. Murphy and Karen K. Rooney entered into an interest purchase agreement (the "Purchase Agreement"), pursuant to which Buyer has agreed to purchase all of the issued and outstanding membership interests in the Companies (including all transactions contemplated by the Purchase Agreement and related ancillary agreements, the "Transaction") for an aggregate purchase price, subject to adjustment, composed of (i) \$150 million in cash (the "Cash Consideration") and (ii) a five-year earn-out (the "Earn-Out") equal to 40% of Precision's EBITDA (as defined in the Purchase Agreement) for the last two months of 2009 and 30% of Precision's annualized EBITDA in excess of \$35 million for the remainder of the Earn-Out period, payable, at MasTec's option, in cash or, under certain circumstances, shares of MasTec's common stock, par value \$0.10 per share, or a combination thereof. MasTec will assume indebtedness of the Companies up to a maximum of \$20 million to the extent that the Companies have property, plant and equipment with a net book value in excess of \$44 million. Any additional indebtedness will reduce the Cash Consideration dollar for dollar. Cash Consideration equal to \$15 million will be held for a period of 18 months following consummation of the Transaction pursuant to the terms of an escrow agreement to fund the Sellers' potential indemnification obligations under the Purchase Agreement.

MasTec's obligation to close the Transaction is conditioned on its obtaining financing, on such terms as may be acceptable to MasTec in its sole and absolute discretion, that provides net proceeds to MasTec of not less than \$75 million (the "Financing"). Consummation of the Transaction is subject to certain additional conditions, including, among others, those relating to the accuracy of the parties' representations and warranties, delivery of certain ancillary agreements, including certain noncompetition and employment agreements, the termination or expiration of the applicable waiting period under the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended, and other customary closing conditions. Either the Buyer or the Sellers may terminate the Purchase Agreement if the Transaction has not closed on or prior to December 16, 2009. No assurance can be given that MasTec will be able to obtain the Financing or that the Buyer will otherwise be able to close the Transaction.

The foregoing description of the Purchase Agreement is only a summary and is qualified in its entirety by reference to the full text of the Purchase Agreement, which is filed as Exhibit 10.1 to this Current Report on Form 8-K and is hereby incorporated herein by reference.

In connection with the Transaction, on November 3, 2009, MasTec entered into an amendment (the "Letter Amendment") to its Second Amended and Restated Loan and Security Agreement, dated July 29, 2008, as amended (the "Credit Facility"), with Bank of America, N.A., as administrative agent for the lenders thereunder, pursuant to which the lenders thereunder consented to the Transaction, subject to the condition that MasTec consummate the Financing. Additionally, interest on outstanding balances on the Credit Facility accrue at rates based, at MasTec's option, on the agent bank's base rate plus a margin of between 1.25% and 1.75%, or at the LIBOR rate plus a margin of between 2.25% and 3.00%, depending on certain financial thresholds.

Item 7.01 Regulation FD Disclosure.

On November 3, 2009, MasTec issued a press release regarding the Transaction. A copy of that press release is furnished as Exhibit 99.1 to this Current Report on Form 8-K.

Included below is additional information relating to the Transaction and a description of Precision's business:

Business of Precision

Based in Eau Claire, Wisconsin and founded in 2004, Precision is a leading natural gas, crude oil and refined products transmission pipeline infrastructure contractor. Precision focuses on large diameter transcontinental projects and related services, such as mainline cross country pipelines, pipeline integrity projects, looping and laterals, hydrostatic testing, pipeline drying services and

meter and pump station construction. Precision had revenue of \$303 million and EBITDA of \$37 million for the year ended December 31, 2007 and revenue of \$507 million and EBITDA of \$93 million for the year ended December 31, 2008. Precision had estimated revenue of \$83 million and EBITDA of \$26 million for the eight months ended August 31, 2009. For the full year of 2009, Precision estimates revenue of slightly less than \$300 million and EBITDA of approximately \$60 million. As of September 30, 2009, Precision had over \$500 million in backlog. Due to the long-term nature of Precision's contracts and other factors that may affect Precision's projects, not all contractual revenues included in this backlog estimate are expected to be realized by the year ending December 31, 2010.

We believe that Precision's business will complement our other existing natural gas services that include constructing gathering systems, compressor stations and processing plants and intra-state mid-stream pipelines. Our existing natural gas infrastructure expertise and capability has been developed both organically and through acquisitions since 2004, and we believe that our acquisition of Precision will enhance our end-to-end design and construction capabilities within the energy industry. As a result of the Precision acquisition, we will become one of the leading energy pipeline contractors in the country, capable of providing a full array of construction services to oil and gas producers, as well as mid-stream and interstate pipeline operators.

We believe that our acquisition of Precision will provide us with a number of strategic benefits, including the following:

- strengthening our technical expertise and footprint in the energy pipeline construction market;
- completing our footprint in the natural gas, crude oil and refined products transportation construction market so that we will have end-to-end
 construction capabilities from "wellhead to burner tip";
- · diversifying our business mix, end markets and customer base and thereby reducing customer concentration risk;
- providing strong opportunities for cross-selling multiple services throughout the natural gas, crude oil and refined petroleum products markets; and
- · providing additional depth to our existing management team through the addition of the Precision senior management team.

Precision employs a team of highly-skilled unionized workers and tradesmen whom it deploys throughout North America. The size of the workforce expands and contracts based on active projects, with Precision currently employing approximately 1,600 personnel. Precision also can support three to four spreads of specialized pipeline and related construction equipment.

Since entering the pipeline construction market in 2004, Precision has constructed or rehabilitated approximately 1,200 miles of pipeline. We believe this expertise puts Precision in a position to capitalize on what we expect to be a growing natural gas, crude oil and refined products pipeline construction market.

Precision's current and prior customers include El Paso Energy, Millennium Pipeline, Enbridge Energy, Kinder Morgan and Fortuna Energy. Precision generally obtains its contracts through a competitive bidding process, and it has been increasingly successful at gaining market share from large pipeline developers, growing revenue from \$50 million in 2006 to over \$500 million in 2008. Precision has also experienced relatively few loss contracts over its short history and has received incentive payments on several jobs for superior performance. Some of Precision's customer contracts may also include "neutral-funding" provisions, which allow Precision to collect payment at the time work is being performed.

We believe that U.S. energy policy goals will continue to promote domestic sources of energy in order to reduce U.S. dependence on foreign energy sources. With recent developments in drilling and completion technologies for oil and gas, we expect new North American producing fields to be developed and old fields to be expanded significantly. We anticipate that the resulting incremental production will provide continuing construction opportunities as oil and gas producers and pipeline operators move this oil, gas and refined products to markets via pipelines. With oil prices close to \$80 per barrel and improving natural gas futures, we

believe that Precision's targeted end-markets will continue to grow and provide us with robust pipeline and related construction opportunities. As evidence of this expected growth, as of October 15, 2009, the Federal Energy Regulatory Commission had over 5,000 miles of pending major pipeline projects on file.

With the acquisition of Precision, combined with our historical gas gathering system business and the recent acquisitions of Pumpco, Inc. (specializing in mid-stream natural gas pipelines) and Wanzek Construction, Inc., (specializing in wind, solar and natural gas compression and industrial plant construction), we expect to be a leading player in both renewable energy and gas, crude oil and crude oil products transmission pipeline infrastructure.

Financial Data of Precision

The audited consolidated financial statements filed as Exhibit 99.2 to this Current Report on Form 8-K are those of PPL, of which Precision Pipeline is a 98% owned subsidiary, and its related entities, Transport and Precision Land Company, LLC ("<u>LandCo</u>"), as of and for the years ended December 31, 2008, 2007 and 2006. We are not acquiring PPL or LandCo, which is further discussed in Note 11 to the consolidated financial statements, or their assets or operations. PPL's operations are substantially limited to employing certain employees who manage Precision. PPL will transfer these employees to Precision as part of the Transaction.

We have not yet filed pro forma financial statements that conform with Rule 11-01(b) of Regulation S-X. However, in the future preparation of pro forma financial statements, we expect to make the following balance sheet and income statement adjustments. We cannot assure you that these are the only pro forma adjustments that we will make when we prepare our pro forma financial statements in accordance with Rule 11-01 of Regulation S-X or that we will not make pro forma adjustments that differ from our expectations set forth below.

Balance Sheet Adjustments:

- recording the payment to the Sellers of the Cash Consideration for Precision with the proceeds of the Financing;
- recording goodwill and intangibles, including the fair value of liabilities for contingent consideration as required under Statement of Financial Accounting Standards No. 141(R). Goodwill and intangibles are generally determined, in total, as the purchase price plus the fair value of contingent consideration (earn-outs) less the fair value of the tangible assets acquired and liabilities assumed. A portion of goodwill and intangibles will be amortized into expense;
- · recording estimated Financing costs;
- recording the exclusion of the LandCo assets and liabilities, which are included in the consolidated financial statements but will not be acquired in connection with our acquisition of Precision. Assets and liabilities include land and buildings of \$2.0 million and long-term debt of \$1.2 million;
- · transfer of employees, cash and payroll related liabilities from PPL, which we are not acquiring; and
- adjusting working capital to \$9 million, if below, with an offset to cash (Precision's sellers have agreed to pay us the amount by which net working capital of Precision is less than \$9 million; conversely, we will pay them the excess of net working capital over \$9 million).

Income Statement Adjustments:

• recording amortization of intangibles and deferred financing costs from the beginning of the pro-forma periods. Amortization of intangibles represents an amount equal to the sum of intangible assets recorded on the balance sheet as a result of the acquisition of Precision, not including goodwill, amortized over a useful life generally determined to be between five and twenty years;

- · recording additional depreciation expense required as a result of purchase accounting changes in value or depreciable life;
- recording interest expense on the Financing;
- · eliminating depreciation expense of \$129,000 and interest expense of \$94,900 related to LandCo, which we will not acquire;
- recording rent expense on the property held by LandCo of \$325,000, which was eliminated in Precision's consolidation. MasTec as the owner of Precision will become LandCo's tenant; and
- adjust the tax provision, if necessary, considering MasTec's tax position and net operating losses versus Precision's original tax position.

Reconciliation of Non-GAAP Disclosures - Unaudited

The following table, which may contain slight summation differences due to rounding, reconciles Precision's EBITDA, a non-GAAP financial measure, to net income, the most comparable GAAP financial measure.

	2009E	YTD 09 ⁽¹⁾	2008	2007
		(in milli	ions)	
Net Income	\$43.9	\$ 17.5	\$83.3	\$32.2
Depreciation & Amortization	10.4	7.5	8.7	4.3
Interest Expense, net	1.6	1.1	8.0	0.7
Tax	4.1	0.0	0.0	0.0
EBITDA	\$60.0	\$ 26.1	\$92.9	\$37.1

(1) For the eight months ended August 31, 2009

The information contained in this Item 7.01 of this Current Report on Form 8-K, including Exhibit 99.1, shall not be deemed "filed" with the Securities and Exchange Commission nor incorporated by reference in any registration statement filed by MasTec under the Securities Act.

Item 8.01 Other Events.

The audited consolidated balance sheet of PPL Management, Inc. and subsidiaries as of December 31, 2008 and 2007, and the audited consolidated statements of operations, equity and cash flows for the years ended December 31, 2008, 2007 and 2006 are filed as Exhibit 99.2 to this Current Report on Form 8-K and are hereby incorporated herein by reference.

Set forth below are certain risks relating to Precision and the Transaction:

Risks Related to Precision's Industry and Precision's Customers' Industries

As a specialty contractor operating in the United States, Precision is generally subject to the same risks that apply to MasTec's other businesses and which are described in our Annual Report on Form 10-K for the year ended December 31, 2008, as updated by our Quarterly Report on Form 10-Q for the quarter ended September 30, 2009. In addition, the following risk factors set forth certain other risks that are specific to Precision.

Precision derives a significant portion of its revenue from a few customers, and the loss of any one of these customers or a reduction in any such customer's demand for Precision's services could impair its financial performance.

For the eight months ended August 31, 2009 and the year ended December 31, 2008, Precision derived approximately 77.8% and 91.2%, respectively, of its revenue from its top two customers. Because Precision's business is concentrated among relatively few major customers, and all of its services are provided on a non-recurring, project-by-project basis, Precision could experience a reduction in its results of operations, cash flows and liquidity if:

- any of Precision's customers cancels, reduces the work under or suspend their contracts; or
- Precision completes the required work under its projects and cannot replace them with similar projects.

Additionally, because Precision's customer contracts may be canceled on short notice even if Precision is not in default under the contracts, its revenues are not guaranteed.

Precision's failure to properly manage projects, or project delays, may result in additional costs or claims, which could have a material adverse effect on its operating results, cash flows and liquidity.

Precision's engagements generally involve large-scale, complex projects. The quality of Precision's performance on such projects depends in large part upon its ability to manage its relationships with clients and to effectively manage the projects and deploy appropriate resources, including third-party contractors and Precision's personnel, in a timely manner. If Precision miscalculates the resources or time needed to complete a project with capped or fixed fees, or the resources or time needed to meet contractual milestones, its results of operations, cash flows and liquidity could be adversely affected. Additionally, delays on a particular project, including permitting delays, may cause Precision to incur costs for standby pay, and may lead to personnel shortages on other projects scheduled to commence at a later date. In addition, many of Precision's contracts require Precision to share in cost overages or in some cases to pay liquidated damages in the event it does not meet project deadlines, and, therefore, any failure to properly estimate or manage cost or delay in the completion of projects could subject it to penalties which could further materially and adversely affect its results of operations, cash flows and liquidity. Further, any defects or errors, or failures to meet its customers' expectations, could result in large damage claims against it, and, because of the substantial cost of, and potentially long lead-time necessary to acquire certain of the materials and equipment used in, Precision's projects, damage claims may substantially exceed the amount Precision charges for its associated services.

Demand for Precision's pipeline construction services depends on oil and natural gas industry activity and expenditure levels that are directly affected by trends in oil and natural gas prices and the cost of energy infrastructure projects.

Demand for Precision's pipeline construction services is particularly sensitive to the level of exploration, development, and production activity of, and the corresponding capital spending by, oil and natural gas companies. Prices for oil and natural gas are subject to large fluctuations in response to relatively minor changes in the supply of and demand for oil and natural gas, market uncertainty, and a variety of other factors that are beyond Precision's control. Low prices for oil and natural gas generally depress levels of exploration, development, and production activity, resulting in a corresponding decline in the demand for Precision's pipeline construction services. Factors affecting the prices of oil and natural gas include:

- governmental regulations, including the policies of governments regarding the exploration for and production and development of their oil and natural gas reserves, as well as environmental laws and initiatives to control global warming;
- global weather conditions and natural disasters;
- worldwide political, military, and economic conditions;
- the level of oil production by non-OPEC countries and the available excess production capacity within OPEC;
- oil refining capacity and shifts in end-customer preferences toward fuel efficiency and the use of natural gas;

- the cost of producing and delivering oil and gas;
- potential acceleration of development of alternative fuels; and
- the level of supply and demand for oil and natural gas, especially demand for natural gas in the United States.

Historically, the markets for oil and gas have been volatile and are likely to continue to be volatile. Spending on exploration and production activities by large oil and gas companies has a significant impact on the activity levels of Precision's business.

In addition, demand for Precision's pipeline construction services may be affected by the costs of energy exploration and constructing energy infrastructure projects. For example, while high oil and gas prices may increase oil and gas exploration, production and transportation activity, high demand for equipment, materials and labor required for such projects may increase the costs of such projects and dampen demand. Furthermore, increased costs for raw materials such as steel and other commodities may make some projects uneconomical despite robust oil and gas prices, thus reducing demand for Precision's services.

Precision's backlog is subject to cancellation and unexpected adjustments and is an uncertain indicator of future operating results.

Precision's backlog as of September 30, 2009 was over \$500 million, which is an amount derived primarily from long-term projects with two customers. Due to the long-term nature of Precision's contracts and other factors that may affect Precision's projects, not all contractual revenues included in this backlog estimate are expected to be realized by the year ending December 31, 2010. We cannot guarantee that the revenue projected in Precision's backlog will be realized or, if realized, will result in profits. In addition, project cancellations, project deferrals or delays, or scope adjustments may occur, from time to time, with respect to contracts reflected in Precision's backlog. For example, all of Precision's contracts are terminable at the discretion of the client with or without cause. Precision may also experience variances in the realization of its backlog because of project delays or cancellations resulting from weather conditions, permitting delays, external market factors and economic factors beyond its control. These types of backlog reductions could adversely affect its results of operations, cash flows and liquidity. Accordingly, Precision's backlog as of any particular date is an uncertain indicator of its future earnings.

Precision's use of fixed price contracts could adversely affect Precision's business.

Precision currently generates, and expects to continue to generate, a portion of its revenues under fixed price contracts. Precision must estimate the costs of completing a particular project to bid for fixed price contracts. The actual cost of labor and materials, however, may vary from the costs originally estimated. These variations, along with other risks inherent in performing fixed price contracts, may cause actual revenue and gross profits for a project to differ from those originally estimated and could result in reduced profitability or losses on projects. Depending upon the size of a particular project, variations from the estimated contract costs could have a significant impact on Precision's operating results for any fiscal quarter or year.

Precision's use of joint ventures exposes it to risks and uncertainties, many of which are outside of Precision's control.

Precision participates in certain projects through joint ventures, partnerships, and similar arrangements. This situation exposes Precision to a number of risks, including the risk that its partners may be unable to fulfill their obligations to Precision or its clients and that Precision may be liable for the actions of its joint venture partners. Precision's partners may also be unable to provide the required levels of financial support to the joint ventures. If these circumstances occur, Precision may be required to provide additional services or make additional investments to ensure adequate performance and delivery of the contracted services. These circumstances could also lead to disputes and litigation with Precision's partners or clients, which could impact its reputation, business, financial condition, and results of operations. Additionally, Precision participates in joint ventures and similar arrangements in which it is not the controlling partner. In these cases, Precision has limited control over the actions of the joint venture. To the extent the controlling partner or group of partners makes decisions that negatively impact the joint venture, Precision's business, financial condition and results of operations could be negatively impacted.

Precision's key management or key employees may leave, which could materially adversely affect Precision's results of operations.

Our business strategy in acquiring Precision is dependent upon the skills and knowledge of Mr. Michael Daniel Murphy and Mr. Steven R. Rooney, who will be responsible for the day-to-day operations of Precision's business following the acquisition. We believe that the special knowledge of Mr. Murphy and Mr. Rooney gives Precision a competitive advantage. Although Mr. Murphy and Mr. Rooney will each enter into an employment agreement with Precision in connection with the acquisition, there is no guarantee that they will remain with the company. If either Mr. Murphy or Mr. Rooney terminates his employment with Precision, then Precision's operating results could be materially adversely affected. Additionally, Precision's field superintendants, who are the key employees responsible for the day-to-day management of Precision's project sites, are not bound, and will not be bound, by employment agreements, and the departure of any of them could negatively affect Precision's ability to complete projects in a timely manner and otherwise perform in accordance with its agreements with its customers, which could materially adversely affect Precision's liquidity, cash flows and results of operations.

Precision's unionized workforce and related obligations could adversely affect our operations.

Substantially all of Precision's employees are represented by labor unions and collective bargaining agreements. Although all of these agreements prohibit strikes and work stoppages, we cannot be certain that strikes or work stoppages will not occur in the future despite the terms of these agreements. Strikes or work stoppages would adversely impact Precision's relationships with its customers and could cause it to lose business and decrease its revenue. Additionally, as current agreements negotiated by the Pipe Line Contractors Association, of which Precision is a member, expire, the Pipe Line Contractors Association may not be able to negotiate extensions or replacements on terms favorable to its members, including Precision, or at all, or avoid strikes, lockouts or other labor actions from time to time that may affect the Association's members, including Precision. Therefore, it cannot be assured that new agreements will be reached with employee labor unions as existing contracts expire or on terms that Precision finds desirable. Any labor action against Precision relating to the Pipe Line Contractors Association's failure to reach an agreement with employee labor unions could have a material adverse effect on Precision's liquidity, cash flows and results of operations.

Precision's participation in multi-employer pension plans may subject it to liabilities that could materially adversely affect its liquidity, cash flows and results of operations.

Substantially all of Precision's union and collective bargaining agreements require Precision to participate with other companies in multi-employer pension plans. To the extent that those plans are underfunded, the Employee Retirement Income Security Act of 1974, as amended by the Multi-Employer Pension Plan Amendments Act of 1980, may subject Precision, and, after consummation of our acquisition of Precision, us, to substantial liabilities under those plans if Precision withdraws from them or they are terminated. Furthermore, the Pension Protection Act of 2006 added new funding rules generally applicable to plan years beginning after 2007 for multi-employer plans that are classified as "endangered," "seriously endangered," or "critical" status. For a plan in "critical" status, benefit reductions may apply and/or Precision may be required to make additional contributions if a plan is determined to be underfunded, which could materially adversely affect Precision's liquidity, cash flows and results of operations.

Precision's business may be affected by difficult work sites and environments, which could cause delays and result in additional costs to Precision.

Precision performs its work under a variety of conditions, including, but not limited to, difficult and hard to reach terrain and difficult site conditions. Performing work under such conditions can result in project delays or cancellations, potentially causing Precision to incur additional, unanticipated costs, reductions in revenues or the payment of liquidated damages. In addition, some of Precision's contracts require that Precision assume the risk that actual site conditions vary from those expected.

Many of Precision's projects involve challenging engineering, procurement and construction phases that may occur over extended time periods, sometimes over several years. Precision may encounter difficulties in engineering, delays in designs or materials provided by the customer or a third party, equipment and material delivery, schedule changes, delays from customer failure to timely obtain rights-of-way, weather-related delays, delays by subcontractors in completing their portion of the project, and other

factors, some of which are beyond Precision's control, but which impact Precision's ability to complete a project within the original delivery schedule. In some cases, delays and additional costs may be substantial, and Precision may be required to cancel a project and/or compensate the customer for the delay. Precision may not be able to recover any of such costs. Any such delays or cancellations, defects or errors or other failures to meet customer expectations could result in damage claims substantially in excess of the revenue associated with a project, as well as negatively impact Precision's reputation or relationships with customers, adversely affecting Precision's ability to secure new contracts.

Precision may incur liabilities or suffer negative financial or reputational impacts relating to occupational health and safety matters.

Precision's operations are subject to state and federal laws and regulations relating to the maintenance of safe conditions in the workplace. While Precision has invested, and will continue to invest, substantial resources in occupational health and safety programs, its industry involves a high degree of operational risk, and there can be no assurance that Precision will avoid significant liability exposure. Although Precision has taken what are believed to be appropriate precautions, Precision has suffered employee injuries and fatalities in the past and may suffer additional injuries or fatalities in the future. Serious accidents of this nature may subject Precision to substantial penalties, civil litigation or criminal prosecution. Personal injury claims for damages, including for bodily injury or loss of life, could result in substantial costs and liabilities, which could materially and adversely affect Precision's financial condition, results of operations or cash flows. In addition, if Precision's safety record were to substantially deteriorate, or if Precision suffered substantial penalties or criminal prosecution for violation of health and safety regulations, customers could cancel existing contracts and not award future business to Precision, which could materially adversely affect Precision's liquidity, cash flows and results of operations.

Precision's operations may impact the environment or cause exposure to hazardous substances, and its properties may have environmental contamination, which could result in material liabilities to Precision.

Certain of Precision's current and historical construction operations have used hazardous materials and, to the extent that such materials are not properly stored, contained or recycled, they could become hazardous waste. Additionally, some of Precision's contracts require that it assume the environmental risk of site conditions. Precision may be subject to claims under federal and state statutes and/or common law doctrines for toxic torts and other damages, as well as for natural resource damages and the investigation and clean up of soil, surface water, groundwater, and other media under laws such as CERCLA. Such claims may arise, for example, out of current or former conditions at project sites, current or former properties owned or leased by Precision, and contaminated sites that have always been owned or operated by third parties. Liability may be imposed without regard to fault and may be strict, joint and several, such that Precision may be held responsible for more than its share of any contamination or other damages, or even for the entire share, and may be unable to obtain reimbursement from the parties causing the contamination. Some of the facilities leased by Precision, at which equipment is stored, contain fuel storage tanks that are above and below ground. If such tanks were to leak, Precision could be responsible for the cost of environmental remediation, as well as potential punitive fines.

We may not be able to integrate Precision effectively.

Even though we have acquired businesses in the past, and, although we expect Precision's senior management to remain with the company post-acquisition, the acquisition may present significant integration challenges and costs to us. Realization of the benefits of the acquisition will require the integration of some or all of Precision's sales and marketing, finance, information technology systems and administrative operations with ours. The successful integration of Precision will require substantial attention from our senior management and Precision's management, which may decrease the time they devote to normal and customary operating, selling and administrative functions. If we cannot successfully integrate Precision within a reasonable timeframe, then we may not be able to realize the potential benefits anticipated from the acquisition. Our results of operations, cash flows and liquidity could be materially and adversely affected if we do not successfully integrate Precision. Even if we are able to successfully integrate Precisions, we may not be able to realize the cost savings, synergies and revenue enhancements that we anticipate from the acquisition, either in the amount or within the timeframe that we expect, and the costs of achieving these benefits may be higher than, and the timing may differ from, what we expect.

Precision may have liabilities that are not known to us.

Precision may have liabilities that we failed, or were unable, to discover in the course of performing due diligence investigations of Precision. We cannot assure you that the indemnification granted to us by the sellers will be sufficient in amount, scope or duration to fully offset the possible liabilities associated with Precision's business or property that we will assume upon consummation of the acquisition. We may learn additional information about Precision that materially adversely affects us, such as unknown or contingent liabilities and liabilities related to compliance with applicable laws. Any such liabilities, individually or in the aggregate, could have a material adverse effect on our business.

The historical financial information filed as Exhibit 99.2 to this Current Report on Form 8-K may not be representative of the results of the acquired business after the acquisition, and accordingly you have limited financial information on which to evaluate Precision and the combined company.

The audited financial statements of Precision that are filed as Exhibit 99.2 to this Current Report on Form 8-K may not be an accurate representation of the results of operations and financial condition of the acquired business as a part of MasTec. We and Precision have been operating as separate companies prior to the acquisition. We have had no prior history as a combined entity, and our operations have not previously been managed on a combined basis. Therefore, the historical financial statements filed as Exhibit 99.2 to this Current Report on Form 8-K may not be indicative of what the results of operations, financial position and cash flows will be in the future.

This Current Report on Form 8-K does not contain, and we have not yet prepared, interim financial statements for Precision in conformity with Rule 3-05 of Regulation S-X or pro forma financial statements in conformity with Rule 11-01 of Regulation S-X, and, when prepared, such financial statements may contain significant adjustments as compared to the historical financial statements filed as Exhibit 99.2 to this Current Report on Form 8-K and the financial information of Precision included in Item 7.01 of this Current Report on Form 8-K. Additionally, the annual audited financial statements filed as Exhibit 99.2 to this Current Report on Form 8-K include the results of certain businesses that we are not acquiring, and the pro forma financial statements that we will prepare in compliance with Rule 11-01 of Regulation S-X should not be considered indicative of actual results that would have been achieved had we acquired Precision as of the dates thereon indicated.

This Current Report on Form 8-K does not contain, and we have not yet prepared, interim financial statements for Precision in conformity with Rule 3-05 of Regulation S-X or pro forma financial statements in conformity with Rule 11-01 of Regulation S-X. At such time as we prepare interim and pro forma financial statements in conformity with Rules 3-05 and 11-01 of Regulation S-X and file such financial statements with the SEC, they may contain significant adjustments as compared to the historical financial statements filed as Exhibit 99.2 to this Current Report on Form 8-K and information of Precision contained in Item 7.01 of this Current Report on Form 8-K. Additionally, the audited financial statements of Precision include the results of Precision Land Company, LLC, a variable interest entity which is historically related to Precision but which we are not acquiring; therefore, the historical results contained in the audited financial statements are not entirely representative of the businesses we are acquiring in connection with the Precision acquisition. Moreover, the historical financial statements and our expectations with respect to certain adjustments that we will make in connection with the preparation of pro forma financial statements in compliance with Rule 11-01 of Regulation S-X, should not be considered indicative of actual results that would have been achieved had the Precision acquisition been completed as of the dates indicated and do not purport to project the financial condition or results of operations for any future date or period.

Item 9.01 Financial Statements and Exhibits.

- (a) Financial Statements of the Business Acquired.
 - Not applicable.
- (b) Pro Forma Financial Information.
 - Not applicable.
- (c) Shell Company Transactions.
 - Not applicable.
- (d) Exhibits.
 - 10.1 Purchase Agreement, dated November 3, 2009, by and among MasTec, Inc., Precision Acquisition, LLC, Precision Pipeline LLC, Precision Transport Company, LLC, PPL Management, Inc., Michael Daniel Murphy, Steven R. Rooney, Angela D. Murphy and Karen K. Rooney.
 - 23.1 Consent of LarsonAllen LLP.
 - 99.1 Press Release dated November 3, 2009 (relating to the Transaction).
 - 99.2 Audited consolidated balance sheet of PPL Management, Inc. and subsidiaries as of December 31, 2008 and 2007, and the audited consolidated statements of operations, equity, and cash flows for the years ended December 31, 2008, 2007 and 2006.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

MASTEC, INC.

Date: November 4, 2009

By: /s/ C. Robert Campbell

Name: C. Robert Campbell

Title: Executive Vice President and Chief Financial Officer

12

EXHIBIT INDEX

Exhibit No.	<u>Description</u>
10.1	Purchase Agreement, dated November 3, 2009, by and among MasTec, Inc., Precision Acquisition, LLC, Precision Pipeline LLC, Precision Transport Company, LLC, PPL Management, Inc., Michael Daniel Murphy, Steven R. Rooney, Angela D. Murphy and Karen K. Rooney.
23.1	Consent of LarsonAllen LLP.
99.1	Press Release dated November 3, 2009 (relating to the Transaction).
99.2	Audited consolidated balance sheet of PPL Management, Inc. and subsidiaries as of December 31, 2008 and 2007, and the audited consolidated statements of operations, equity, and cash flows for the years ended December 31, 2008, 2007 and 2006.

MEMBERSHIP INTEREST PURCHASE AGREEMENT

Dated November 3, 2009

by and among

Precision Acquisition, LLC,

MasTec, Inc.,

Precision Pipeline LLC,

Precision Transport Company, LLC,

PPL Management, Inc.,

Michael Daniel Murphy,

Steven R. Rooney,

Angela D. Murphy,

and

Karen K. Rooney

TABLE OF CONTENTS

2

PURCHASE AND SALE OF UNITS.

2.

1.1. Purchase Price 2 1.2. Purchase Price Allocation 2 1.3 Estrate of Purchase Price 3 1.5 Escrow 3 1.6 Escrow 4 1.7 Allocation of Payments 4 1.8 Post-Closing Purchase Price Adjustment 5 1.9 Closing 7 1.10 Earn-Out 10 1.11 Collection and Assignment of Actual Uncollected Receivables 15 1.12 Reimbursement for and Assignment of Uncollected Receivables 15 1.12 Reimbursement for and Assignment of Uncollected Receivables 16 REPRESENTATIONS AND WARRANTIES OF THE SELLERS 16 2.1 Organization, Predecessors 16 2.2 Capitalization of the Companies; Title to Units 17 2.3 Organization, Power and Authorization 18 2.4 Authorization of Governmental Authorities 18 2.5 Noncontravention 18 2.6 Financial Statements 19 2.7 Absence of Undisclosed Liabilities 20 2.8 Absence of Certain Developments 20 2.9 Assets 22 2.11 Backlog 23 2.12 Tapitolic Every Life and Developments 23 <		D 1 101 177 1	
1.1.3 Purchase Price Allocation 3 1.4 Estimate of Purchase Price 3 1.5 Payment of Closing Date Purchase Price 3 1.6 Escrow 4 1.7 Allocation of Payments 4 1.8 Post-Closing Purchase Price Adjustment 5 1.10 Colosing 7 1.11 Collection and Assignment of Actual Uncollected Receivables 15 1.12 Reimbursement for and Assignment of Uncollected Retainage 16 REPRESENTATIONS AND WARRANTIES OF THE SELLERS 16 2.1 Organization: Predecessors 16 2.2 Capitalization of the Companies; Title to Units 17 2.3 Organization: Predecessors 18 2.4 Authorization of Governmental Authorities 18 2.4 Authorization of Governmental Authorities 18 2.5 Noncontravention 18 2.6 Financial Statements 19 2.7 Absence of Undisclosed Liabilities 20 2.8 Absence of Certain Developments 20 2.9 Assets 22	1.1	Purchase and Sale of Units	2
1.4. Estimate of Purchase Price 3 1.5. Payment of Closing Date Purchase Price 3 1.6. Excrow 4 1.7. Allocation of Payments 4 1.8. Post-Closing Purchase Price Adjustment 5 1.9. Closing 7 1.0. Iam-Out 10 1.1.1 Collection and Assignment of Actual Uncollected Receivables 15 1.1.2 Reimbursement for and Assignment of Uncollected Receivables 15 1.1.2 Reimbursement for and Assignment of Uncollected Receivables 16 2.1 Organization; Predecessors 16 2.2 Capitalization of the Companies; Title to Units 17 2.2 Capitalization of the Companies; Title to Units 17 2.3 Organization; Predecessors 18 2.4 Authorization of Governmental Authorities 18 2.5 Noncontravention 18 2.6 Financial Statements 19 2.6 Financial Statements 20 2.7 Absence of Undisclosed Liab			
1.5 Payment of Closing Date Purchase Price 3 1.6 Escrow 4 1.7 Allocation of Payments 4 1.8 Post-Closing Purchase Price Adjustment 5 1.9 Closing 7 1.10 Earn-Out 10 1.11 Collection and Assignment of Actual Uncollected Receivables 15 1.12 Reimbursement for and Assignment of Uncollected Retainage 16 Reimbursement for and Assignment of Uncollected Retainage 16 REPRESENTATIONS AND WARRANTIES OF THE SELLERS 16 2.1 Organization; Predecessors 16 2.1 Organization, Power and Authorization 18 2.3 Organization, Power and Authorization 18 2.4 Authorization of Governmental Authorities 18 2.5 Noncontravention 18 2.6 Financial Statements 20 2.8 Absence of Undisclosed Liabilities 20 2.9 Asset 22 2.1 Backlog 23 2			
1.6 Escrow 4 1.7 Allocation of Payments 4 1.8 Post-Closing Purchase Price Adjustment 5 1.9 Closing 7 1.10 Earn-Out 10 1.11 Collection and Assignment of Actual Uncollected Receivables 15 1.12 Reimbursement for and Assignment of Uncollected Retainage 16 REPRESENTATIONS AND WARRANTIES OF THE SELLERS 16 2.1 Organization; Predecessors 17 2.2 Capitalization of the Companies: Title to Units 17 2.3 Organization, Power and Authorization 18 2.4 Authorization of Governmental Authorities 18 2.5 Noncontravention 18 2.6 Financial Statements 19 2.7 Absence of Undisclosed Liabilities 20 2.8 Absence of Undisclosed Liabilities 20 2.9 Assets 22 2.10 Accounts Receivable 23 2.11 Backlog 23 2.12 <t< td=""><td></td><td></td><td></td></t<>			
1.7 Allocation of Payments 4 1.8 Post-Closing Purchase Price Adjustment 5 1.9 Closing 7 1.10 Earn-Out 10 1.11 Collection and Assignment of Actual Uncollected Receivables 15 1.12 Reimbursement for and Assignment of Uncollected Retainage 16 REPRESENTATIONS AND WARRANTIES OF THE SELLERS 16 2.1 Organization; Predecessors 16 2.2 Capitalization of the Companies; Title to Units 17 2.3 Organization, Power and Authorization 18 2.4 Authorization of Governmental Authorities 18 2.5 Noncontravention 18 2.6 Financial Statements 19 2.7 Absence of Undisclosed Liabilities 20 2.8 Absence of Undisclosed Liabilities 20 2.8 Absence of Certain Developments 20 2.8 Absence of Undisclosed Liabilities 20 2.1 Backlog 23 2.1 Backlog 23			
1.8 Post-Closing Purchase Price Adjustment 5 1.9 Closing 7 1.10 Eam-Out 10 1.11 Collection and Assignment of Actual Uncollected Reteinage 15 1.12 Reimbursement for and Assignment of Uncollected Retainage 16 REPRESENTATIONS AND WARRANTIES OF THE SELLERS 16 2.1 Capitalization; Predecessors 16 2.2 Capitalization of the Companies; Title to Units 17 2.3 Organization, Power and Authorization 18 2.4 Authorization of Governmental Authorities 18 2.5 Noncontravention 18 2.6 Financial Statements 19 2.7 Absence of Undisclosed Liabilities 20 2.8 Absence of Certain Developments 20 2.9 Assets 22 2.10 Accounts Receivable 23 2.11 Backlog 23 2.12 Real Property 25 2.13 Tangible Personal Property 25 2.16			
1.9 Closing 7 1.10 Earn-Out 10 1.11 Collection and Assignment of Actual Uncollected Receivables 15 1.12 Reimbursement for and Assignment of Uncollected Retainage 16 REPRESENTATIONS AND WARRANTIES OF THE SELLERS 16 2.1 Organization; Predecessors 16 2.2 Capitalization of the Companies; Title to Units 17 2.3 Organization, Power and Authorization 18 2.4 Authorization of Governmental Authorities 18 2.5 Noncontravention 18 2.6 Financial Statements 19 2.7 Absence of Undisclosed Liabilities 20 2.8 Absence of Certain Developments 20 2.9 Assets 22 2.10 Accounts Receivable 23 2.11 Backlog 23 2.12 Real Property 25 2.13 Tanglible Personal Property 25 2.14 Intellectual Property 25 2.15			
1.10 Eam-Out 10 1.11 Collection and Assignment of Actual Uncollected Receivables 15 1.12 Reimbursement for and Assignment of Uncollected Retainage 16 REPRESENTATIONS AND WARRANTIES OF THE SELLERS 16 2.1 Organization; Predecessors 16 2.2 Capitalization of the Companies; Title to Units 17 2.3 Organization, Power and Authorization 18 2.4 Authorization of Governmental Authorities 18 2.5 Noncontravention 18 2.6 Financial Statements 19 2.7 Absence of Undisclosed Liabilities 20 2.8 Absence of Undisclosed Liabilities 20 2.9 Assets 22 2.10 Accounts Receivable 23 2.21 Accounts Receivable 23 2.12 Red Property 23 2.13 Tangible Personal Property 25 2.14 Intellectual Property 25 2.15 Legal Compliance; Illegal Payments; Permits 29 2.16 Environmental Matters 33			
1.11 Collection and Assignment of Actual Uncollected Receivables 15 1.12 Reimbursement for and Assignment of Uncollected Retainage 16 REPRESENTATIONS AND WARRANTIES OF THE SELLERS 16 2.1 Organization; Predecessors 16 2.2 Capitalization of the Companies; Title to Units 17 2.3 Organization, Power and Authorization 18 2.4 Authorization of Governmental Authorities 18 2.5 Noncontravention 18 2.6 Financial Statements 19 2.7 Absence of Undisclosed Liabilities 20 2.8 Absence of Undisclosed Liabilities 20 2.9 Assets 22 2.10 Accounts Receivable 23 2.11 Backlog 23 2.12 Backlog 23 2.13 Tangible Personal Property 25 2.14 Intellectual Property 25 2.15 Legal Compliance; Illegal Payments; Permits 27 2.16 Inventories 29 2.17 Employee Benefit Plans 33			
1.12 Reimbursement for and Assignment of Uncollected Retainage 16 REPRESENTATIONS AND WARRANTIES OF THE SELLERS 16 2.1 Organization; Predecessors 16 2.2 Capitalization of the Companies; Title to Units 17 2.3 Organization, Power and Authorization 18 2.4 Authorization of Governmental Authorities 18 2.5 Noncontravention 18 2.6 Financial Statements 19 2.7 Absence of Undisclosed Liabilities 20 2.8 Absence of Certain Developments 20 2.9 Assets 22 2.10 Accounts Receivable 23 2.11 Backlog 23 2.12.1 Real Property 23 2.13 Tangible Personal Property 25 2.14 Intellectual Property 25 2.15 Legal Compliance; Illegal Payments; Permits 27 2.16 Inventories 29 2.17 Employee Benefit Plans 29 2.18 Environmental Matters 33 2.20 Affiliate Transacti	1.10		
REPRESENTATIONS AND WARRANTIES OF THE SELLERS 16 2.1 Organization; Predecessors 16 2.2 Capitalization of the Companies; Title to Units 17 2.3 Organization, Power and Authorization 18 2.4 Authorization of Governmental Authorities 18 2.5 Noncontravention 18 2.6 Financial Statements 19 2.7 Absence of Undisclosed Liabilities 20 2.8 Absence of Certain Developments 20 2.9 Assets 22 2.10 Accounts Receivable 23 2.11 Backlog 23 2.12 Real Property 23 2.13 Tangible Personal Property 25 2.14 Intelectual Property 25 2.15 Legal Compliance; Illegal Payments; Permits 27 2.16 Inventories 29 2.17 Employee Benefit Plans 29 2.18 Environmental Matters 33 2.19 Customers and Suppliers 37 <td></td> <td></td> <td></td>			
2.1 Organization; Predecessors 16 2.2 Capitalization of the Companies; Title to Units 17 2.3 Organization, Power and Authorization 18 2.4 Authorization of Governmental Authorities 18 2.5 Noncontravention 18 2.6 Financial Statements 19 2.7 Absence of Undisclosed Liabilities 20 2.8 Absence of Certain Developments 20 2.9 Assets 22 2.10 Accounts Receivable 23 2.11 Backlog 23 2.12 Real Property 23 2.13 Tangible Personal Property 25 2.14 Intellectual Property 25 2.15 Legal Compliance; Illegal Payments; Permits 27 2.16 Inventories 29 2.17 Employee Benefit Plans 29 2.18 Environmental Matters 33 2.19 Customers and Suppliers 37 2.21 Customers and Suppliers 37 2.22 Customers warranties 38	1.12	Reimbursement for and Assignment of Uncollected Retainage	16
2.2 Capitalization of the Companies; Title to Units 17 2.3 Organization, Power and Authorization 18 2.4 Authorization of Governmental Authorities 18 2.5 Noncontravention 18 2.6 Financial Statements 19 2.7 Absence of Undisclosed Liabilities 20 2.8 Absence of Certain Developments 20 2.9 Assets 22 2.10 Accounts Receivable 23 2.11 Backlog 23 2.12 Real Property 23 2.13 Tangible Personal Property 25 2.14 Intellectual Property 25 2.15 Legal Compliance; Illegal Payments; Permits 27 2.16 Inventories 29 2.17 Employee Benefit Plans 29 2.18 Contracts 33 2.21 Customers and Suppliers 37 2.22 Customers and Suppliers 38 2.23 Capital Expenditures and Investments 38 2.24 Employees 39	REPR	ESENTATIONS AND WARRANTIES OF THE SELLERS	16
2.3 Organization, Power and Authorization 18 2.4 Authorization of Governmental Authorities 18 2.5 Noncontravention 18 2.6 Financial Statements 19 2.7 Absence of Undisclosed Liabilities 20 2.8 Absence of Certain Developments 20 2.9 Assets 22 2.10 Accounts Receivable 23 2.11 Backlog 23 2.12 Real Property 23 2.13 Tangible Personal Property 25 2.14 Intellectual Property 25 2.15 Legal Compliance; Illegal Payments; Permits 27 2.16 Inventories 29 2.17 Employee Benefit Plans 29 2.18 Environmental Matters 33 2.21 Customers and Suppliers 37 2.22 Customers and Suppliers 37 2.23 Capital Expenditures and Investments 38 2.24 Employees 39 2.25 Litigation; Governmental Orders 40 <td< td=""><td>2.1</td><td>Organization; Predecessors</td><td>16</td></td<>	2.1	Organization; Predecessors	16
2.4 Authorization of Governmental Authorities 18 2.5 Noncontravention 18 2.6 Financial Statements 19 2.7 Absence of Undisclosed Liabilities 20 2.8 Absence of Certain Developments 20 2.9 Assets 22 2.10 Accounts Receivable 23 2.11 Backlog 23 2.12 Real Property 23 2.13 Tangible Personal Property 25 2.14 Intellectual Property 25 2.15 Legal Compliance; Illegal Payments; Permits 27 2.16 Inventories 29 2.17 Employee Benefit Plans 29 2.18 Environmental Matters 33 2.19 Contracts 34 2.20 Affiliate Transactions 37 2.21 Customer and Suppliers 37 2.22 Customer Warranties 38 2.23 Capital Expenditures and Investments 38 2.24 Employees 39 2.25 Linsuance	2.2	Capitalization of the Companies; Title to Units	17
2.5 Noncontravention 18 2.6 Financial Statements 19 2.7 Absence of Undisclosed Liabilities 20 2.8 Absence of Certain Developments 20 2.9 Assets 22 2.10 Accounts Receivable 23 2.11 Backlog 23 2.12 Real Property 23 2.13 Tangible Personal Property 25 2.14 Intellectual Property 25 2.15 Legal Compliance; Illegal Payments; Permits 27 2.16 Inventories 29 2.17 Employee Benefit Plans 29 2.18 Environmental Matters 33 2.19 Contracts 34 2.20 Affiliate Transactions 37 2.21 Customers and Suppliers 37 2.22 Customer Warranties 38 2.23 Capital Expenditures and Investments 38 2.24 Employees 39 2.25 Itigation; Governmental Orders 41 2.26 Insurance 41 </td <td>2.3</td> <td>Organization, Power and Authorization</td> <td>18</td>	2.3	Organization, Power and Authorization	18
2.6 Financial Statements 19 2.7 Absence of Undisclosed Liabilities 20 2.8 Absence of Certain Developments 20 2.9 Assets 22 2.10 Accounts Receivable 23 2.11 Backlog 23 2.12 Real Property 23 2.13 Tangible Personal Property 25 2.14 Intellectual Property 25 2.15 Legal Compliance; Illegal Payments; Permits 27 2.16 Inventories 29 2.17 Employee Benefit Plans 29 2.18 Environmental Matters 33 2.19 Contracts 34 2.20 Affiliate Transactions 37 2.21 Customers and Suppliers 37 2.22 Customer Warranties 38 2.23 Capital Expenditures and Investments 38 2.24 Employees 39 2.25 Litigation; Governmental Orders 40 2.26 Insurance 41 2.27 Banking Facilities	2.4	Authorization of Governmental Authorities	18
2.7 Absence of Undisclosed Liabilities 20 2.8 Absence of Certain Developments 20 2.9 Assets 22 2.10 Accounts Receivable 23 2.11 Backlog 23 2.12 Real Property 23 2.13 Tangible Personal Property 25 2.14 Intellectual Property 25 2.15 Legal Compliance; Illegal Payments; Permits 27 2.16 Inventories 29 2.17 Employee Benefit Plans 29 2.18 Environmental Matters 33 2.20 Affiliate Transactions 34 2.21 Customers and Suppliers 37 2.22 Customer Warranties 38 2.23 Capital Expenditures and Investments 38 2.24 Employees 39 2.25 Litigation; Governmental Orders 40 2.26 Insurance 41 2.27 Banking Facilities 41	2.5	Noncontravention	18
2.8 Absence of Certain Developments 20 2.9 Assets 22 2.10 Accounts Receivable 23 2.11 Backlog 23 2.12 Real Property 23 2.13 Tangible Personal Property 25 2.14 Intellectual Property 25 2.15 Legal Compliance; Illegal Payments; Permits 27 2.16 Inventories 29 2.17 Employee Benefit Plans 29 2.18 Environmental Matters 33 2.19 Contracts 34 2.20 Affiliate Transactions 34 2.21 Customers and Suppliers 37 2.22 Customer Warranties 38 2.23 Capital Expenditures and Investments 38 2.24 Employees 39 2.25 Litigation; Governmental Orders 40 2.26 Insurance 41 2.27 Banking Facilities 41	2.6	Financial Statements	19
2.9 Assets 2.10 Accounts Receivable 2.11 Backlog 2.12 Real Property 2.13 Tangible Personal Property 2.14 Intellectual Property 2.15 Legal Compliance; Illegal Payments; Permits 2.16 Inventories 2.17 Employee Benefit Plans 2.18 Environmental Matters 2.19 Contracts 2.20 Affiliate Transactions 2.21 Customers and Suppliers 2.22 Customer Warranties 2.23 Capital Expenditures and Investments 2.24 Employees 2.25 Litigation; Governmental Orders 2.26 Insurance 40 2.27 Banking Facilities	2.7	Absence of Undisclosed Liabilities	20
2.9 Assets 2.10 Accounts Receivable 2.11 Backlog 2.12 Real Property 2.13 Tangible Personal Property 2.14 Intellectual Property 2.15 Legal Compliance; Illegal Payments; Permits 2.16 Inventories 2.17 Employee Benefit Plans 2.18 Environmental Matters 2.19 Contracts 2.20 Affiliate Transactions 2.21 Customers and Suppliers 2.22 Customer Warranties 2.23 Capital Expenditures and Investments 2.24 Employees 2.25 Litigation; Governmental Orders 2.26 Insurance 40 2.27 Banking Facilities	2.8	Absence of Certain Developments	20
2.11 Backlog 23 2.12 Real Property 23 2.13 Tangible Personal Property 25 2.14 Intellectual Property 25 2.15 Legal Compliance; Illegal Payments; Permits 27 2.16 Inventories 29 2.17 Employee Benefit Plans 29 2.18 Environmental Matters 33 2.19 Contracts 34 2.20 Affiliate Transactions 37 2.21 Customers and Suppliers 37 2.22 Customer Warranties 38 2.23 Capital Expenditures and Investments 38 2.24 Employees 39 2.25 Litigation; Governmental Orders 40 2.26 Insurance 41 2.27 Banking Facilities 41	2.9		22
2.12 Real Property 23 2.13 Tangible Personal Property 25 2.14 Intellectual Property 25 2.15 Legal Compliance; Illegal Payments; Permits 27 2.16 Inventories 29 2.17 Employee Benefit Plans 29 2.18 Environmental Matters 33 2.19 Contracts 34 2.20 Affiliate Transactions 34 2.21 Customers and Suppliers 37 2.22 Customer Warranties 38 2.23 Capital Expenditures and Investments 38 2.24 Employees 39 2.25 Litigation; Governmental Orders 40 2.26 Insurance 41 2.27 Banking Facilities 41	2.10	Accounts Receivable	23
2.12 Real Property 23 2.13 Tangible Personal Property 25 2.14 Intellectual Property 25 2.15 Legal Compliance; Illegal Payments; Permits 27 2.16 Inventories 29 2.17 Employee Benefit Plans 29 2.18 Environmental Matters 33 2.19 Contracts 34 2.20 Affiliate Transactions 34 2.21 Customers and Suppliers 37 2.22 Customer Warranties 38 2.22 Customer Warranties 38 2.23 Capital Expenditures and Investments 38 2.24 Employees 39 2.25 Litigation; Governmental Orders 40 2.26 Insurance 41 2.27 Banking Facilities 41	2.11	Backlog	23
2.14 Intellectual Property 25 2.15 Legal Compliance; Illegal Payments; Permits 27 2.16 Inventories 29 2.17 Employee Benefit Plans 29 2.18 Environmental Matters 33 2.19 Contracts 34 2.20 Affiliate Transactions 37 2.21 Customers and Suppliers 37 2.22 Customer Warranties 38 2.23 Capital Expenditures and Investments 38 2.24 Employees 39 2.25 Litigation; Governmental Orders 40 2.26 Insurance 41 2.27 Banking Facilities 41		-	23
2.14 Intellectual Property 25 2.15 Legal Compliance; Illegal Payments; Permits 27 2.16 Inventories 29 2.17 Employee Benefit Plans 29 2.18 Environmental Matters 33 2.19 Contracts 34 2.20 Affiliate Transactions 37 2.21 Customers and Suppliers 37 2.22 Customer Warranties 38 2.23 Capital Expenditures and Investments 38 2.24 Employees 39 2.25 Litigation; Governmental Orders 40 2.26 Insurance 41 2.27 Banking Facilities 41	2.13	Tangible Personal Property	25
2.15 Legal Compliance; Illegal Payments; Permits 27 2.16 Inventories 29 2.17 Employee Benefit Plans 29 2.18 Environmental Matters 33 2.19 Contracts 34 2.20 Affiliate Transactions 37 2.21 Customers and Suppliers 37 2.22 Customer Warranties 38 2.23 Capital Expenditures and Investments 38 2.24 Employees 39 2.25 Litigation; Governmental Orders 40 2.26 Insurance 41 2.27 Banking Facilities 41			25
2.16 Inventories 29 2.17 Employee Benefit Plans 29 2.18 Environmental Matters 33 2.19 Contracts 34 2.20 Affiliate Transactions 37 2.21 Customers and Suppliers 37 2.22 Customer Warranties 38 2.23 Capital Expenditures and Investments 38 2.24 Employees 39 2.25 Litigation; Governmental Orders 40 2.26 Insurance 41 2.27 Banking Facilities 41			
2.17 Employee Benefit Plans 29 2.18 Environmental Matters 33 2.19 Contracts 34 2.20 Affiliate Transactions 37 2.21 Customers and Suppliers 37 2.22 Customer Warranties 38 2.23 Capital Expenditures and Investments 38 2.24 Employees 39 2.25 Litigation; Governmental Orders 40 2.26 Insurance 41 2.27 Banking Facilities 41			29
2.18 Environmental Matters 33 2.19 Contracts 34 2.20 Affiliate Transactions 37 2.21 Customers and Suppliers 37 2.22 Customer Warranties 38 2.23 Capital Expenditures and Investments 38 2.24 Employees 39 2.25 Litigation; Governmental Orders 40 2.26 Insurance 41 2.27 Banking Facilities 41	2.17	Employee Benefit Plans	
2.19 Contracts 34 2.20 Affiliate Transactions 37 2.21 Customers and Suppliers 37 2.22 Customer Warranties 38 2.23 Capital Expenditures and Investments 38 2.24 Employees 39 2.25 Litigation; Governmental Orders 40 2.26 Insurance 41 2.27 Banking Facilities 41			33
2.21 Customers and Suppliers372.22 Customer Warranties382.23 Capital Expenditures and Investments382.24 Employees392.25 Litigation; Governmental Orders402.26 Insurance412.27 Banking Facilities41	2.19	Contracts	
2.21 Customers and Suppliers372.22 Customer Warranties382.23 Capital Expenditures and Investments382.24 Employees392.25 Litigation; Governmental Orders402.26 Insurance412.27 Banking Facilities41	2.20	Affiliate Transactions	37
2.22 Customer Warranties382.23 Capital Expenditures and Investments382.24 Employees392.25 Litigation; Governmental Orders402.26 Insurance412.27 Banking Facilities41			
2.24 Employees392.25 Litigation; Governmental Orders402.26 Insurance412.27 Banking Facilities41			38
2.24 Employees392.25 Litigation; Governmental Orders402.26 Insurance412.27 Banking Facilities41	2.23	Capital Expenditures and Investments	38
2.25 Litigation; Governmental Orders402.26 Insurance412.27 Banking Facilities41			
2.26 Insurance412.27 Banking Facilities41			
2.27 Banking Facilities 41			

2.22	W D 1	
2.29	No Brokers	41
2.30	Securities Law Matters	41
2.31	Surety Bonds	42
2.32	Transport Operations	43
2.33	PPL	43
REPR	ESENTATIONS AND WARRANTIES OF THE BUYER AND MASTEC	43
3.1	Organization	43
3.2	Power and Authorization	43
3.3	Authorization of Governmental Authorities	43
3.4	Noncontravention	44
3.5	No Brokers	44
3.6	Securities Law Compliance	44
3.7	MasTec Financial Statements	44
3.8	MasTec Material Adverse Effect	44
3.9	No Actual Knowledge of Certain Breaches	44
PRE-C	CLOSING COVENANTS	44
4.1		4.4
4.1	Access and Investigation	44
4.2	Operation of the Companies and their Businesses	45
4.3	Commercially Reasonable Efforts; Notification; Hart-Scott-Rodino Act Filings	48
4.4	Acquisition Proposals	49
4.5	Bank Accounts; Powers of Attorney	50
4.6	Notice and Cure	50
4.7	Consultation	51
4.8	Cooperation Regarding: Financing	51
4.9	Interim Financial Statements	51
4.10	Termination of Certain Employees	52
4.11	Assistance	52
4.12	Guarantees by Sellers	52
4.13	Consents for Assumption of Debt	52
4.14	Confidentiality	52
POST-	CLOSING COVENANTS	53
5.1	Confidentiality	53
5.2	Restrictive Covenants	54
5.3	Cooperation Regarding Financial Statements	57
5.4	Litigation Support	57
5.5	Access to Books and Records	57
5.6	LLC Indemnification Provisions	58
5.7	Employment and Benefit Arrangements	58
5.8	Warranty Repairs	59
5.9	Workers' Compensation Insurance	59
5.10	Lien Releases and Termination Statements	59 59
2.10	Pien Veicazes and Telliniquon statements	59

3.

4.

5.

õ.	CONI	DITIONS PRECEDENT TO BUYER'S OBLIGATION TO CLOSE	59
	6.1	Sellers' and the Companies' Performance	59
	6.2	No Actions	60
	6.3	No Claim Regarding Units or Sale Proceeds	60
	6.4	No Prohibition	60
	6.5	No Material Adverse Effect	60
	6.6	HSR Act	60
	6.7	Escrow Agreement	61
	6.8	Third-Party Consents	61
	6.9	Financing	61
7.	CONI	DITIONS PRECEDENT TO SELLERS' OBLIGATION TO CLOSE	61
	7.1	Buyer's Performance	61
	7.2	No Prohibition	61
	7.3	HSR Act	62
	7.4	No MasTec Material Adverse Effect	62
3.	TERM	MINATION	62
	8.1	Termination Events	62
	8.2	Effect of Termination	63
€.	INDE	MNIFICATION	63
	9.1	Indemnification by the Sellers	63
	9.2	Indemnity by the Buyer and MasTec	65
	9.3	Time for Claims	65
	9.4	Third Party Claims	66
	9.5	Sellers' Representative	68
	9.6	Knowledge and Investigation	68
	9.7	Remedies Cumulative	68
	9.8	Right of Setoff	68
	9.9	Companies' Indemnification Obligations	68
	9.10	Time of Payment of Claims	69
	9.11	Insurance and Other Sources of Reimbursement	69
	9.12	Purchase Price Adjustment	69
	9.13	Effect of Computation of Actual Net Working Capital	70
	9.14	Satisfaction of Indemnity Claims	70
	9.15	Limitations on Liability	70
		Exclusive Remedy	70
		Allocation of Purchase Price Is Not Agreement of Losses	71
10.	TAX I	MATTERS	71
	10.1	Representations and Obligations Regarding Taxes	71
	10.2	Covenants With Respect To Taxes	72

10.3	Indemnification for Taxes	74
MISC	ELLANEOUS	77
11.1	Notices	77
11.2	Sellers' Representative	79
11.3	Publicity	80
11.4	Succession and Assignment; No Third-Party Beneficiary	81
11.5	Amendments and Waivers	81
	Further Assurances	81
	Entire Agreement	81
11.8		82
	Counterparts; Execution	82
	O Survival	82
	Severability	82
	2 Headings	82
	3 Construction	82
	4 Governing Law	83
	5 Jurisdiction; Venue; Service of Process	83
	6 Waiver of Jury Trial	83
	7 Expenses	84
	3 Specific Performance	84
	9 Spousal Consent	84
) Joint and Several	84
	Legal Representation	85
11.22	2 Disclaimer	85
DEFIN	NITIONS; CERTAIN RULES OF CONSTRUCTION	86
12.1	Definitions	86
12.2	Glossary of Other Defined Terms	98
12.3	Rules of Construction	101
oits:		
oit A	Form of Closing Certificate	

Exhibits:

12.

11.

Exhibit A	Form of Closing Certificate
Exhibit B	Escrow Agreement
Exhibit C	Final Closing Statement
Exhibit D	Form of Headquarters Lease
Exhibit E	Form of General Release
Exhibit F	Form of Registration Rights Agreement
Exhibit G	Form of Principals Employment Agreement
Exhibit H	Form of CAC Representation Letter
Exhibit I	Form of Gardere Representation Letter

SCHEDULES

1.3	Purchase Price Allocation
1.5.1(f)	Buyer's Expenses
1.9.2(a)(vii)	Payoff Letters
2.1.1	Jurisdictions in which Company is qualified to transact business
2.1.2	Predecessor Status, etc.
2.2.2	Ownership of Units
2.2.3	Encumbrances, Etc.
2.2.4	Debt
2.4	Authorization of Governmental Authorities
2.5	Noncontravention
2.6.1	Financial Statements
2.6.2	Compliance with GAAP
2.7	Absence of Undisclosed Liabilities
2.8	Certain Developments
2.9.1	Ownership of Assets
2.9.2	Sufficiency of Assets
2.10	Accounts Receivable
2.11	Backlog
2.12	Real Property
2.12.2	Real Property Consents
2.14	Intellectual Property
2.15.3	Permits
2.16	Inventories
2.17(a)	Employee Benefit Plans
2.17(c)	Employee Benefit Plan Administration
2.17(e)	Employee Benefit Plan Actions, Suits and Claims
2.17(g)	Employee Benefit Plans Increases in the Expenses of Maintaining
2.17(h)	Employee Plan Compensation Subject to 409A of the Code
2.17(i)	Certain Matters with Respect to Employee Benefit Plans
2.18	Environmental Matters
2.19	Contracts
2.19.3	Breach
2.19.4	Bids
2.20	Affiliate Transactions
2.21	Customers and Suppliers
2.22	Customer Warranty Claims
2.23	Capital Expenditures and Investments
2.24	Employees
2.24.3	Actions Relating to Employees
2.25	Litigation
2.26	Insurance
2.27	Banking Facilities
2.28	Powers of Attorney

Allocation of the Purchase Price among the Sellers

1.2

2.31	Surety Bonds
2.32	Transport Operations
2.33	PPL
4.10	Transferred Employees
9.1.5	Certain Indemnification Matters
9.11	Insurance to be maintained by the Companies following Closing
10	Taxes
10.1.1	Tax Returns
10.1.2	Tax Clearances
10.1.3	Tax Audits and Disputes
10.1.4	Ventures Taxed as Partnerships
10.1.6	Treatment of Income From Prior Periods
10.1.9	Withholding Taxes
12.1	Precision GAAP

MEMBERSHIP INTEREST PURCHASE AGREEMENT

This Membership Interest Purchase Agreement, dated November 3, 2009, as amended or otherwise modified (the "<u>Agreement</u>"), by and among Precision Acquisition, LLC, a Wisconsin limited liability company (the "<u>Buyer</u>"), MasTec, Inc., a Florida corporation ("<u>MasTec</u>"), Precision Pipeline LLC, a Wisconsin limited liability company ("<u>Precision</u>"), Precision Transport Company, LLC, a Wisconsin limited liability company ("<u>Transport</u>" and, together with Precision, the "<u>Companies</u>" and each individually, a "<u>Company</u>"), PPL Management, Inc., a Wisconsin corporation ("<u>PPL</u>"), Michael Daniel Murphy, an individual resident in the State of Wisconsin ("<u>Mooney</u>", and, together with Murphy, the "<u>Transport Sellers</u>", and the Transport Sellers, together with PPL, the "<u>Sellers</u>"), Murphy and Rooney as the Sellers' representatives (collectively, the "<u>Sellers</u>" Representative"), and Angela D. Murphy, an individual resident in the State of Wisconsin ("<u>Mrs. Murphy</u>"), and Karen K. Rooney, an individual resident in the State of Wisconsin ("<u>Mrs. Rooney</u>"), each of Mrs. Murphy and Mrs. Rooney solely with respect to <u>Sections 11.19 and 11.20</u> and with respect to no other provision of this Agreement.

RECITALS

WHEREAS, the Sellers are the record and beneficial owners of 100% of the issued and outstanding membership interests in Precision (the "<u>Precision Units</u>");

WHEREAS, the Transport Sellers are the record and beneficial owners of 100% of the issued and outstanding membership interests in Transport (the "Transport Units" and, together with the Precision Units, the "Units");

WHEREAS, Precision is Transport's only customer, and Transport's sole business is the provision of carrier services with the use of certain drivers, trucks, trailers and motor vehicles for the purpose of transporting Precision's Equipment to jobsites (the "<u>Transport Business</u>");

WHEREAS, PPL employs the Transferred Employees and desires to as of the Closing Date terminate its employment of the Transferred Employees and Precision desires to employ the Transferred Employees as of the Closing Date;

WHEREAS, Mrs. Murphy is the spouse of Murphy and Mrs. Rooney is the spouse of Rooney and each of Mrs. Murphy and Mrs. Rooney is entering into this Agreement to waive all rights they have with respect to the Units, the Businesses and the Companies, including all rights pursuant to the Wisconsin Marital Property Law, and for each of them to be jointly and severally liable with her respective spouse for all of such spouse's obligations under this Agreement; and

WHEREAS, the Buyer desires to purchase the Precision Units and the Transport Units from the Sellers and the Transport Sellers, respectively, and the Sellers and the Transport Sellers desire to sell all of the Precision Units and the Transport Units, respectively, to the Buyer upon the terms and subject to the conditions set forth in this Agreement.

AGREEMENT

NOW THEREFORE, in consideration of the premises and mutual promises herein made, and in consideration of the representations, warranties and covenants herein contained, the Buyer, MasTec, the Sellers, Mrs. Murphy and Mrs. Rooney hereby agree as follows:

1. PURCHASE AND SALE OF UNITS.

- 1.1 <u>Purchase and Sale of Units</u>. At the Closing, subject to the terms and conditions of this Agreement, the Sellers and the Transport Sellers will sell, transfer and deliver to the Buyer, and the Buyer will purchase from the Sellers and the Transport Sellers, the Precision Units and the Transport Units, respectively, free and clear of any and all Encumbrances other than restrictions on the transfer of securities arising under federal and state securities laws and Encumbrances created by the Buyer.
- 1.2 <u>Purchase Price</u>. The aggregate purchase price for the Units (the "<u>Purchase Price</u>") shall be paid to the Sellers and CAC as set forth on <u>Schedule 1.2</u> and is equal to:
 - 1.2.1 the Cash Purchase Price; plus
 - 1.2.2 the Earn-Out Payments; and
 - 1.2.3 Subject to adjustment as provided in this Section 1.
- 1.3 <u>Purchase Price Allocation</u>. The parties agree that the Contemplated Transactions will be governed by Revenue Ruling 99-6 for federal, state and local income tax purposes. The Purchase Price plus the liabilities of the Companies assumed, or taken subject to, by the Buyer (together with the Purchase Price, the "<u>Allocated Purchase Price</u>") shall be allocated among the Companies' Assets and the covenants set forth in <u>Section 5.2.1</u> in accordance with the Code (and any similar provision of state or local law, as appropriate) as set forth on <u>Schedule 1.3</u> (the "<u>Allocation</u>"), including those fair market values set forth in the appraisal attached to <u>Schedule 1.3</u>. Sellers and Buyer shall each report the federal, state, local and other Tax consequences of the Contemplated Transactions, including the preparation and filing of Form 8594, consistent with the Allocation. Buyer and Sellers agree that the Purchase Price reflects the fair market value of the Companies' Assets, and the Allocated Purchase Price will be allocated among the Assets and the covenants set forth in <u>Section 5.2.1</u> as described in this <u>Section 1.3</u>. Buyer and Sellers will act in good faith and reasonably cooperate with each other to agree on the final Allocation on or before the thirtieth (30th) day after the Closing Date consistent with <u>Schedule 1.3</u>, will cooperate with each other in the preparation of such forms, and will furnish each other with a copy of the final version of Form 8594 at least ten (10) days before the filing date thereof. Notwithstanding the foregoing, if the amount contained in the Allocation in respect of the covenants set forth in <u>Section 5.2.1</u> exceeds Five Hundred Thousand Dollars (\$500,000), then, on or prior to April 10, 2010, Buyer shall pay to Sellers, as set forth on <u>Schedule 1.2</u>, an amount equal to (x) the amount of such excess *multiplied* by (y) 0.265. For example: if One Million Dollars (\$1,000,000) is allocated to the covenants set forth in <u>Section 5.2.1</u>, then the amount of the excess would be Fiv

- 1.4 Estimate of Purchase Price. At least five (5) Business Days prior to the Closing Date, the Sellers shall prepare in accordance with Precision GAAP and deliver to Buyer the Sellers' good faith reasonable estimate of the Companies' consolidated balance sheet as of the Effective Date (the "Estimated Closing Balance Sheet") and a certificate (substantially in the form of Exhibit A) executed by both the chief executive officer and vice president of finance of each Company (the "Closing Certificate") setting forth Sellers' estimate of each of the following as set forth on the Estimated Closing Balance Sheet:
 - 1.4.1 all Debt (the "Estimated Indebtedness");
 - 1.4.2 the net book value of the PP&E (the "Estimated PP&E"); and
 - 1.4.3 Net Working Capital (the "Estimated Net Working Capital").

Following receipt of the Closing Certificate, Sellers shall permit Buyer and its Representatives at all reasonable times and upon reasonable notice to review the Sellers', the Companies' and their accountants' working papers relating to the Estimated Closing Balance Sheet and Closing Certificate as well as the Sellers' and the Companies' accounting books and records relating to the determination of the Estimated Closing Balance Sheet and Closing Certificate, and Sellers shall make reasonably available their Representatives responsible for the preparation of the Estimated Closing Balance Sheet and the Closing Certificate in order to respond to the inquiries of the Buyer. Prior to the Closing, the parties shall act reasonably in resolving in good faith any disagreements concerning the computation of any of the items on the Estimated Closing Balance Sheet and Closing Certificate; provided that it is acknowledged and agreed that if any disagreements cannot be resolved, then the Closing shall occur on the basis of the Closing Certificate provided by Sellers, and that any unresolved disagreements shall be deferred for resolution pursuant to the post-closing purchase price adjustment process described in Section 1.8.

- 1.5 <u>Payment of Closing Date Purchase Price</u>. At Closing, the Buyer shall:
 - 1.5.1 pay to the Sellers and CAC as set forth on Schedule 1.2 in cash (the "Cash Purchase Price") an amount equal to:
 - (a) One-Hundred-Fifty Million Dollars (\$150,000,000); minus
 - (b) if the Estimated PP&E is less than Forty-Four Million Dollars (\$44,000,000), the difference between Forty-Four Million Dollars (\$44,000,000) and the Estimated PP&E; minus
 - (c) the Estimated Indebtedness; provided that if the Estimated PP&E is in excess of Forty-Four Million Dollars \$44,000,000 (the amount of such excess, if any, an "Estimated PP&E Surplus"), then Estimated Indebtedness shall be deemed automatically reduced, without any action by the parties, by an amount equal to the lower of (x) the Estimated PP&E Surplus and (y) Twenty Million Dollars (\$20,000,000); minus

- (d) if the Estimated Net Working Capital is less than Nine Million Dollars (\$9,000,000), the difference between Nine Million Dollars (\$9,000,000) and the Estimated Net Working Capital; <u>plus</u>
- (e) if the Estimated Net Working Capital is greater than Nine Million Dollars (\$9,000,000), the difference between the Estimated Net Working Capital and Nine Million Dollars (\$9,000,000); and minus
 - (f) those Buyer's Expenses set forth on Schedule 1.5.1(f) as delivered by Buyer to the Sellers' Representative at or prior to Closing;

provided, however, that Buyer shall, out of the Cash Purchase Price to be paid to the Sellers, deposit in the Escrow Account on behalf of the Sellers pursuant to Section 1.6 below (such deposit, the "Escrow Fund"), comprising Fifteen Million Dollars (\$15,000,000) in cash from the Cash Purchase Price; and, provided further, Sellers authorize Buyer to pay out of the Cash Purchase Price all Closing Date Seller Expenses in accordance with such instruction as Sellers' Representative may give to Buyer and any Closing Date Seller Expenses so paid will not be reflected in the Actual Closing Balance Sheet or Actual Net Working Capital.

- 1.5.2 <u>Method of Cash Payments</u>. All cash payments made under this <u>Section 1.5</u> shall be made by wire transfer of immediately available funds to one or more accounts designated by the recipient in writing no fewer than two (2) Business Days immediately preceding the scheduled Closing Date.
- 1.6 Escrow. To provide for an escrow account or accounts to secure and to serve as a fund in respect of the indemnification obligations of Sellers under this Agreement, Buyer, Sellers and JP Morgan Chase Bank, National Association, as Escrow Agent, (the "Escrow Agent"), at Closing shall enter into an Escrow Agreement substantially in the form of Exhibit B (the "Escrow Agreement"). At the Closing, Buyer shall deposit the Escrow Fund with the Escrow Agent to be held in an account or accounts (the "Escrow Account"), which shall bear interest on the Escrow Fund, pursuant to the terms of the Escrow Agreement. Except with respect to amounts that have been previously paid from the Escrow Account to Buyer pursuant to the Escrow Agreement, and except with respect to claims pursuant to Section 1.8 and indemnity claims made in accordance with Section 9 or Section 10 on or before the eighteen (18) month anniversary of the Closing Date (the "Escrow Period"), all amounts (in excess of all pending claims) in the Escrow Account (with any interest or other earnings paid thereon) shall be distributed to Sellers in accordance with the Escrow Agreement on the first Business Day after the expiration of the Escrow Period; provided that with respect to any pending claim, promptly following resolution of such pending claim, the amount, if any, of such pending claim which has not been paid, which is not payable to any Buyer Indemnified Person pursuant to Section 1.8, Section 9.1 or Section 10 in connection with such resolution, and which is not required to remain in the Escrow Account to satisfy other pending claims shall be paid to the Sellers.
- 1.7 <u>Allocation of Payments</u>. As among Sellers and CAC, any Purchase Price payable to Sellers or CAC, payments payable to Sellers from the Escrow Account or otherwise payable to Sellers or CAC under this <u>Section 1</u>, shall be paid as set forth on <u>Schedule 1.2</u>.

1.8 Post-Closing Purchase Price Adjustment.

1.8.1 As soon as practicable but in no event more than one hundred twenty (120) days following the Closing Date, the Buyer shall prepare, or cause to be prepared, and deliver to the Sellers' Representative: (a) a consolidated balance sheet of the Companies as of the Effective Date prepared in accordance with Precision GAAP (the "Closing Balance Sheet"); and (b) a certificate substantially in the form of Exhibit C executed by MasTec's Chief Financial Officer (the "Final Closing Statement") setting forth each of the following as set forth on the Closing Balance Sheet: (i) the actual net book value of PP&E (the "Actual PP&E"); (ii) the actual aggregate amount of the Companies' Debt (the "Actual Indebtedness"); (iii) the actual Net Working Capital ("Actual Net Working Capital included in the Closing Balance Sheet which were not collected on or prior to the ninetieth (90th) day after the Closing Date (the "Actual Uncollected Receivables").

1.8.2 The Sellers and their accountants shall complete their review of the Closing Balance Sheet and Final Closing Statement within 30 days after delivery thereof by the Buyer. During such review period, Buyer shall provide Sellers' Representative with access to all books and records reasonably requested by Sellers' Representative to review the Closing Balance Sheet and the Final Closing Statement and any work papers prepared by Buyer or its accountants in connection with such calculations, and Buyer shall make reasonably available its Representatives responsible for the preparation of the Closing Balance Sheet and Final Closing Statement in order to respond to the inquiries of Sellers' Representative. If the Sellers object to the Closing Balance Sheet or Final Closing Statement for any reason, the Sellers' Representative shall, on or before the last day of such 30-day period, so inform the Buyer in writing (a "Sellers' Objection"), setting forth a specific description of the basis of the Sellers' determination and the adjustments to the Closing Balance Sheet or Final Closing Statement that the Sellers believe should be made. To the extent any disagreement therewith is not described in a Sellers' Objection received by the Buyer on or before the last day of such 30-day period, then all items described on the Closing Balance Sheet and Final Closing Statement delivered by the Buyer to the Sellers' Representative shall be deemed agreed, final and binding on the parties.

1.8.3 If the Sellers' Representative timely delivers a Sellers' Objection to the Buyer and the Sellers' Representative and the Buyer are unable to resolve all of their disagreements with respect to the proposed adjustments set forth in the Sellers' Objection within thirty (30) days following the Buyer's receipt of the Sellers' Objection, then they shall jointly retain the CPA Firm, which, acting as an expert and not as an arbitrator, shall determine, on the basis set forth in and in accordance with this Section 1.8, and only with respect to those items in the Sellers' Objection on which the Buyer and the Sellers' Representative have not agreed, whether and to what extent, if any, the Cash Purchase Price requires adjustment. The Buyer and the Sellers' Representative shall instruct the CPA Firm to deliver its written determination to the Buyer and the Sellers' Representative no later than 30 days after submitting the matter to it for resolution. At the time of retention of the CPA Firm, the Buyer shall specify in writing to the CPA Firm and the Sellers' Representative shall specify in writing to the CPA Firm and to the Buyer's computation of the Cash Purchase Price (the "Buyer's Position"), and the Sellers' Position"). The CPA Firm's determination shall be

conclusive and binding upon the Buyer and the Sellers. In resolving any disputed item, the CPA Firm may not assign a value to any disputed item that is greater than the greatest value claimed by the Buyer or the Sellers' Representative at the time the CPA Firm is retained or less than the smallest value claimed for the item by the Buyer or the Sellers' Representative at such time. The scope of the disputes to be resolved by the CPA Firm is limited to whether the preparation of the Closing Balance Sheet and the calculation of Actual PP&E, Actual Indebtedness and Actual Net Working Capital were done in a manner consistent with Precision GAAP and otherwise in accordance with this Agreement, and the CPA Firm is not to make any other determination unless jointly requested in writing by the Sellers' Representative and the Buyer. The fees and disbursements of the CPA Firm and the reasonable attorneys' fees and expenses of the parties relating to the disputes submitted to the CPA Firm (collectively, the "Purchase Price Dispute Expenses") shall be borne (i) jointly and severally by the Sellers in that proportion equal to a fraction (expressed as a percentage) the numerator of which is equal to Sellers' Position minus the Cash Purchase Price determined by the CPA Firm, and the denominator of which is equal to Sellers' Position minus Buyer's Position and (ii) by Buyer in that proportion equal to a fraction (expressed as a percentage) equal to one (1) minus the fraction described in clause (i). For example, if the Sellers' Position is that the Cash Purchase Price should be \$175,000,000 and the Buyer's Position is that the Cash Purchase Price should be \$174,000,000, the CPA Firm determines that the Cash Purchase Price should be \$174,600,000 and the Purchase Price Dispute Expenses are \$50,000, then (i) the Sellers shall pay \$20,000 (40%) of the Purchase Price Dispute Expenses and (ii) the Buyer shall pay \$30,000 (60%) of the Purchase Price Dispute Expenses. The Buyer and the Sellers shall cooperate with the CPA Firm during its resolution of the disagreement and make readily available to the CPA Firm all relevant books and records and any work papers (including those of the parties' respective accountants, to the extent permitted by such accountants) relating to the Closing Balance Sheet, Final Closing Statement and the Sellers' Objection and all other items reasonably requested by the CPA Firm in connection

- 1.8.4 The Closing Balance Sheet, PP&E, Debt and Net Working Capital, as agreed to (or deemed to have been agreed to) between Buyer and Sellers' Representative or as determined by the CPA Firm, as applicable, shall be conclusive and binding on all of the parties hereto and shall be deemed the "Actual Closing Balance Sheet", "Actual PP&E", "Actual Indebtedness", "Actual Net Working Capital" and "Actual Uncollected Receivables", respectively, for all purposes herein.
 - (a) Upon completion of the calculation of the Actual Closing Balance Sheet, Actual PP&E, Actual Indebtedness and Actual Net Working Capital in accordance with this <u>Section 1.8</u>, the Cash Purchase Price shall be recalculated, and the following adjustments (the "<u>Purchase Price Adjustment</u>") made:
 - (i) If the Cash Purchase Price calculated using the Actual PP&E, Actual Indebtedness and Actual Net Working Capital shown on the Actual Closing Balance Sheet is greater than the Cash Purchase Price calculated using the Estimated PP&E, Estimated Indebtedness and Estimated Net Working Capital shown on the Estimated Closing Balance Sheet, then the Buyer shall pay such difference to the Sellers and CAC as set forth on Schedule 1.2 by wire transfer in immediately available funds within two (2) Business Days thereafter to the accounts designated by the Sellers' Representative.

(ii) If the Cash Purchase Price calculated using the Actual PP&E, Actual Indebtedness and Actual Net Working Capital shown on the Actual Closing Balance Sheet is less than the Cash Purchase Price calculated using the Estimated PP&E, Estimated Indebtedness and Estimated Net Working Capital shown on the Estimated Closing Balance Sheet, then within (2) Business Days thereafter the amount of such difference will be (A) paid to the Buyer, jointly and severally by the Sellers, or (B) at Buyer's election, by withdrawal of such amount from the Escrow Fund (and in the case of clause (B), the Buyer and Sellers' Representative will, during such two (2) Business Day period, deliver to the Escrow Agent a joint written instruction to make such distribution).

1.9 Closing.

1.9.1 Closing Date. Subject to the terms and conditions of this Agreement, the sale and purchase of the Units contemplated by this Agreement shall take place at a closing (the "Closing" and the date on which the Closing actually occurs, the "Closing Date") to be held at the offices of Greenberg Traurig, P.A., located at 1221 Brickell Avenue, Miami, Florida 33131 at 10:00 a.m. Miami time, on the third (3rd) Business Day following the date on which all of the conditions set forth in Section 6 and Section 7 shall have been satisfied or waived (other than delivery of items to be delivered at the Closing and other than satisfaction of those conditions that by their nature are to be satisfied at the Closing, it being understood that the occurrence of the Closing shall remain subject to the delivery of such items and the satisfaction or waiver of such conditions at the Closing). Upon consummation, the Closing will be deemed for all purposes, to have taken place as of 12:01 a.m. on the Effective Date (except in the case of income Tax Returns, in which case, the Closing Date). By agreement of the Buyer and the Sellers' Representative, the Closing may take place by delivery of the items to be delivered at Closing by facsimile or other electronic transmission. Subject to the provisions of Section 8, failure to consummate the purchase and sale provided for in this Agreement on the date and time and at the place determined pursuant to this Section 1.9.1 will not result in the termination of this Agreement and will not relieve any party of any obligation under this Agreement. All deliveries by the parties at Closing shall be deemed to have occurred simultaneously, and none shall be effective until and unless all have occurred in accordance with this Agreement or have been waived.

1.9.2 <u>Deliveries at Closing</u>. At the Closing:

- (a) Deliveries by Sellers. Sellers will deliver, or cause to be delivered, to Buyer:
 - (i) the Escrow Agreement executed by Sellers;
 - (ii) assignments of the Units, duly endorsed by each Seller (as to such Seller's Units) for unconditional and irrevocable transfer to Buyer;

- (iii) a receipt for the Closing Date Purchase Price;
- (iv) a certificate signed by the Sellers, certifying to the fulfillment of the conditions specified in Section 6.1.1 and Section 6.1.2;
- (v) an affidavit from each Seller certifying the non-foreign status of each Seller and that no Seller is a disregarded entity, dated as of the Closing Date and in form and substance required under Section 1.1445-2(b)(2) of the Treasury Regulations;
- (vi) resignations of those officers and managers of the Companies (solely with respect to such offices and positions as managers, and not with respect to employment) as requested by Buyer at or at any time prior to Closing;
- (vii) executed payoff letters, releases, or other similar instruments providing for the repayment in full of all Debt of the Companies set forth on Schedule 1.9.2(a)(vii) and the release of all Encumbrances granted with respect thereto, together with all instruments, documents and UCC financing statements relating thereto;
- (viii) a written consent executed by all of the directors and shareholders of PPL authorizing its execution and delivery of this Agreement and performance of the Contemplated Transactions;
- (ix) a written consent executed by all of the members of Precision authorizing its execution and delivery of this Agreement and the performance of the Contemplated Transactions, including the admission of Buyer as the sole member of Precision, effective upon Closing;
- (x) a written consent executed by all of the members of Transport authorizing its execution and delivery of this Agreement and the performance of the Contemplated Transactions, including the admission of Buyer as the sole member of Transport, effective upon Closing;
- (xi) a compact disc (which shall be permanent and accessible, without the need for any password, with readily and commercially available software) containing, in electronic format, all documents posted to the datasite maintained by Merrill Corporation on behalf of the Company as of Closing (the "Data Room");
- (xii) a real estate lease for the Headquarters Facility (the "<u>Headquarters Lease</u>"), substantially in the form of <u>Exhibit D</u> between Precision and Precision Land Company, LLC, a Wisconsin limited liability company owned 50% by Murphy and 50% by Rooney ("LandCo") executed by Precision and LandCo;

- (xiii) a general release (a "<u>General Release</u>") substantially in the form of <u>Exhibit E</u> executed by each Seller, Mrs. Murphy and Mrs. Rooney;
- (xiv) the registration rights agreement substantially in the form of Exhibit F (the "Registration Rights Agreement"), executed by the Sellers and CAC;
- (xv) the employment agreements substantially in the form of <u>Exhibit G</u> between Precision and each of Murphy and Rooney (the "<u>Principals Employment Agreements</u>"), executed by the parties thereto;
 - (xvi) the representation letter, substantially in the form of Exhibit H, executed by CAC; and
 - (xvii) the representation letter substantially in the form of Exhibit I, executed by Gardere.
- (b) Deliveries by Buyer. Buyer will deliver:
 - (i) to the Sellers:
 - 1 the Escrow Agreement, executed by Buyer and MasTec;
 - 2 the Cash Purchase Price less the Escrow Fund;
 - 3 the Registration Rights Agreement, executed by MasTec;
 - 4 reasonable documentation of Buyer's Expenses; and
 - 5 a certificate signed by the Buyer, certifying to the fulfillment of the conditions specified in Section 7.1.1 and Section 7.1.2; and
 - (ii) to the Escrow Agent, the Escrow Fund.
- (c) <u>Other Action by Precision and PPL</u>. PPL shall pay all cash held by it on the Closing Date to Precision net of amounts necessary to satisfy all base salary and benefits and withholding Taxes payable to or in respect of Transferred Employees accrued for the period after the last payment therefor by PPL through and including the Closing Date and PPL shall thereafter make such payments to the appropriate payees in a timely fashion.

1.10 Earn-Out.

- 1.10.1 In addition to the Closing Date Purchase Price, and as additional consideration for the purchase of the Units, the Buyer shall pay to the Sellers and CAC as set forth on <u>Schedule 1.2</u>, contingent payments (the "<u>Earn-Out Payments</u>") as determined pursuant to this <u>Section 1.10</u> at the times, in the manner and to the extent Earn-Out Payments are earned pursuant to the following terms:
 - (a) with respect to the Companies' consolidated 2009 Earn-Out Period EBITDA, 40% of the Companies' consolidated 2009 Earn-Out Period EBITDA will be paid to the Sellers;
 - (b) with respect to the Companies' consolidated 2010 Earn-Out Period EBITDA, 30% of the amount (if any) by which the Companies' consolidated 2010 Earn-Out Period EBITDA exceeds \$35,000,000 will be paid to the Sellers;
 - (c) with respect to the Companies' consolidated 2011 Earn-Out Period EBITDA, 30% of the amount (if any) by which the Companies' consolidated 2011 Earn-Out Period EBITDA exceeds \$35,000,000 will be paid to the Sellers;
 - (d) with respect to the Companies' consolidated 2012 Earn-Out Period EBITDA, 30% of the amount (if any) by which the Companies' consolidated 2012 Earn-Out Period EBITDA exceeds \$35,000,000 will be paid to the Sellers;
 - (e) with respect to the Companies' consolidated 2013 Earn-Out Period EBITDA, 30% of the amount (if any) by which the Companies' consolidated 2013 Earn-Out Period EBITDA exceeds \$35,000,000 will be paid to the Sellers; and
 - (f) with respect to the Companies' consolidated 2014 Earn-Out Period EBITDA, 30% of the amount (if any) by which the Companies' consolidated 2014 Earn-Out Period EBITDA exceeds \$29,166,667 will be paid to the Sellers.
 - (g) Notwithstanding the foregoing, if the Companies' consolidated EBITDA for any Earn-Out Period is negative (any such Earn-Out Period, a "Negative EBITDA Earn-Out Period"), taking into account the adjustment to such EBITDA resulting from a prior Negative EBITDA Earn-Out Period as set forth in this subsection 1.10.1(g), the Companies' consolidated EBITDA for the subsequent Earn-Out Period shall be reduced by the amount by which EBITDA for the Negative EBITDA Earn-Out Period was less than zero. For example, if the 2009 Earn-Out Period is a Negative EBITDA Earn-Out Period in which the Companies' consolidated EBITDA is negative \$1 million and if the 2010 Earn-Out Period is a Negative EBITDA Earn-Out Period in which the Companies' consolidated EBITDA (excluding any reduction for the negative EBITDA for the 2009 Earn-Out Period) is negative \$2 million, then the Companies' consolidated EBITDA for the 2011 Earn-Out Period shall be reduced by \$3 million as a result of the negative \$2 million of EBITDA in the 2010 Earn-Out Period.

1.10.2 For purposes of this <u>Section 1.10</u>, "<u>Earn-Out Period</u>" shall mean the period commencing on January 1 and ending of December 31 of the relevant year, provided, that (i) the "<u>Earn-Out Period</u>" for 2009 shall commence on the Effective Date and end on December 31, 2009, and (ii) the "<u>Earn-Out Period</u>" for 2014 shall commence on January 1, 2014 and end one day prior to the five year anniversary of the Effective Date.

1.10.3 For purposes of this Section 1.10, "EBITDA" means for any Earn-Out Period the sum of (a) the consolidated net income or loss generated by the Companies' consolidated operations as they exist on the Effective Date and as they may grow or contract after the Effective Date in the normal course of business through internal growth and acquisitions made by Buyer or Precision with the prior written consent of Sellers' Representative ("Approved Acquisitions") and not through acquisitions other than Approved Acquisitions (which acquisitions which are not Approved Acquisitions the Buyer and its Affiliates may make in their sole discretion so long as such acquisitions are kept separate from the Companies) or other extraordinary transactions (the "Existing Operations"), as reflected in the Companies' consolidated EO Financial Statements for the Earn-Out Period, after deduction (to the extent included in the consolidated net income of the Companies and without duplication) of all costs, expenses, interest, taxes, depreciation, amortization, and other proper charges (including (i) the cost of any bonuses or incentive payments earned with respect to the applicable period (excluding any Earn-Out Payment), (ii) the cost of any equity granted to employees of the Companies, or other derivative securities granted during the applicable period, valued in accordance with Statement of Financial Accounting Standards No. 123(R) and (iii) all interest and amortization with respect to the Companies' Debt, including all Debt reflected on the Closing Date Balance Sheet) plus (b) all (i) interest expense (including imputed interest in capital lease payments), (ii) federal, state, or local income taxes and state franchise taxes to the extent based upon net income and not revenue (including any payments under taxsharing arrangements with the Buyer or its Affiliates), (iii) depreciation expense, (iv) amortization expense and (v) any non-cash impairment charges, in each case of (i) through (v) as attributable to the Existing Operations, but only to the extent deducted in determining the amount in clause (a). For the determination of EBITDA, the Buyer will cause to be prepared for each Earn-Out Period a set of financial statements reflecting the consolidated financial position and the consolidated results of operations ("EO Financial Statements"). Each of the EO Financial Statements shall be prepared in accordance with Precision GAAP and with (A) no deduction of any expenses relating to or arising out of this Agreement or the Contemplated Transactions (including any fees and expenses related to the financing of the Contemplated Transactions or any amortization of goodwill), (B) no deduction for any management fees or general overhead expenses, (C) any purchases of any goods or services from either or both of the Companies by the Buyer or any of its Affiliates to be adjusted to reflect the lowest amount that the Companies would have realized if dealing with an independent party in an arm's-length commercial transaction at the best price and best other terms (from the customer's perspective) for such goods or services provided to any Person by the Companies (D) any costs for goods, services or benefits (including insurance and employee-welfare benefits) provided to the Companies by the Buyer or any Affiliate of the Buyer to not exceed the Buyer's or such Affiliate's actual cost of furnishing such goods, services or benefits and (E) no deduction for any Loss to the extent that the Buyer has received indemnification from the Sellers pursuant to this Agreement and which payment is not subject to any dispute.

1.10.4 The Buyer will pay to the Sellers the Earn-Out Payment with respect to a specific Earn-Out Period, if any Earn-Out Payment is owed, no later than the March 31st next succeeding such Earn-Out Period (or ninety (90) days after the 2014 Earn-Out Period). If Sellers' Representative delivers an Earn-Out Objection pursuant to Section 1.10.6 and it is ultimately determined pursuant to Section 1.10.6 that an additional amount for a particular Earn-Out Period is payable to Sellers, Buyer shall, within three (3) Business Days after the CPA Firm's determination is delivered to the Buyer and the Sellers' Representative, pay the such additional amount. Each Earn-Out Payment will be made, at the Buyer's option (i) in cash in immediately available funds or (ii) in the common stock, par value \$0.10 per share, of MasTec Common Stock"), valued at the average closing price of the MasTec Common Stock as reported by the New York Stock Exchange (or as quoted on such other principal U.S. exchange or over-the-counter market on which MasTec Common Stock is then traded) for the ten (10) trading days prior to the applicable payment date or (iii) in any combination of cash or MasTec Common Stock. If Buyer issues MasTec Common Stock for some, or any portion, of any Earn-Out Payment, such shares, at the time of their issuance, will be: (i) validly authorized shares and (ii) registered for resale pursuant to an effective registration statement that has been filed with the SEC which registration statement will remain effective as set forth in the Registration Rights Agreement.

1.10.5 Except as set forth in <u>Section 1.10.11</u>, the Sellers will be entitled to the Earn-Out Payments, if earned, whether or not Rooney or Murphy remain employed by the Buyer or its Affiliates, including the Companies following the Closing; provided that nothing in this Agreement (other than, for clarity, the Principals Employment Agreements attached hereto) constitutes an agreement or understanding to employ Rooney or Murphy for a term of years or otherwise guarantee Rooney's or Murphy's employment with the Buyer or any of its Affiliates, including the Companies following the Closing.

1.10.6 On or prior to March 31st of each calendar year immediately following the expiration of each Earn-Out Period (other than with respect to the 2014 Earn-Out Period, which will be calculated on or before the last day of the third month following the last day of the 2014 Earn-Out Period), the Buyer will calculate the Earn-Out Payment, if any, for such period and provide the Sellers' Representative with the Buyer's calculation of the Earn-Out Payment and a copy of the EO Financial Statements for the Earn-Out Period. For sixty (60) days following Buyer's delivery of the Earn-Out Payment calculation to the Sellers' Representative (the "Earn-Out Review Period"), the Sellers may review the Earn-Out Payment calculation and other documentation. During the Earn-Out Review Period, the Buyer shall provide Sellers' Representative with access to all books and records and any work papers utilized by Buyer or its accountants in preparing the EO Financial Statements and calculating the proposed Earn-Out Payments, and to the Buyer's personnel involved in preparing the EO Financial Statements and calculating the proposed Earn-Out Payments, as reasonably requested to review the Buyer's calculation of the Earn-Out Payment. If the Sellers object to the amount of the Earn-Out Payment for any reason, the Sellers' Representative shall, on or before the last day of the Earn-Out Review Period, so inform the Buyer, in writing (an "Earn-Out Objection"), setting forth a

specific description of the basis of the Sellers' objection and the adjustments to the Earn-Out Payment that the Sellers believe should be made. To the extent any disagreement therewith is not described in an Earn-Out Objection received by the Buyer on or before the last day of the Earn-Out Review Period, then the Earn-Out Payment calculation prepared by the Buyer and delivered to the Sellers' Representative shall be deemed agreed, final and binding on the parties. If the Sellers' Representative timely delivers an Earn-Out Objection to the Buyer, and the Buyer and the Sellers' Representative are unable to resolve all of their disagreements with respect to the Earn-Out Payment within thirty (30) days after the end of the Earn-Out Review Period, then the Buyer and the Sellers' Representative shall jointly retain the CPA Firm, which, acting as an expert and not as an arbitrator, shall determine, on the basis set forth in and in accordance with this Section 1.10, and only with respect to those specific adjustments in the Earn-Out Objection on which the Buyer and the Sellers' Representative have not agreed, whether and to what extent, if any, the Earn-Out Payment amount requires adjustment. The Buyer and the Sellers' Representative shall instruct the CPA Firm to deliver its written determination to the Buyer and the Sellers' Representative no later than 30 days after submitting the matter to it for resolution. At the time of retention of the CPA Firm, Buyer shall specify in writing to the CPA Firm and the Sellers' Representative the amount of Buyer's computation of the Earn-Out Payment (the "Buyer's Earn-Out Position"), and the Sellers' Representative shall specify in writing to the CPA Firm and to the Buyer the amount of the Sellers' computation of the Earn-Out Payment (the "Sellers' Earn-Out Position"). The CPA Firm's determination shall be conclusive and binding upon the parties. In resolving any disputed item, the CPA Firm may not assign a value to any disputed item that is greater than the greatest value claimed by Buyer or the Sellers' Representative at the time the CPA Firm is retained or less than the smallest value claimed for the item by the Buyer or the Sellers' Representative at such time. The scope of the disputes to be resolved by the CPA Firm is limited to whether the calculation of the Earn-Out Payment was done in a manner consistent with Precision GAAP (as modified by Section 1.10.3) and in accordance with this Agreement, and the CPA Firm is not to make any other determination unless jointly requested in writing by the Sellers' Representative and Buyer. The fees and disbursements of the CPA Firm and the reasonable attorneys' fees and expenses of the parties relating to the disputes submitted to the CPA Firm (collectively, the "Earn-Out Dispute Expenses") shall be borne (i) jointly and severally by the Sellers in that proportion equal to a fraction (expressed as a percentage) the numerator of which is equal to Sellers' Earn-Out Position minus the Earn-Out Payment determined by the CPA Firm, and the denominator of which is equal to Sellers' Earn-Out Position minus Buyer's Earn-Out Position and (ii) by Buyer in that proportion equal to a fraction (expressed as a percentage) equal to one (1) minus the fraction described in clause (i). For example, if the Sellers' Earn-Out Position is that the Earn-Out Payment should be \$5,000,000 and the Buyer's Earn-Out Position is that the Earn-Out Payment should be \$4,000,000, the CPA Firm determines that the Earn-Out Payment should be \$4,600,000 and the Earn-Out Dispute Expenses are \$50,000, then (i) the Sellers shall pay \$20,000 (40%) of the Earn-Out Dispute Expenses and (ii) the Buyer shall pay \$30,000 (60%) of the Earn-Out Dispute Expenses. The Buyer and the Sellers shall cooperate with the CPA Firm during its resolution of the disagreement and make readily available to the CPA Firm all relevant books and records and any work papers (including those of the parties' respective accountants, to the extent permitted by such accountants) relating to the determination of the Earn-Out Payment and all other items reasonably requested by the CPA Firm in connection therewith.

1.10.7 Following the Closing Date, the Buyer will continue to operate the Companies during the period for which the Sellers are entitled to Earn-Out Payments substantially as previously operated, as one or more separate entities that continue to own and conduct the Existing Operations, subject to the business requirements of the Buyer and its Affiliates taken as a whole. The Buyer will be permitted, following the Closing Date, to make changes at its sole discretion to the operations, corporate organization, personnel, accounting practices, and other aspects of the Companies and the Businesses so long as such changes are made in good faith, in the best interests of the Companies, the Buyer, and their respective Affiliates or to conform to standard practices applicable generally to other Buyer Affiliates, and not with the specific intent of reducing amounts that otherwise would be payable to the Sellers.

1.10.8 By no later than November 30th of each calendar year containing an Earn-Out Period, the Companies and the Sellers' Representative will prepare and present to MasTec's Chief Operating Officer an annual budget along with the proposed capital allocation requirements (collectively, the "Annual Budget") for the next succeeding year. MasTec's Chief Operating Officer will be entitled to review and propose revisions to the Annual Budget, and, subject to the Sellers' Representative's acceptance of the proposed revisions, the Companies shall operate in accordance with such revised Annual Budget (the "Revised Annual Budget"). If the Sellers' Representative has a reasonable basis to believe that the proposed Revised Annual Budget is insufficient to allow the Companies to pursue a material portion of identified Precision Business ("Identified Excluded Business"), then the Sellers' Representative may provide projections for such Identified Excluded Business, together with the corresponding additional capital allocations omitted from the proposed Revised Annual Budget, to MasTec's Chief Executive Officer for review. MasTec's Chief Executive Officer will review Sellers' Representative's projections for such Identified Excluded Business, together with the proposed additional capital allocations, and will determine whether such Identified Excluded Business, together with the corresponding increase in capital allocations, should be incorporated as part of the Revised Annual Budget. If Sellers' Representative is dissatisfied with the Chief Executive Officer's proposed revisions to the Revised Annual Budget, taking into consideration such Identified Excluded Business and the corresponding increase in capital allocations requested by Sellers' Representative, then upon the Sellers' Representative's request, the Chief Executive Officer shall submit his determination, Sellers' Representative's projections for such Identified Excluded Business and the corresponding increase in capital allocations related thereto to MasTec's Board of Directors, or to the appropriate committee thereof, (the "Board"), for its consideration. The Board shall have an opportunity to review the Annual Budget, the proposed Revised Annual Budget and to receive a presentation by Sellers' Representative, setting forth the reasons why the Sellers disapprove of the proposed Revised Annual Budget. The determination of the Board, which shall include its requested revisions to the Revised Budget, shall become the approved Annual Budget (the "Approved Annual Budget") and shall be accepted, final and binding upon MasTec's Chief Executive Officer and Chief Operating Officer and the Sellers.

1.10.9 <u>Affiliated Transactions</u>. From the Effective Date through the last day of the 2014 Earn Out Period, Buyer shall permit Precision to continue conducting business with K&M Equipment, Inc., an Illinois corporation, Midwest Underground, Inc., a Wisconsin corporation, and Indianhead Pipeline Services, LLC, a Wisconsin limited liability company, provided that such Persons provide services to Precision (i) at prices less than or equal to the lowest bid submitted to Precision by an unaffiliated third Person and (ii) in a manner consistent with or comparable to similar services provided by unaffiliated third Persons.

1.10.10 <u>Limited Transferability of Earn-Out</u>. Without the Buyer's prior written consent, which may be withheld in Buyer's sole discretion, the Sellers' and CAC's rights to receive Earn-Out Payments, if any, are personal to the Sellers and CAC and shall not, in whole or in part, be sold, transferred, assigned or otherwise conveyed or pledged or encumbered or have Encumbrances created with respect to them; <u>provided</u>, <u>however</u>, that a Seller may transfer such right to a Member of the Immediate Family of such Seller, in each case subject to the transferee's agreeing in writing to the terms and conditions of this <u>Section 1.10</u> and with a copy of such written agreement being promptly provided to Buyer.

1.10.11 Forfeiture of Earn-Out Payments. It is the intent of the parties that Earn-Out Payments will be paid to Sellers and CAC only for such periods during which Sellers are in compliance with their obligations pursuant to Section 5.2. Accordingly, should at any time following the Closing Date any Seller breach its obligations pursuant Section 5.2 (a "Restrictive Covenant Breach"), then during the continuance of such Restrictive Covenant Breach (the "Noncompliance Period"), no Earn-Out Payments shall be earned by or payable to Sellers or CAC. If a Restrictive Covenant Breach is cured, then Buyer's obligation to make Earn-Out Payments in accordance with Section 1.10 shall resume commencing on the next succeeding day, but no amounts shall be payable with respect to the Noncompliance Period. If a Noncompliance Period covers less than the whole of one or more Earn-Out Periods (each such Earn-Out Period, a "Hybrid Earn-Out Period"), then the amount of the Earn-Out Payments for the portion of each Hybrid Earn-Out Period which is not included in the Noncompliance Period (the "Compliance Period") shall be determined as set forth in Section 1.10.1 without amendment, except that only the Companies' consolidated EBITDA generated during the Compliance Period shall be counted and all EBITDA generated during the Noncompliance Period shall be disregarded. If MasTec's Chief Executive Officer, Chief Operating Officer or General Counsel has actual knowledge of any Restrictive Covenant Breach he shall promptly cause Buyer and/or MasTec to give notice to the Sellers' Representative of such Restrictive Covenant Breach.

1.11 Collection and Assignment of Actual Uncollected Receivables. If at any time prior to the one (1) year anniversary of the Closing Date any Actual Uncollected Receivables shall be collected by either Company, then the relevant Company shall pay to Sellers and CAC as set forth on Schedule 1.2 an amount equal to such collections less all reasonable collection costs related to all Actual Uncollected Receivables which have not been previously deducted from any payment pursuant to this Section 1.11. If upon the one (1) year anniversary of the Closing Date any Actual Uncollected Receivables remain uncollected, and Rooney or Murphy continues to be employed by Precision or any of its Affiliates on such date, then the relevant Company shall assign all of its rights to such Actual Uncollected Receivables to the Sellers' Representative. If neither Rooney nor Murphy is employed by Precision or any of its Affiliates at such time, then the relevant Company shall hold such Actual Uncollected Receivables and, if collected, will pay to Sellers and CAC as set forth on Schedule 1.2 such amounts collected less all reasonable collection costs. Notwithstanding the foregoing, the Companies shall make such efforts to collect such Actual Uncollected Receivables only as either such Company deems reasonable in its discretion, taking into account the detrimental impact such collection efforts could have on the Businesses.

1.12 Reimbursement for and Assignment of Uncollected Retainage. On or after March 31, 2011 (the "<u>Uncollected Retainage Determination Date</u>"), Buyer shall determine the amount of retainage included in Actual Net Working Capital not collected by the Companies on or prior to the Uncollected Retainage Determination Date ("<u>Uncollected Retainage</u>"). On or prior to the fifth (5th) day following delivery by Buyer to the Sellers' Representative of notice of the Uncollected Retainage, the Sellers, jointly and severally, shall pay to Buyer an amount equal to the Uncollected Retainage. If, upon the Uncollected Retainage Determination Date, Rooney or Murphy continues to be employed by Precision or any of its Affiliates, then the relevant Company shall assign all of its rights to such Uncollected Retainage to the Sellers' Representative. If neither Rooney nor Murphy is employed by Precision or any of its Affiliates at such time, then the relevant Company shall hold such Uncollected Retainage and, if collected, will pay to Sellers as set forth on <u>Schedule 1.2</u> such amounts collected less all reasonable collection costs. Notwithstanding the foregoing, the Companies shall make such efforts to collect such Uncollected Retainage only as either such Company deems reasonable in its discretion, taking into account the detrimental impact such collection efforts could have on the Businesses.

2. REPRESENTATIONS AND WARRANTIES OF THE SELLERS

In order to induce the Buyer to enter into and perform this Agreement and to consummate the Contemplated Transactions, the Sellers hereby jointly and severally represent and warrant to the Buyer as follows as of the date hereof (except when an earlier date is specified):

2.1 Organization; Predecessors.

- 2.1.1 Organization. Each Company is a limited liability company duly organized, validly existing and in good standing under the Laws of the State of Wisconsin with full power and authority to carry on its business as it is now being conducted and to own, operate and lease its properties and assets. Each Company is duly qualified or licensed as a foreign limited liability company to do business and is in good standing in every jurisdiction in which the conduct of its business, and the ownership or lease of its properties, require it to be so qualified or licensed, except for failures to be so qualified or licensed and in good standing that do not have a Material Adverse Effect; all of such jurisdictions are listed on Schedule 2.1.1. Except as set forth on Schedule 2.1.1, neither Company has any Subsidiaries, holds any Investments or has any obligation to make any Investments in any Person. PPL is a corporation duly incorporated, validly existing and in good standing under the Laws of the State of Wisconsin. Contained in the Data Room are true, accurate and complete copies of (x) the Organizational Documents of the Companies and PPL and (y) except as set forth on Schedule 2.1.1, the minute books of the Companies and PPL which contain records of all meetings held of, and other actions taken by, their members, managing members shareholders, board of directors and any committees appointed by its members, managing members, board of directors or shareholders for which minutes were taken and recorded.
- 2.1.2 <u>Predecessors. Schedule 2.1.2</u> sets forth a true, accurate and complete list of (a) any Person that has ever merged with or into either Company, (b) any Person a majority of whose capital stock (or similar outstanding ownership interests) has ever been acquired by either Company, (c) any Person all or substantially all of whose assets have ever been acquired by either Company and (d) any prior names of the Companies or any Person described in clauses (a) through (c) (each such Person, a "<u>Predecessor</u>").

2.2 Capitalization of the Companies; Title to Units.

- 2.2.1 <u>Units</u>. The Precision Units are the only outstanding Equity Securities of Precision. The Transport Units are the only outstanding Equity Securities of Transport. All of the Units have been issued and paid for in accordance with the terms of the Organizational Documents of Precision and Transport, as the case may be. The Units are not subject to, nor were any of them issued in violation of, any purchase option, call option, right of first refusal or offer, preemptive right, subscription right or any similar right. Neither Company has violated the 1933 Act, any state "blue sky" or securities laws, any other similar Legal Requirement or any preemptive or other similar rights of any Person in connection with the issuance or redemption of any Units or other Equity Securities. Except the Units, there are no bonds, debentures, notes, other Debt or Equity Securities of either Company with voting rights on any matters on which members of either Company may vote.
- 2.2.2 <u>Ownership</u>. The Units are held of record and beneficially owned by the Sellers free and clear of all Encumbrances, other than restrictions on the transfer of securities arising under federal and state securities laws, as set forth on <u>Schedule 2.2.2</u>. The Sellers have the full right, power and authority to transfer and deliver to the Buyer valid title to the Units, free and clear of all Encumbrances, other than restrictions on the transfer of securities arising under federal and state securities laws and Encumbrances created by the Buyer. Immediately following the Closing, the Buyer will be the record and beneficial owner of the Units, and will have good and marketable title to the Units, free and clear of all Encumbrances, other than restrictions on the transfer of securities arising under federal and state securities laws and Encumbrances created by the Buyer. The assignments, endorsements, membership interest powers and other instruments of transfer delivered by the Sellers to the Buyer at Closing will be sufficient to transfer the Sellers' entire interest, legal and beneficial, in the Units to the Buyer. The Companies do not maintain membership interest ledgers (or similar registers).
- 2.2.3 Encumbrances, etc. There are no outstanding agreements to which either Company is a party or by which it is bound obligating such Company to issue, deliver or sell, or cause to be issued, delivered or sold Equity Securities of such Company. There are no outstanding obligations of either Company (contingent or otherwise) to repurchase, redeem or otherwise acquire any Units or other Equity Securities of such Company. There are no unit-appreciation rights, membership interest-based performance units, "phantom" membership interest unit rights or other Contractual Obligations (contingent or otherwise) pursuant to which any Person is or may be entitled to receive any payment or other value based on the revenues, earnings or financial performance or other attribute of either Company or either Business or the Assets or calculated in accordance therewith or to cause either Company to file a registration statement under the 1933 Act, or which otherwise relate to the registration of any securities of either Company. Except as set forth on Schedule 2.2.3, there are no voting trusts, proxies or other Contractual Obligations to which either Company or any Seller is a party or by which any of them is bound with respect to the issuance, holding, acquisition, voting or disposition of any Units of or other membership or similar interests in either Company. Except as set forth on Schedule 2.2.3, there are no existing Contractual Obligations between any Seller on the one hand, and any other Person, on the other hand, regarding the Units.

- 2.2.4 <u>Debt; Guarantees</u>. The Companies have no Debt except as set forth on <u>Schedule 2.2.4</u>. For each item of Debt, <u>Schedule 2.2.4</u> correctly sets forth the debtor, the principal amount of the Debt as of September 30, 2009 or the remaining lease payments as of such date, the creditor, the maturity date and the collateral, if any, securing the Debt. Except as set forth on <u>Schedule 2.2.4</u>, neither Company has any Liability in respect of a Guarantee of any Liability of any other Person other than the other Company.
- 2.3 <u>Organization, Power and Authorization</u>. Each Seller and each Company has all requisite power and authority to execute and deliver this Agreement and each other agreement, document, instrument or certificate contemplated by this Agreement or to be executed by any Seller or any Company in connection with the consummation of the Contemplated Transactions (the "<u>Company Agreements</u>") and to consummate the Contemplated Transactions. The execution and delivery by each of PPL and the Companies of this Agreement and the Company Agreements and the consummation by each of PPL and the Companies of the Contemplated Transactions have been authorized by its respective board of directors and shareholders or members, as the case may be, and no other authorization on the part of PPL or the Companies or any other Person is necessary to authorize the execution and delivery of this Agreement and the Company Agreements by PPL or the Companies or the consummation by PPL or the Companies of the Contemplated Transactions. This Agreement has been and each Company Agreement will be at or prior to Closing duly executed and delivered by the Sellers and the Companies, and this Agreement constitutes and the Company Agreements will constitute legal, valid and binding obligations of the Sellers and the Companies, Enforceable against the Sellers and the Companies in accordance with its and their terms, except as limited by the Enforceability Exceptions.
- 2.4 <u>Authorization of Governmental Authorities</u>. Except for compliance with the HSR Act and as disclosed on <u>Schedule 2.4</u>, no action by (including any authorization, consent or approval), or in respect of, or filing with, any Governmental Authority is required for, or in connection with, the valid and lawful (a) execution, delivery and performance by the Sellers and the Companies of this Agreement and the Company Agreements or (b) the consummation of the Contemplated Transactions by the Sellers and the Companies.
- 2.5 <u>Noncontravention</u>. Except as disclosed on <u>Schedule 2.5</u>, none of the execution, delivery and performance by the Sellers or the Companies of this Agreement or the Company Agreements nor the consummation of the Contemplated Transactions will:
 - (a) violate any provision of any Legal Requirement applicable to the Sellers or either Company;
 - (b) conflict with, result in a material breach or violation of, default under, or give rise to a right for any third-party to accelerate, terminate or obtain any prepayment penalty under (in any such case, with or without notice, lapse of time or both) any Contractual Obligation of any Seller or either Company;

- (c) result in the creation or imposition of an Encumbrance upon, or the forfeiture of, any Asset, other than any Encumbrances created by the Buyer;
 - (d) result in a breach or violation of, or default under, the Organizational Documents of either Company or PPL; or
- (e) require any action by (including any authorization, consent or approval) or in respect of (including notice to), any Person under any Contractual Obligation of either Company or the Sellers.

2.6 Financial Statements.

2.6.1 Financial Statements. Attached to Schedule 2.6.1 is a copy of the audited balance sheets of the Companies, LandCo and PPL as of December 31, 2008, 2007 and 2006 and the related audited statements of income and cash flows of the Companies, LandCo and PPL for the fiscal years then ended, which audited statements are accompanied in each case by any notes thereto and the report of the independent registered certified public accounting firm (collectively, the "Audited Financials", and together with the Interim Financials and the other financial statements provided pursuant to Section 4.9, the "Financials"). Except as set forth on Schedule 2.6.1, since September 30, 2009 (the "Most Recent Balance Sheet Date"), (i) neither Company has distributed, sold or otherwise disposed of any property or other Assets other than in the Ordinary Course of Business, and (ii) neither Company has made or granted any cash or non-cash distributions or dividends. Except as set forth on Schedule 2.6.1, there are no extraordinary items of income or expense of the Companies, LandCo or PPL during the periods covered by the Financial Statements.

2.6.2 Compliance with Precision GAAP, etc.

(a) Except as set forth on Schedule 2.6.2, the Financials have been prepared in accordance with Precision GAAP, consistently applied (subject, in the case of Interim Financials, to normal year-end audit adjustments, the effect of which will not, individually or in the aggregate, be materially adverse and the absence of footnotes that, if presented, would not differ materially from those included in the Audited Financial Statements). The Financials fairly present, in all material respects, in accordance with Precision GAAP the financial position of the Companies, LandCo and PPL as at the respective dates thereof and the results of the operations of the Companies, LandCo and PPL for the respective periods covered thereby. Except as set forth on Schedule 2.6.2, Precision's utilization of the percentage of completion methodology for the recognition of revenues is in conformity with Precision GAAP. Except as set forth on Schedule 2.6.2, the gross profit and gross profit percentage of jobs currently in process have been estimated in good faith in the Ordinary Course of Business and without any intent to shift profits or revenues from the period in which they should be recognized in accordance with GAAP. Except as set forth on Schedule 2.6.2, the gross profit and gross profit percentage of all jobs currently in process have been properly recognized on a collective basis in accordance with Precision GAAP. Precision maintains adequate internal controls to ensure the proper recording of all assets, Liabilities and transactions in Precision's records and books of account and to safeguard the Company's assets.

- (b) Except as set forth on <u>Schedule 2.6.2</u>, the books of account, minute books, member record books, and other records of each Company have been maintained in accordance with sound business practices since such Company's formation. At the Closing, all of those books and records of each Company will be in the possession, or under the control of, such Company.
- 2.7 <u>Absence of Undisclosed Liabilities</u>. Except as set forth on <u>Schedule 2.7</u>, the Companies have no Liabilities except for (a) Liabilities reflected or reserved in the Most Recent Balance Sheet or described in the notes to the Audited Financials, (b) Liabilities incurred in the Ordinary Course of Business since the Most Recent Balance Sheet Date, (c) Liabilities under any of the transaction documents relating to any of the Contemplated Transactions and (d) Liabilities not required to be presented on a balance sheet prepared in accordance with GAAP.
- 2.8 <u>Absence of Certain Developments</u>. Except as set forth on <u>Schedule 2.8</u>, since the Most Recent Balance Sheet Date through the date hereof, the Businesses have been conducted only in the Ordinary Course of Business in all material respects and:
 - (a) neither Company has (i) amended its Organizational Documents, (ii) admitted any Person as a member or (iii) issued, sold, granted or otherwise disposed of any Equity Security;
 - (b) neither Company has become liable in respect of any Guarantee nor has it incurred, assumed or otherwise become liable in respect of any Debt or made any loans, advances or capital contributions to or Investments in any Person (except for travel advances in the Ordinary Course of Business);
 - (c) neither Company has sold, leased, licensed, transferred or otherwise disposed of any of its Assets, except Inventory and excess or obsolete PP&E, which has been replaced by all substitute PP&E necessary to continue to operate the Businesses in the Ordinary Course of Business;
 - (d) neither Company has permitted any of its Assets to become subject to an Encumbrance other than a Permitted Encumbrance;
 - (e) neither Company has made or committed to make any capital expenditure except expenditures, which in the aggregate are no more than Two Hundred Fifty Thousand Dollars (\$250,000) and which were incurred in the Ordinary Course of Business;
 - (f) except for cash distributions identified on <u>Schedule 2.6.1</u>, neither Company has (i) made any declaration, setting aside or payment of any distribution with respect to, or any repurchase, redemption or other acquisition of, any Equity Security or (ii) entered into, or performed, any transaction with, or for the benefit of, the Sellers or any Affiliate of the Sellers;

- (g) there has been no material loss, destruction, damage or eminent domain taking (in each case, whether or not insured) affecting any Business or any material Asset;
- (h) neither Company has increased the Compensation payable or paid, whether conditionally or otherwise, to (i) any employee, consultant, independent contractor or agent other than in the Ordinary Course of Business, (ii) any managing member of such Company or (iii) the Sellers or any Affiliate of the Sellers;
- (i) neither Company has entered into any Contractual Obligation providing for the employment or consultancy of any Person on a full-time, part-time, consulting or other basis other than in the Ordinary Course of Business or otherwise providing Compensation or other benefits to any Person other than in the Ordinary Course of Business and other than the retention of Gardere Wynne Sewell LLP ("<u>Gardere</u>");
- (j) neither Company has made any change in its methods of accounting or accounting practices (including with respect to reserves) or its pricing policies, payment or credit practices or failed to pay any creditor any amount owed to such creditor when due or granted any extensions of credit other than in the Ordinary Course of Business, except such creditors, if any, as listed on <u>Schedule 2.21</u>, the claims of which are being contested in good faith;
- (k) neither Company has made, changed or revoked any material Tax election, elected or changed any method of accounting for Tax purposes, settled any Action in respect of Taxes or entered into any Contractual Obligation in respect of Taxes with any Governmental Authority;
 - (l) neither Company has terminated or closed any Facility, business or operation;
- (m) no customer or supplier required to be disclosed on Schedule 2.21 has canceled, terminated or otherwise altered (including any reduction in the rate or amount of sales or purchases or change to the supply or credit terms, as the case may be) or notified either Company or any Seller of any intention to do any of the foregoing (other than change orders, which are immaterial to the Companies' revenue from the project to which they relate or upon completion of the project or projects pursuant to the terms of existing Contractual Obligations) or otherwise threatened in writing to cancel, terminate or materially alter (including any reduction in the rate or amount of sales or purchases, as the case may be) its relationship with either Company;
- (n) no insurer (i) has questioned, denied or disputed (or otherwise reserved its rights with respect to) the coverage of any claim pending under any Liability Policy or (ii) has provided any notice of cancellation or any other indication that it plans to cancel any Liability Policy or raise the premiums or materially alter the coverage under any Liability Policy;

- (o) neither Company has adopted any Employee Plan or increased any benefits under any Employee Plan;
- (p) except as contemplated pursuant to this Agreement, neither Company has modified or cancelled any material third-party Debt or written up or written down any of its material Assets or revalued its Inventory;
- (q) neither Company has failed to make any scheduled capital expenditures or investments or failed to pay trade accounts payable or any other Liability when due, except to such creditors, if any, as listed on <u>Schedule 2.21</u>, the claims of which are being contested in good faith;
 - (r) neither Company has failed to reasonably maintain or properly repair any of its material Assets;
- (s) neither Company has acquired or agreed to acquire by merging or consolidating with, or by purchasing a substantial portion of the assets of, or by any other manner, any business of any Person or acquired any capital asset or related capital assets with a fair market value in excess of Two Hundred Fifty Thousand Dollars (\$250,000);
 - (t) neither Company has threatened, commenced or settled any Action;
 - (u) neither Company has entered into any Contractual Obligation to do any of the things referred to elsewhere in this Section 2.8; and
 - (v) no event or circumstance has occurred which has had, or would reasonably be expected to have, a Material Adverse Effect.

2.9 Assets.

2.9.1 Ownership of Assets. Each Company has sole and exclusive, good and marketable title to, or, in the case of property held under a lease or other Contractual Obligation, a sole and exclusive, Enforceable (except as limited by the Enforceability Exceptions) leasehold interest in, or right to use, all of its properties, rights and assets, whether real or personal and whether tangible or intangible, including all Assets reflected in the Most Recent Balance Sheet or acquired after the Most Recent Balance Sheet Date (except for such Assets which have been sold or otherwise disposed of since the Most Recent Balance Sheet Date in the Ordinary Course of Business and Assets owned by PPL (the "PPL Assets" and, collectively with all of the foregoing with respect to the Companies, the "Assets"). Except as disclosed on Schedule 2.9.1, none of the Assets is subject to any Encumbrance other than Permitted Encumbrances.

- 2.9.2 <u>Sufficiency of Assets</u>. Except as disclosed on <u>Schedule 2.9.2</u>, the Assets comprise all of the assets, properties and rights of every type and description, whether real or personal, tangible or intangible, used or necessary to conduct the Businesses as now conducted and contemplated to be conducted through December 31, 2010, including to perform the work required for the Backlog.
- 2.10 Accounts Receivable. Except as disclosed on Schedule 2.10, all Accounts Receivable, unbilled invoices, costs in excess of billings, work in process and other amounts, excluding retainage ("Receivables"), reflected on the Most Recent Balance Sheet and in the records and books of account of the Companies since the Most Recent Balance Sheet Date through the Closing Date as being due to the Companies have arisen in the Ordinary Course of Business, represent legal, valid, binding and enforceable obligations to the respective Company and are not subject to any contests, claims, counterclaims or setoffs. Except as disclosed on Schedule 2.10, there has been no material adverse change since the Most Recent Balance Sheet Date in the amount or collectability of the Receivables due either Company. Except as set forth on Schedule 2.10, as of September 30, 2009, (i) no account debtor or note debtor is delinquent for payments in excess of Fifty Thousand Dollars (\$50,000) or for more than ninety (90) days, (ii) no account debtor or note debtor has refused or threatened to refuse to pay its obligations to either Company for any reason, or has otherwise made a claim to set-off or similar claim, and (iii) to the Sellers' Knowledge, no account debtor or note debtor is insolvent or bankrupt.
- 2.11 <u>Backlog.</u> Schedule 2.11 sets forth a true, accurate and complete list of, on a contract-by-contract basis, the Contractual Obligations underlying the Companies' Backlog, including the specific amount of Backlog as of September 30, 2009 attributable to each such Contractual Obligation, together with a description of the remaining work to be performed under each such uncompleted Contractual Obligation. The Backlog amounts set forth on <u>Schedule 2.11</u>, both individually and in the aggregate, constitute the Sellers' best reasonable estimates of such amounts. Since the Most Recent Balance Sheet Date, the aggregate amount of Backlog, as adjusted for reductions resulting from the performance of work pursuant to such Contractual Obligations, has not been reduced by more than 5%. Except as set forth on <u>Schedule 2.11</u>, contained in the Data Room are true, accurate and complete copies or summaries of each written Contractual Obligation listed on <u>Schedule 2.11</u>, in each case, as amended or otherwise modified and in effect.

2.12 Real Property.

2.12.1 The Companies own no real property. Schedule 2.12 describes each leasehold interest in real property leased, subleased by, licensed or with respect to which a right to use or occupy has been granted to or by either Company (such leased real property is referred to as the "Real Property"), and specifies the lessor(s) of such Real Property, and identifies each lease (or in the case of a sublease, the underlying leases ("Underlying Leases") or in the case of the lease for the Headquarters Facility, a summary of the rental arrangement as no written lease currently exists) or any other Contractual Obligation under which such Real Property is leased by any Company (the "Real Property Leases"). Except as described on Schedule 2.12 there are no written or oral subleases, licenses, concessions, occupancy agreements or other Contractual Obligations granting to any other Person the right of use or occupancy of the Real Property, and there is no Person (other than the Companies and/or any lessee(s) of the Real Property specifically identified on Schedule 2.12) in possession of the Real Property. With respect to each Real

Property Lease that is a sublease, <u>Schedule 2.12</u> describes each of the Underlying Leases, if any. Except as set forth on <u>Schedule 2.12</u>, the Companies have valid leasehold interests in and to the Real Property and any and all Facilities located thereon, free and clear of all Encumbrances except for Permitted Encumbrances.

- 2.12.2 The Real Property Leases do not impose restrictions on any portion of either Business that interfere in any material respect with such Business. Neither the Companies nor any Seller is obligated to pay any leasing or real estate brokerage commission as a result of the Contemplated Transactions. There is no pending or, to the Sellers' Knowledge, threatened eminent domain taking affecting any of the Real Property. Contained in the Data Room are true, correct and complete copies of all of the Real Property Leases and Underlying Leases. Except as set forth on Schedule 2.12.2, no consents or approvals are required to be obtained under the Real Property Leases and/or the Underlying Leases in connection with the Contemplated Transactions, including from the landlord(s) thereunder.
- 2.12.3 To Sellers' Knowledge, no Facility currently existing on the Real Property encroaches upon, and no Facility under construction on the Real Property will encroach upon, the real property or easement estate of any other Person. To Sellers' Knowledge, no facility of any other Person encroaches upon the Real Property or any easement estate appurtenant to the Real Property. Each Facility is supplied with utilities and other services (including gas, electricity, water, drainage, sanitary sewer, storm sewer, fire protection and telephone) necessary for the operation of such Facility as the same is currently operated. To Sellers' Knowledge, each Facility is in good repair and operating condition, normal wear and tear excepted, with all required maintenance having been regularly performed and free from structural defects. Each parcel of Real Property abuts on, and has direct vehicular access to, a public road, or has access to a public road, in each case, to the extent necessary for the conduct of the Businesses.
- 2.12.4 Except as set forth on Schedule 2.12, to the Sellers' Knowledge, all Permits necessary in connection with the present use and operation of, the Real Property by the Companies and their lawful occupancy thereof have been issued by the appropriate Governmental Authorities. To the Sellers' Knowledge, all such Permits will continue in full force and effect immediately after giving effect to the Contemplated Transactions. Except as set forth on Schedule 2.12, to the Sellers' Knowledge, the Real Property and its current use, occupancy and operation by the Companies and the Facilities located thereon do not (a) constitute a nonconforming use under any applicable building, zoning, subdivision or other land use or similar Legal Requirement or (b) otherwise violate or conflict with any covenants, conditions, restrictions or other Contractual Obligation, including the requirements of any applicable Encumbrances thereto. Except as set forth on Schedule 2.12, no Company nor any Predecessor: (x) to Sellers' Knowledge is in violation of any Legal Requirement relating to Real Property which would adversely affect the Companies' use of the Real Property, including setback requirements, zoning restrictions and ordinances, building, life, access, safety, health and fire codes and ordinances affecting the Real Property; or (y) has received notice of any eminent domain, condemnation or similar proceeding pending or, to the Sellers' Knowledge, threatened, or any Governmental Order relating thereto. To the Sellers' Knowledge, the Real Property is not located within any flood plain or subject to any similar type of restriction for which any Permits necessary to the use thereof have not been obtained.

2.13 <u>Tangible Personal Property</u>. All of the fixtures and other improvements to the Real Property included in the Assets (including any Facilities) and all of the tangible personal property other than inventory included in the Assets (the "<u>Equipment</u>") (a) are adequate and suitable for their present and intended uses, (b) are in good working order, operating condition and state of repair, reasonable wear and tear excepted, (c) have no material defects (whether patent or latent) and (d) have been maintained in all material respects in accordance with the commercially reasonable standards of any manufacturer or any other governmental or regulatory entities.

2.14 Intellectual Property.

- 2.14.1 Schedule 2.14 identifies: (a) all registered Intellectual Property which has been issued to or is otherwise owned by either Company or that relates to or is otherwise used by either Company in the Businesses; and (b) each Contractual Obligation pursuant to which either Company or any Seller has granted or has been granted rights to any Intellectual Property or to which any of them is otherwise bound to any third party, excluding however "shrink wrap," "click through" or other standard agreements for commercially available, unmodified third party Software ("Commercial Software"). Neither Company has any pending application for registration which such Company has made or otherwise owns with respect to any Intellectual Property. True, accurate and complete copies or summaries of all such Contractual Obligations, as amended or otherwise modified and in effect, are contained in the Data Room, together with true, accurate and complete copies of all other written documentation evidencing ownership and prosecution (if applicable) of each such item. All such Contractual Obligations are in full force and effect. To the Sellers' Knowledge, there does not exist any claim, allegation, or basis for any claim or allegation, that any Intellectual Property owned or otherwise used by either Company is invalid or unenforceable or that such Company's rights with respect thereto are subject to claims or defenses that would impair or preclude enforcement of such rights, including misuse, laches, acquiescence, statute of limitations, abandonment or fraudulent registration. Schedule 2.14 also identifies each trade name, trade dress and unregistered trademark or service mark used by either Company or in connection with either Business or the Company Technology.
- 2.14.2 A Company is the sole owner of all rights, title and interest in and to all Intellectual Property with respect to, or has the right to use as specified on Schedule 2.14, all the Company Technology free and clear of any Encumbrances (other than Permitted Encumbrances) other than those contained in the Contractual Obligations identified in Schedule 2.14. Each Company has the right to use all other Technology and Intellectual Property used in its Business as currently conducted and as currently contemplated to be conducted through December 31, 2010. The Intellectual Property identified on Schedule 2.14 constitutes all registered Intellectual Property that is used in the Businesses and all registered Intellectual Property necessary for the conduct of the Businesses as currently conducted and as currently contemplated through December 31, 2010. No Seller has any ownership or other interest in any registered Intellectual Property used in either Businesses.
- 2.14.3 To the Sellers' Knowledge, no Company nor any Predecessor (a) has, nor has the conduct of either Business, interfered with, infringed upon, misappropriated, diluted or otherwise violated or come into conflict with any Intellectual Property rights of third parties or (b) has received any charge, complaint, claim, demand, or notice alleging any such

interference, infringement, misappropriation, dilution or other violation or conflict (including any claim that it must obtain a license or refrain from using any Intellectual Property rights of any third party in connection with the conduct of the Business or the use of the Company Technology, or, with respect to any other Technology used in the Business, that it must obtain a license that it does not already possess or refrain from using same). To the Sellers' Knowledge, there does not exist any basis for any such claim or allegation. To the Sellers' Knowledge, no third party has interfered with, infringed upon, misappropriated, diluted or otherwise violated or come into conflict with any Company Technology or any of the Company's other Intellectual Property. No claim or legal proceeding involving any Intellectual Property by or against either Company is pending or, to the Sellers' Knowledge, has been threatened. Except as otherwise specified on Schedule 2.14, and except for licenses to Commercial Software, neither Company is bound by any Contractual Obligation containing any covenant or other provision relating to Intellectual Property that in any way limits or restricts the ability of any such Company to use, exploit, assert, or enforce any of the Intellectual Property owned by either Company or conduct its Business as currently conducted and as currently contemplated to be conducted through December 31, 2010.

- 2.14.4 Except for that Software described on <u>Schedule 2.14</u>, the Company Software constitutes all the Software necessary to conduct the Businesses as currently conducted by the Companies and as currently contemplated to be conducted through December 31, 2010.
- 2.14.5 With respect to each item of the Company Technology or other Technology or Intellectual Property owned by either Company, and except as set forth on Schedule 2.14:
 - (a) a Company possesses all right, title, and interest in and to such item, or has the valid and Enforceable (subject to the Enforceability Exceptions) right to use same, free and clear of any Encumbrance (other than Permitted Encumbrances), except for Contractual Obligations identified on Schedule 2.14;
 - (b) to Sellers' Knowledge, such item is not subject to any outstanding Governmental Order, and no Action is pending or threatened, which challenges the legality, validity, enforceability, use or ownership of such item, nor, to the Sellers' Knowledge, is there any basis for such a challenge; and
 - (c) no Company has agreed to nor otherwise has a Contractual Obligation to indemnify any Person for or against any interference, infringement, misappropriation or other conflict with respect to such item.
- 2.14.6 Schedule 2.14 identifies each item of Technology that is used by either Company pursuant to any license, sublicense or other Contractual Obligation except for agreements for Commercial Software (the "Licenses"). Except as disclosed on Schedule 2.14 and the Licenses, there are no royalties or other compensation payable for the use of any such Technology. Contained in the Data Room are true, accurate and complete copies or summaries of all of the Licenses, if any, in each case, as amended or otherwise modified and in effect, and each of such Licenses is in full force and effect, is valid and enforceable in

accordance with its terms and to the Sellers' Knowledge no party thereto is in material breach of any of the terms thereof. With respect to each such item identified on <u>Schedule 2.14</u>: (a) to Sellers' Knowledge such item is not subject to any outstanding Governmental Order, and no Action is pending or threatened which challenges the legality, validity or enforceability of such item, nor, to the Sellers' Knowledge, is there any basis for such a challenge and (b) neither any Seller nor either Company has granted any sublicense or similar right with respect to any License covering such item.

2.14.7 Each Company has taken reasonable and customary measures and precautions necessary to protect and maintain the confidentiality of all Trade Secrets in which such Company has any right, title or interest and otherwise to maintain and protect all such Trade Secrets.

2.15 Legal Compliance; Illegal Payments; Permits.

- 2.15.1 Compliance. Except as set forth on Schedule 2.15, The Companies and PPL are and have been in compliance with:
 - (a) their respective Organizational Documents and, to the Sellers' Knowledge, there is no basis which could constitute a failure to comply; and
 - (b) all material Legal Requirements and, to the Sellers' Knowledge, there is no basis which could constitute a failure to comply.
- 2.15.2 <u>Anti-Corruption</u>. No agent, Affiliate, employee or other Person associated with or acting on behalf of either Company or PPL directly or indirectly, has (a) made any unlawful contribution, gift, bribe, rebate, payoff, influence payment, kickback, or other payment to any Person, private or public, regardless of form, whether in money, property, or services (i) to unlawfully obtain favorable treatment in securing business, (ii) to unlawfully pay for favorable treatment for business secured, (iii) to unlawfully obtain special concessions or for special concessions already obtained or (iv) for any other illegal purpose or (b) established or maintained any fund or asset for the benefit of either Company or PPL that has not been recorded in the books and records of such Company or PPL.
- 2.15.3 <u>Permits</u>. Except as disclosed on <u>Schedule 2.15.3</u>, the Companies or PPL have been granted all Permits under all Legal Requirements necessary for the conduct of the Businesses. <u>Schedule 2.15.3</u> describes each Permit affecting, or relating to, the Assets or either Business, including the holder of such Permit, together with the Governmental Authority or other Person responsible for issuing such Permit. Except as disclosed on <u>Schedule 2.15.3</u>, (a) the Permits are in full force and effect, (b) neither Company or PPL is in material breach or violation of, or default under, any such Permit, and, to the Sellers' Knowledge, no basis exists which, with notice or lapse of time or both, would constitute any such breach, violation or default, (c) the Permits will continue to be in full force and effect, on identical terms following the consummation of the Contemplated Transactions, and (d) each Company has properly and validly completed in all material respects all filings and registrations that are required for the operation of its Business. There is no investigation or Action pending or, to Sellers' Knowledge, threatened that would result in the termination, revocation, suspension or restriction of any Permit or the imposition of any fine, penalty or other sanctions for violation of any Legal Requirement relating to any Permit.

2.15.4 Additional Compliance Representations.

- (a) No petition under the federal bankruptcy laws or any state insolvency law has been filed by or against, or a receiver, fiscal agent or similar officer appointed by a court for the business or property of, (i) any Seller, or either Company, (ii) any partnership in which any Seller or either Company was a general partner at or within two years before the Closing Date, or (iii) any corporation or business association of which either Rooney or Murphy was an executive officer at or within two years before the Closing Date.
- (b) No Seller, within the last five (5) years, nor either Company has been convicted in a criminal proceeding (excluding traffic violations and other immaterial or minor offenses), no Seller nor either Company is a named subject of a pending criminal proceeding and, to the Sellers Knowledge, no Seller nor either Company is a named subject or target of a pending criminal investigation.
- (c) No Seller nor either Company has been the subject of any Governmental Order not subsequently reversed, suspended or vacated, permanently or temporarily enjoining such Seller or such Company from, or otherwise limiting such Seller's or such Company's involvement or right to engage in, any of the following activities:
 - (i) acting as a futures commission merchant, introducing broker, commodity trading advisor, commodity pool operator, floor broker, leverage transaction merchant, any other Person regulated by the Commodity Futures Trading Commission, or as an associated Person of any of the foregoing, or as an investment adviser, underwriter, broker or dealer in securities, or as an affiliated Person, director or employee of any investment company, bank, savings and loan association or insurance company, or engaging in or continuing any conduct or practice in connection with such activity;
 - (ii) engaging in any type of business practice; or
 - (iii) engaging in any activity in connection with the purchase or sale of any security or commodity, or in connection with any violation of federal or state securities laws or federal commodities laws.
- (d) No Seller nor either Company has been found by a Governmental Authority in a civil action or by the Securities and Exchange Commission ("<u>SEC</u>") to have violated any securities Law, which judgment in such civil action or finding by the SEC has not been subsequently reversed, suspended or vacated.

- (e) No Seller nor either Company has been found by a Governmental Authority in a civil action or by the Commodities Futures Trading Commission to have violated any federal commodities law, which judgment in such civil action or finding by the Commodity Futures Trading Commission has not been subsequently reversed, suspended or vacated.
- (f) No Seller nor either Company has been involved in any of the following matters: (a) making any unlawful political contribution; (b) the disbursement or receipt of funds of either Company outside the normal system of accountability; (c) unlawful payments, whether direct or indirect, to or from foreign or domestic governments, officials, employees or agents for purposes other than the satisfaction of lawful obligations, or any transaction which has as its intended effect the transfer of assets of any Seller or either Company for the purpose of effecting such payment; (d) the intentionally improper or inaccurate recording of payments and receipts on the books of either Company; or (e) any other matters of a similar nature involving unlawful or noncompliant disbursements of assets of either Company.
- 2.16 <u>Inventories</u>. Except as disclosed on <u>Schedule 2.16</u>, the Companies maintain sufficient Inventory in the Ordinary Course of Business to conduct their respective Businesses consistent with past practices. Except as disclosed on <u>Schedule 2.16</u>, the Inventory of the Companies reflected on the Most Recent Balance Sheet and in the records and books of account of the Companies since the Most Recent Balance Sheet Date are of a quality and a quantity usable or saleable, as the case may be, in the Ordinary Course of Business. Except as disclosed on <u>Schedule 2.16</u>, each Company has sole custody and control of and maintains adequate insurance coverage on all materials, supplies, parts or other assets delivered to such Company by or on behalf of its customers for use in connection with projects such Company is undertaking for such customers (the "<u>Customer Assets</u>"); no material Customer Asset has been damaged, lost, stolen, or otherwise suffered a material diminution in value from the time of receipt by either Company; and neither Company has received notice of any claim, loss, or damage related to the Customer Assets.

2.17 Employee Benefit Plans.

(a) <u>Schedule 2.17(a)</u> contains, a true, accurate and complete list of each "employee benefit plan" (within the meaning of Section 3(3) of ERISA, including multiemployer plans within the meaning of Section 3(37) of ERISA, each a "<u>Multiemployer Plan</u>"), and all stock purchase, stock option, severance, employment, change-in-control, fringe benefit, plans provided in collective bargaining agreements, bonus, incentive, deferred compensation, employee loan and all other employee benefit plans, agreements, programs, policies or other arrangements, whether or not subject to ERISA (including any funding mechanism therefor now in effect or required in the future as a result of the Contemplated Transactions or otherwise), whether formal or informal, oral or written, under which (i) any current or former employee, director or

consultant of either Company (the "<u>Company Employees</u>") or Transferred Employee has any present or future right to benefits and which are contributed to, sponsored by or maintained by either Company or PPL or (ii) either Company or PPL has had, has or may have any actual or contingent present or future material liability or obligation. All such plans, agreements, programs, policies and arrangements shall be collectively referred to as the "<u>Employee Plans</u>"; provided, however, <u>Schedule 2.17(a)</u> does not set forth all Employee Plans that are Multiemployer Plans.

- (b) With respect to each Employee Plan other than a Multiemployer Plan, the Sellers have made available to Buyer in the Data Room a true, accurate and complete copy (or, to the extent no such copy exists, an accurate description) thereof and, to the extent applicable: (i) any contracts or agreements relating to any Employee Plan other than a Multiemployer Plan, including all trust agreements, insurance or annuity contracts, investment management agreements, record keeping agreements and other material documents or instruments related thereto; (ii) the most recent determination letter, if applicable; (iii) any summary plan description and other written communications (or a description of any oral communications) by either Company to the Company Employees or PPL to the Transferred Employees concerning the extent of the benefits provided under a Employee Plan; (iv) a summary of any proposed amendments or changes anticipated to be made to the Employee Plans at any time within the twelve months immediately following the date hereof; and (v) for the three most recent years (A) the Form 5500 and attached schedules, (B) reviewed financial statements, (C) actuarial valuation reports and (D) any non-discrimination testing results.
- (c) Except as set forth on Schedule 2.17(c), (i) each Employee Plan other than a Multiemployer Plan has been established and administered in all material respects in accordance with its terms and in compliance with the applicable provisions of ERISA, the Code, the Health Insurance Portability and Accountability Act of 1996 and other applicable Laws; (ii) each Employee Plan which is intended to be qualified within the meaning of Section 401(a) of the Code, other than a Multiemployer Plan, has been established and maintained in the form of a prototype plan with respect to which the sponsor has received an opinion letter from the Internal Revenue Service relating to the form of the plan document upon which the relevant Company, PPL or Controlled Group member that maintains the plan may rely, and nothing has occurred, whether by action or failure to act, that could reasonably be expected to cause such Employee Plan to not be considered a qualified plan under Section 401(a) of the Code; (iii) no event has occurred and to the Sellers' Knowledge or with respect to any Multiemployer Plan Sellers' Internal Knowledge, no condition exists that would reasonably be expected to subject either Company or PPL, either directly or by reason of its affiliation with any entity that at any time within the current or any of the 5 previous calendar years, is or was a member of its "Controlled Group" (defined as any

organization which is a member of a controlled group of organizations within the meaning of Sections 414(b), (c), (m) or (o) of the Code or Section 4001 of ERISA), to any Tax, fine, Encumbrance (other than Permitted Encumbrances), penalty or other Liability imposed by ERISA, the Code or other applicable Laws; (iv) for each Employee Plan (other than a Multiemployer Plan) with respect to which a Form 5500 has been filed, no material change has occurred with respect to the matters covered by the most recent Form since the date thereof; (v) no "reportable event" (as such term is defined in Section 4043 of ERISA) that could reasonably be expected to result in Liability has occurred with respect to any Employee Plan (other than a Multiemployer Plan); (vi) no nonexempt "prohibited transaction" (as such term is defined in Section 406 of ERISA and Section 4975 of the Code) or failure to make at least the minimum contributions required under Section 302 of ERISA, Sections 430 or 431 of the Code, or Section 412 of the Code (whether or not waived)) has occurred with respect to any Employee Plan other than a Multiemployer Plan; (vii) no Employee Plan other than a Multiemployer Plan has had benefit reductions under Section 436 of the Code or is or is expected to be in "at risk status" within the meaning of Section 430(i)(4) of the Code, and no Employee Plan other than a Multiemployer Plan has, or is expected to have, an "adjusted funding target attainment percentage", as defined in Section 436 of the Code, that is less than 60%; (viii) there is no present intention that any Employee Plan other than a Multiemployer Plan be materially amended, suspended or terminated, or otherwise modified to change benefits (or the levels thereof) under any Employee Plan other than a Multiemployer Plan at any time within the twelve months immediately following the date hereof; (ix) no Employee Plan is a split-dollar life insurance program or otherwise provides for loans to executive officers (within the meaning of the Sarbanes-Oxley Act of 2002); and (x) neither Company nor PPL has incurred any current or projected Liability in respect of post-employment or post-retirement health, medical or life insurance benefits for current, former or retired employees of either Company or PPL, except as required to avoid an excise tax under Section 4980B of the Code or otherwise except as may be required pursuant to any other applicable Law.

- (d) No Employee Plan other than a Multiemployer Plan is an "employee pension benefit plan" (within the meaning of Section 3(2) of ERISA) subject to Title IV of ERISA, and neither Company nor PPL has any obligation to contribute, and has not incurred any actual or contingent liability or obligation (including any obligation to make any contribution) to or in respect of any such plan other than a Multiemployer Plan.
- (e) Except as set forth on Schedule 2.17(e), with respect to each Employee Plan other than a Multiemployer Plan, (i) no actions, suits or claims (other than routine claims for benefits in the ordinary course) are pending or to the Sellers' Knowledge threatened; (ii) to the Sellers' Knowledge no facts or circumstances exist that could give rise to any such actions, suits or claims; (iii) no written or oral communication has been received by Sellers, the Companies or PPL from

the Pension Benefit Guaranty Corporation (the "<u>PBGC</u>") in respect of any Employee Plan subject to Title IV of ERISA concerning the funded status of any such plan or any transfer of assets and liabilities from any such plan in connection with the Contemplated Transactions; and (iv) no administrative investigation, audit or other administrative proceeding by the Department of Labor, the PBGC, the Internal Revenue Service (the "<u>IRS</u>") or other governmental agencies are pending, in progress or to the Sellers' Knowledge threatened (including any routine requests for information from the PBGC).

- (f) Except as set forth on Schedule 2.17(f), no Employee Plan or Legal Requirement exists that, as a result of the execution of this Agreement, approval of this Agreement by the Companies' respective members, or the Contemplated Transactions would (i) result in severance pay, termination indemnity or any similar payment or any increase in severance pay, termination indemnity or any similar payment, (ii) accelerate the time of payment or vesting or result in any payment or funding (through a grantor trust or otherwise) of compensation or benefits under, increase the amount payable or result in any other material obligation pursuant to, any of the Employee Plans, (iii) limit or restrict the right of either Company to merge, amend or terminate any of the Employee Plans, (iv) cause either Company to record additional compensation expense on its income statement with respect to any outstanding stock option or other equity-based award, or (v) result in payments under any of the Employee Plans which would not be deductible under Section 280G of the Code.
- (g) Except as set forth on <u>Schedule 2.17(g)</u>, there has been no amendment to, written interpretation of or announcement (whether or not written) by either Company or PPL relating to, or any change in employee participation or coverage under, any Employee Plan that would increase the expense of maintaining such Employee Plan above the level of the expense incurred in respect thereof for the most recent fiscal year ended prior to the date hereof.
- (h) Except as set forth in Schedule 2.17(h), no compensation under any Employee Plan subject to Section 409A of the Code is or has been required to be includible in the gross income of any participant or beneficiary by reason of Section 409A(a)(i)(A) of the Code or is or has been subject to any additional tax under Section 409A(a)(i)(B) of the Code, and no amounts are or have been includible in the gross income of any participants or beneficiaries by reason of Section 409A(b) of the Code.
- (i) Except as specified on Schedule 2.17(i), with respect to each Multiemployer Plan: (i) all contributions required to be made with respect to employees of the relevant Company or PPL, or any member of their respective Controlled Groups, have been timely paid; (ii) neither the relevant Company nor PPL, nor any member of their respective Controlled Groups, has or is expected to incur, directly or indirectly, any withdrawal liability under ERISA with respect to any such plan

(whether by reason of the Contemplated Transactions or otherwise); (iii) to the Sellers' Internal Knowledge, no such plan is (or is expected to be) insolvent or in reorganization and no accumulated funding deficiency (as defined in Section 302 of ERISA and Section 431 of the Code), whether or not waived, exists or is expected to exist with respect to any such plan and no such plan is or is expected to be in "endangered status" or "critical status" within the meaning of Section 432 of the Code; (iv) neither the relevant Company, PPL nor any member of their respective Controlled Groups, has withdrawn, partially withdrawn, or received any notice of any claim or demand for withdrawal liability or partial withdrawal liability against any of them; (v) the Sellers have made available to Buyer in the Data Room, true, correct and complete copies of (A) all information that has been provided to the relevant Company, PPL, or any members of their respective Controlled Groups, regarding assessed or potential withdrawal liability under any Multiemployer Plan, and (B) the most recent actuarial report or other information received from a Multiemployer Plan regarding its funding status, financial condition, or related matters; (vi) the rules contained in Sections 4203(b) and 4208(d)(1) of ERISA, which relate to employers in the building and construction industry, would apply in determining whether any complete withdrawal or partial withdrawal from any Multiemployer Plan has occurred; and (vii) neither the relevant Company, PPL nor any member of their respective Controlled Groups, has any actual or contingent liability under Section 4204 of ERISA.

2.18 Environmental Matters. Except as set forth on Schedule 2.18:

- 2.18.1 The Companies have complied with, and are in compliance with, in all material respects all Environmental Laws, including with respect to the conduct of the Businesses at jobsites and other locations. Except as set forth on <u>Schedule 2.18</u>, the Companies have not received, orally or in writing, any actual or threatened order, notice, report or other communication of any actual or potential violation or failure by the Company to comply with any Environmental Laws.
- 2.18.2 Neither the Companies nor the Sellers have received any notice, orally or in writing, that there are any pending or, threatened claims or Encumbrances resulting from any Environmental, Health and Safety Liabilities or arising under or pursuant to any Environmental Law, with respect to or affecting any of the Facilities or any other properties and assets (whether real, personal or mixed) owned or operated by the Companies.
- 2.18.3 The Companies have not treated, stored, disposed of, arranged for or permitted the disposal of, transported, handled, Released or exposed any Person to, any Contaminants, or owned or operated any property or facility that is or has been contaminated by any Contaminants, so as to give rise to any current or future Liability of either of the Companies pursuant to any Environmental Laws.

- 2.18.4 Except as set forth on <u>Schedule 2.18</u>, the Companies have not assumed, undertaken or otherwise become subject to, or provided any indemnity with respect to, any Liability pursuant to any Environmental Laws of any other Person.
- 2.18.5 Other than in compliance with all applicable Legal Requirements, the Companies have not manufactured, sold, marketed, installed or distributed products or items containing asbestos, silica or other Contaminants and do not have any Liability with respect to the presence or alleged presence of Contaminants in any product or item or at or upon any property or Facility.
- 2.18.6 Contained in the Data Room are true, accurate and complete copies of all environmental audits, reports, assessments and other documents in the Companies' or Sellers' possession or under their reasonable control that materially bear on environmental, health or safety Liabilities relating to the past or current operations, Facilities or properties of the Businesses.
- 2.18.7 Notwithstanding anything to the contrary in Section 2.18, the storage or use of minimal quantities of Contaminants on the Real Property or in the Businesses shall not be a breach of the representations and warranties in Section 2.18; provided that, such materials are of a type and are used and stored only in a quantity normally used or stored in connection with the occupancy of the Companies' facilities and the operations of the businesses in the industry in which the Companies operate (including fuel, paint, cleaning fluids and supplies normally used in the Businesses), and such materials are being and have been held, stored and used in compliance in all material respects with all Environmental Laws.

2.19 Contracts.

- 2.19.1 Contracts. Except as disclosed on Schedule 2.19 or as contemplated by this Agreement, neither Company nor PPL is bound by or a party to:
- (a) any Contractual Obligation (or group of related Contractual Obligations) for the purchase or sale of inventory, raw materials, commodities, supplies, goods, products, equipment or other personal property, or for the furnishing or receipt of services, in each case, the performance of which will extend over a period of more than one year or which provides for either an aggregate or annual payments to or by the Companies and/or PPL in the aggregate in excess of One Hundred Thousand Dollars (\$100,000);
- (b) (i) any capital lease or (ii) any other lease or other Contractual Obligation relating to the Equipment providing for annual rental payments in excess of Fifty Thousand Dollars (\$50,000), under which any Equipment is held or used by either Company or PPL;
- (c) any Contractual Obligation, other than Real Property Leases or leases relating to the Equipment, relating to the lease or license of any Asset, including Technology and Intellectual Property (and including all customer license and maintenance agreements) that is not included on Schedule 2.14;

- (d) any Contractual Obligation relating to the acquisition or disposition of (i) any business (whether by merger, consolidation or other business combination, sale of securities, sale of assets or otherwise) or (ii) any asset with a value of One Hundred Thousand Dollars (\$100,000) or more;
- (e) any Contractual Obligation under which either Company or PPL is, or may become, obligated to pay any amount in respect of indemnification obligations, purchase price adjustment or otherwise in connection with any (i) acquisition or disposition of assets or securities (other than the sale of inventory and excess or obsolete PP&E, which has been replaced by all substitute PP&E necessary to continue to operate the Businesses), (ii) merger, consolidation or other business combination or (iii) series or group of related transactions or events of the type specified in clauses (i) and (ii) above.
- (f) any Contractual Obligation (other than each Company's Organizational Documents) concerning or consisting of a partnership, limited liability company or joint venture agreement or any other relationship involving the sharing of profits, losses or costs;
- (g) any Contractual Obligation (or group of related Contractual Obligations) (i) under which either Company or PPL has created, incurred, assumed or guaranteed any Debt or (ii) under which either Company or PPL has permitted any Asset to become subject to an Encumbrance other than Permitted Encumbrances;
 - (h) any Contractual Obligation under which any other Person has guaranteed any Debt of either Company or PPL;
- (i) any Contractual Obligation to purchase goods or services exclusively from a given Person or Persons or purchase a minimum amount of goods or services from a given Person or Persons, or all or a portion of the supply of certain goods or services utilized by either Company or PPL from a given Person or Persons;
- (j) any Contractual Obligation involving any obligation on the part of either Company, PPL, Rooney or Murphy to refrain from competing with any Person, from soliciting any employees, independent contractors or customers of any Person or from conducting any other lawful commercial activity (including in any geographic region) or any such Contractual Obligation for either Company's or PPL's benefit from any other Person(s);
- (k) any Contractual Obligation under which either Company or PPL is, or may become, obligated to incur any severance pay or special Compensation obligations which would become payable by reason of this Agreement or the Contemplated Transactions;

- (l) any Contractual Obligation under which either Company or PPL is, or may, have any Liability to any investment bank, broker, financial advisor, finder or other similar Person (including an obligation to pay any legal, accounting, brokerage, finder's, or similar fees or expenses in connection with this Agreement or the Contemplated Transactions) other than the obligations of the Companies under Contractual Obligations with CAC and Gardere which will be terminated as of the Closing Date;
- (m) any profit sharing, equity option, equity purchase, equity appreciation, deferred compensation, severance or other plan or arrangement for the benefit of either Company's or PPL's current or former members, officers or employees, consultants or independent contractors;
- (n) any Contractual Obligation providing for the employment or consultancy (including on an independent contractor basis) with an individual (or in the case of a consultant or independent contractor, an entity) on a full-time, part-time, consulting or other basis or otherwise providing Compensation or other benefits to any officer, member, employee or consultant (other than an Employee Plan and the Contractual Obligations with CAC and Gardere);
- (o) any agency, dealer, distributor, sales representative, marketing, handler, third party service provider, or other similar agreement (other than the Contractual Obligations with CAC and Gardere);
- (p) any Contractual Obligation under which either Company or PPL has advanced (other than travel allowances in the Ordinary Course of Business) or loaned an amount to any of its Affiliates or employees;
 - (q) any settlement, conciliation or similar Contractual Obligations imposing an obligation on either Company or PPL after the Closing Date;
- (r) any Contractual Obligation that limits the ability of either Company, PPL or any of their Affiliates to incur any Debt or to Guarantee any Debt or other obligation of any Person, or that limits the amount of any Debt that either Company or PPL may incur or Guarantee, or prohibits it from granting any Encumbrance (other than Permitted Encumbrances) on any Asset to secure any Debt incurred or Guaranteed (other than Contractual Obligations which will be terminated as of the Closing); or
- (s) any Contractual Obligation not otherwise disclosed on <u>Schedule 2.19</u> and (A) pursuant to which the Companies and/or PPL have an aggregate future liability to any Person in excess of One Hundred Thousand (\$100,000), (B) entered into other than in the Ordinary Course of Business or other than on arms-length terms, or (C) that is material to the conduct or operation of either Business.

Contained in the Data Room are true, accurate and complete copies, in all material respects, of each written Contractual Obligation (or to the extent no such copy exists, an accurate description) listed on <u>Schedule 2.19</u>, in each case, as amended or otherwise modified and in effect.

- 2.19.2 Enforceability, etc. Each Contractual Obligation required to be disclosed on Schedule 2.24 (Debt), Schedule 2.12 (Real Property Leases), Schedule 2.14 (Intellectual Property), Schedule 2.17(a) (Employee Benefit Plans), Schedule 2.19 (Contracts), Schedule 2.21 (Customers and Suppliers) or Schedule 2.26 (Insurance) (each, a "Disclosed Contract") is Enforceable (subject to any Enforceability Exceptions) against the relevant Company and, to the Sellers' Knowledge, each other party to such Contractual Obligation, and is in full force and effect, and subject to getting the consents set forth in Schedule 2.5, will continue to be so Enforceable and in full force and effect on identical terms, and subject to getting the consents set forth in Schedule 2.5, the consummation of the Contemplated Transactions shall not (either alone or upon the occurrence of additional acts or events other than acts or events by the Companies or Buyer after the Closing) result in any payment or payments being due from either Company or PPL to any Person following the consummation of the Contemplated Transactions.
- 2.19.3 <u>Breach, etc.</u> Except as set forth on <u>Schedule 2.19.3</u>, neither Company, PPL nor, to the Sellers' Knowledge, any other party to any Disclosed Contract is in material breach or violation of, or default under, or has repudiated any provision of, any Disclosed Contract (including all performance bonds, warranty obligations or otherwise), nor to the Sellers' Knowledge has any event occurred which, with the passage of time or the giving of notice, or both, would constitute a material breach or violation of, or default under, any Disclosed Contract (including all Surety Bonds, warranty obligations or otherwise). Except as set forth on <u>Schedule 2.19.3</u>, neither Sellers nor either Company has received notice from any other party to any Disclosed Contract or to the Sellers' Knowledge has any reason to believe that such party intends to terminate such Disclosed Contract or alter in any material way the relationship of the parties under such Disclosed Contract. No party to any Disclosed Contract has given either Company or any Seller notice of any action to terminate, cancel, rescind or procure a judicial reformation thereof nor, to the Sellers' Knowledge, are there any circumstances which would lead to any of the foregoing.
- 2.19.4 <u>Bids</u>. <u>Schedule 2.19.4</u> sets forth a true, accurate and complete list of each outstanding bid or proposal for business submitted by each Company.
- 2.20 <u>Affiliate Transactions</u>. Except for the matters disclosed on <u>Schedule 2.20</u>, no Seller nor any Affiliate of any Seller is, directly or indirectly, an officer, director, member, manager, employee, consultant, competitor, creditor, debtor, customer, distributor, supplier or vendor of, or is a party to any Contractual Obligation with, either Company. Except as disclosed on <u>Schedule 2.20</u>, no Seller nor any Affiliate of any Seller owns or has any ownership interest in any Asset used in, or necessary to, the Business.
- 2.21 <u>Customers and Suppliers</u>. <u>Schedule 2.21</u> sets forth a true, accurate and complete list of (a) the ten largest customers of Precision (measured by aggregate billings) for the fiscal year ended December 31, 2008 and for the nine months ended September 30, 2009, indicating the existing Contractual Obligations with each such customer by product or service provided and (b) the ten largest suppliers of materials, products or services to the Companies (measured by the aggregate amount purchased by the Companies) for the fiscal year ended December 31, 2008 and for the nine months ended September 30, 2009, indicating the Contractual Obligations for continued supply from each such supplier. Except as set forth on <u>Schedule 2.21</u>, the relationships of the Companies with the customers and the suppliers required to be listed on <u>Schedule 2.21</u>, and with El Paso Energy for both current and planned projects,

including the Ruby Ridge Pipeline project, are good commercial working relationships and none of such customers or the suppliers has canceled, terminated or otherwise altered (including any reduction in the rate or amount of sales or purchases, material increase in the prices charged or paid, or change to the supply or credit terms, as the case may be) or notified either Company of any intention to do any of the foregoing or otherwise threatened to cancel, terminate or alter (including any reduction in the rate or amount of sales or purchases, the amount of the purchase price for sales or purchases, material increase in the prices charged or paid, or change to the supply or credit terms, as the case may be) its relationship with either Company (other than upon the completion of the project covered by the applicable Contractual Obligation). As of the Closing Date, to the Sellers' Knowledge, there is no reason to believe that there will be any change in the relationships of the Companies with any of their customers or suppliers, including El Paso Energy for both current and planned projects, including the Ruby Ridge Pipeline project, as a result of the Contemplated Transactions. The Sellers have no notice of any facts or circumstances that have resulted or would reasonably be expected to result in a change in the relationship that the Companies have with any of their customers and suppliers listed or required to be listed on Schedule 2.21, including El Paso Energy for both current and planned projects, including the Ruby Ridge Pipeline project.

2.22 <u>Customer Warranties</u>. Except as set forth on <u>Schedule 2.6.1</u>, there are no pending, nor to the Sellers' Knowledge, threatened, claims under or pursuant to any warranty, whether expressed or implied, on the products or services sold prior to the Closing Date by Precision. Except as set forth on <u>Schedule 2.6.1</u> and <u>Schedule 2.22</u>, all of the services rendered by Precision (whether directly or indirectly through independent contractors) have been performed in conformity with all express warranties and, in all material respects, with all applicable contractual commitments, and the Companies do not have any pending claims for replacement or repair or for other damages relating to or arising from any such sources, except for amounts which are immaterial in the aggregate, and neither Company has or shall have any Liability for replacement or repair or for other damages relating to or arising from any such services, except for amounts incurred in the Ordinary Course of Business which are immaterial individually and in the aggregate and not required by GAAP to be disclosed in the Interim Financials. Except as set forth on <u>Schedule 2.22</u>, none of Precision's Disclosed Contracts with any customer contains any unusual warranty provisions that would impose material Liability on either Company. Set forth on <u>Schedule 2.22</u> is the aggregate amount of warranty claims incurred by Precision during each of the last three completed fiscal years. Except as set forth on <u>Schedule 2.5</u> and <u>Schedule 2.22</u>, there is no reason to expect an increase in the amount of warranty claims in proportion to revenues in connection with current projects subject to Contractual Obligations.

2.23 <u>Capital Expenditures and Investments</u>. The Companies' outstanding Contractual Obligations and budget for capital expenditures and Investments are set forth on <u>Schedule 2.23</u>, which includes a schedule of all monies disbursed on account of capital expenditures and Investments made by the Companies since the Most Recent Balance Sheet Date through September 30, 2009.

2.24 Employees.

2.24.1 All non-union employees of the Companies and PPL are, including employer, job title, location and base compensation, listed on Schedule 2.24, together with a list of all employees whose employment with either Company or PPL is currently contemplated to be terminated by the Companies or PPL at or prior to the Closing, other than the termination of the Transferred Employees pursuant to this Agreement. Except as set forth on Schedule 2.19, no present or past employee of either Company or PPL is bound by an unexpired non-competition agreement with either Company or PPL or, to Sellers' Knowledge, any other Person. Except as set forth on Schedule 2.24, pursuant to Employee Plans or travel advances to employees in the Ordinary Course of Business, neither Company or PPL is a party to any Contractual Obligation with any employee or former employee.

2.24.2 Except as disclosed on Schedule 2.24, there are no material labor troubles (including any work slowdown, lockout, stoppage, picketing or strike) pending, or to the Sellers' Knowledge, threatened between either Company or PPL, on the one hand, and its employees, on the other hand. Except as disclosed on Schedule 2.24, (a) no employee of either Company or PPL is represented by a labor union, association or representative body, (b) neither Company nor PPL is a party to, or otherwise subject to, any collective bargaining agreement or other labor union, association or representative body Contractual Obligation, (c) during the past five years there have been no strikes, slowdowns, work stoppages, lockouts, or, to the Sellers' Knowledge, threats thereof, by or with respect to any employees of either Company or PPL, (d) no petitions or proceedings by an employee or group of employees of either Company or PPL are pending with any labor relations board seeking recognition of a bargaining representative and (e) there is no organizational effort currently being made or threatened by, or on behalf of, any labor union, association or representative body to organize employees of either Company or PPL and no demand for recognition of employees of either Company or PPL is currently pending on behalf of, any labor union, association or representative body. Neither Company or PPL is a party to, or otherwise bound by, any consent decree with, or citation or other Governmental Order relating to employees or employment practices. Except as disclosed on Schedule 2.24, the Companies and PPL are in compliance with applicable Legal Requirements, Contractual Obligations, and policies relating to employment, employment practices, wages, hours, and terms and conditions of employment, including the obligations of the Fair Labor Standards Act ("FLSA") and the Worker Adjustment and Retraining Notification Act of 1988 ("WARN"), and, in all material respects, all other notification and bargaining obligations arising under any collective bargaining agreement, by Legal Requirement or otherwise. Neither Company nor PPL has effectuated a "plant closing" or "mass layoff" as those terms are defined in WARN, affecting in whole or in part any site of employment, facility, operating unit or employee of either Company or PPL without complying with all applicable provisions of WARN, or implemented any early retirement, separation or window program within the past five years, nor has either Company or PPL planned or announced any such action or program for the future except as contemplated by this Agreement. Except as provided in Section 4.10 and except as contemplated by this Agreement, no current executive officer's or other key employee's employment with either Company or PPL has been terminated for any reason nor has any such officer or employee notified the applicable Company or PPL of his or her intention to resign or retire.

2.24.3 Neither Company nor PPL is delinquent in payments to any of its (a) employees for any wages, salaries, overtime pay, commissions, bonuses, benefits or other Compensation, or (b) consultants for payments for any services arising under any policy, practice, Contractual Obligation, plan, program or Legal Requirement. None of either Company's or PPL's employment policies or practices is currently being audited or, to Sellers' Knowledge investigated by any Governmental Authority. Except as set forth on Schedule 2.24.3, there is no pending or, to the Sellers' Knowledge, threatened Action, unfair labor practice charge, or other charge or inquiry against either Company or PPL brought by or on behalf of any employee, prospective employee, former employee, retiree, labor organization or other representative of either Company's or PPL's employees, or other individual or any Governmental Authority with respect to employment practices brought by or before any Governmental Authority.

2.24.4 To the Sellers' Knowledge, no employee, officer, manager, contractor or consultant of either Company or PPL is obligated under any applicable Law or under any Contractual Obligation of any nature, or is subject to any judgment, decree or Governmental Order that would interfere with the use of such employee's, officer's, contractor's or consultant's best efforts to promote the interests of the applicable Company or PPL or that would conflict with either Business. To the Sellers' Knowledge, the conduct of the Businesses has not and will not, and the consummation of the Contemplated Transactions will not, conflict with or result in a material breach of the terms, conditions or provisions, or constitute a default under any Contractual Obligation under which any employee, officer, contractor or consultant of either Company or PPL is obligated.

2.25 Litigation; Governmental Orders.

2.25.1 <u>Litigation</u>. Except as disclosed on <u>Schedule 2.25</u>, there is no Action to which either Company or PPL is a party (either as plaintiff or defendant) or to which their Assets are subject pending, or to the Sellers' Knowledge, threatened. There is no Action to which either Company or any Seller is a party (either as plaintiff or defendant) or to which its assets are subject pending, or to the Sellers' Knowledge, threatened, which (a) in any manner challenges or seeks the rescission of, or seeks to prevent, enjoin, alter or delay the consummation of, or otherwise relates to, this Agreement or the Contemplated Transactions, or (b) may result in any change in the current equity ownership of either Company or PPL, nor, to the Sellers' Knowledge, is there any basis for any of the foregoing. Except as disclosed on <u>Schedule 2.25</u>, there is no Action that either Company or any Seller (with respect to any Seller, related to the Businesses) presently intends to initiate. Except as disclosed on <u>Schedule 2.25</u>, no Seller has any notice of any facts or circumstances that would reasonably be expected to result in any Action to which either Company or PPL is a party (either as plaintiff or defendant) or to which its Assets are subject.

2.25.2 <u>Governmental Orders</u>. Except as disclosed on <u>Schedule 2.25</u>, no Governmental Order has been issued that is applicable to, or otherwise affects, either Company, PPL or their Assets or Businesses.

- 2.26 <u>Insurance</u>. <u>Schedule 2.26</u> sets forth a true, correct and complete list of insurance policies, including policies by which the Companies, PPL or any of their Assets, employees, officers or directors or the Businesses have been insured since January 1, 2007 (the "<u>Liability Policies</u>"). The list includes for each Liability Policy the type of policy, policy number and name of insurer. The Sellers have made available in the Data Room true, accurate and complete copies of all Liability Policies, in each case, as amended or otherwise modified and in effect. <u>Schedule 2.26</u> describes any self-insurance arrangements affecting either Company or PPL. The Companies and PPL have since January 1, 2007 maintained in full force and effect with, to Sellers' Knowledge, financially sound and reputable insurers insurance with respect to their Assets and the Businesses, in such amounts and against such losses and risks as is customarily carried by Persons engaged in the same or similar business and as is required under the terms of any applicable Real Property Leases or other Contractual Obligations. Except as disclosed on <u>Schedule 2.26</u>, no insurer (a) has questioned, denied or disputed (or otherwise reserved its rights with respect to) the coverage of any claim pending under any Liability Policy other than denials of employee health claims or (b) has provided any notice of cancellation and, to Sellers' Knowledge, no insurer plans to cancel any Liability Policy or materially alter the coverage under any Liability Policy.
- 2.27 <u>Banking Facilities</u>. <u>Schedule 2.27</u> sets forth a true, correct and complete list of: (a) each bank, savings and loan or similar financial institution with which either Company or PPL has an account or safety deposit box or other arrangement, and any numbers or other identifying codes of such accounts, safety deposit boxes or such other arrangements maintained by such Company or PPL thereat; (b) the names of all Persons authorized to draw on any such account or to have access to any such safety deposit box facility or such other arrangement; and (c) any outstanding powers of attorney executed by or on behalf of either Company or PPL in connection with such accounts, safety deposit boxes or other arrangements.
- 2.28 <u>Powers of Attorney</u>. Except as set forth on <u>Schedule 2.28</u>, neither Company has any general or special powers of attorney outstanding (whether as grantor or grantee thereof).
- 2.29 <u>No Brokers</u>. Except for CAC, the fees and expenses of which are Closing Date Seller Expenses to be paid pursuant to <u>Section 1.5</u> or to be paid solely by Sellers following Closing, neither Company nor any Seller has Liability of any kind to, and is not subject to any claim of, any broker, finder or agent in connection with the Contemplated Transactions.

2.30 Securities Law Matters.

- 2.30.1 The shares of MasTec Common Stock that may be issued in connection with Earn-Out Payments to Sellers (the "Earn-Out Shares") are being acquired by the Sellers for the Sellers' own accounts and not with a view to, or intention of, distribution thereof in violation of the 1933 Act, or any applicable state securities laws, and the Earn-Out Shares will not be disposed of by any Seller in contravention of the 1933 Act or any applicable state securities laws.
- 2.30.2 Each Seller is an "accredited investor" as defined in Rule 501(a) under the 1933 Act, is sophisticated in financial matters and is able to evaluate the risks and benefits of the investment in the Earn-Out Shares and make an informed investment decision.
- 2.30.3 Each Seller has had an opportunity to ask questions and receive answers concerning MasTec and the Earn-Out Shares and has had full access to such other information concerning MasTec and the Earn-Out Shares as such Seller has requested or which has been filed by MasTec with the SEC.

- 2.30.4 Each Seller has discussed with and relied upon the advice of its independent legal counsel, tax and financial advisors with regard to the meaning and legal consequences of such Seller's representations and warranties contained herein and the considerations involved in making an investment in the Earn-Out Shares, and such Seller understands that MasTec is relying on such Seller's representations set forth herein in respect of the offer and sale of the Earn-Out Shares.
- 2.30.5 Each Seller acknowledges that the Earn-Out Shares are "restricted securities" as defined by Rule 144 promulgated under the 1933 Act, are not registered under the 1933 Act or any applicable state securities laws and are being issued by MasTec to the Sellers in reliance upon the Section 4(2) private placement exemption contained in the 1933 Act.
- 2.30.6 Each Seller understands that he or it must bear the economic risk of his or its acquisition of the Earn-Out Shares for an indefinite period of time because (i) the acquisition of Earn-Out Shares pursuant to this Agreement has not been registered under the 1933 Act and applicable state securities laws; and (ii) the Earn-Out Shares may therefore not be sold, transferred, pledged, or otherwise disposed of unless subsequently so registered or, in the opinion (reasonably satisfactory to MasTec) of Gardere (which is reasonably satisfactory for purposes of this opinion only) or other counsel (reasonably satisfactory to MasTec) registration under the 1933 Act or any applicable state securities laws is not required.
 - 2.30.7 Legends. It is understood that the certificates evidencing the Earn-Out Shares may bear one or all of the following legends:
 - (a) "These securities have not been registered under the Securities Act of 1933, as amended. They may not be sold, offered for sale, pledged or hypothecated in the absence of a registration statement in effect with respect to the securities under such Act or an opinion of counsel reasonably satisfactory to the Company that such registration is not required or unless sold pursuant to Rule 144 of such Act."
 - (b) Any legend required by Law, including any legend required by the Business Corporation Act of the State of Florida.

2.31 Surety Bonds.

2.31.1 The Companies have posted all deposits, letters of credit, trust funds, bid bonds, performance bonds, reclamation bonds, surety bonds and all such similar undertakings (collectively, "Surety Bonds") required to be posted in connection with their operations. Schedule 2.31 contains a true and complete list of all Surety Bonds posted by the Companies in connection with their operations including the name of each surety and the estimated cost of completion as of September 30, 2009 for the project or job secured by such Surety Bond.

- 2.31.2 Except as disclosed on Schedule 2.31, each Company is in compliance in all material respects with all Surety Bonds applicable to it, and the operation of its Business and the state of reclamation with respect to the Surety Bonds are "current" or in "deferred status" regarding reclamation obligations and otherwise are in compliance in all material respects with all applicable reclamation and other applicable Legal Requirements. Except as set forth on Schedule 2.31, neither Company is nor to Sellers' Knowledge will be required to obtain a substitute Surety Bond with respect to any of its Surety Bonds and the Surety Bonds will remain in effect on identical terms immediately following the Closing.
- 2.32 <u>Transport Operations</u>. Except as set forth in <u>Schedule 2.32</u>, Transport's sole business is and at all times has been the Transport Business. Transport does not provide, and has never provided, any services to any Person other than Precision; and Transport's sole customer is, and has always been, Precision.
- 2.33 <u>PPL</u>. Except as set forth on <u>Schedule 2.33</u>, PPL holds no assets, except for its Precision Units to be sold to the Buyer pursuant to this Agreement, and conducts no business or operations other than the employment of the Transferred Employees and the provision of the services of the Transferred Employees to Precision.

3. REPRESENTATIONS AND WARRANTIES OF THE BUYER AND MASTEC.

In order to induce the Sellers and the Companies to enter into and perform this Agreement and to consummate the Contemplated Transactions, the Buyer and MasTec, jointly and severally, hereby represent and warrant to the Sellers and the Companies as follows:

- 3.1 <u>Organization</u>. MasTec is duly organized, validly existing and in good standing under the laws of the State of Florida and has the corporate power and authority to own its properties and to carry on its business as it is now being conducted. Buyer is a limited liability company duly organized, validly existing and in good standing under the laws of the State of Wisconsin. Since formation, Buyer has been wholly-owned by MasTec and is an eligible entity as defined in Section 301.7701-3(a) of the Treasury Regulations that is disregarded as an entity for federal tax purposes under Treasury Regulation §301.7701-3.
- 3.2 <u>Power and Authorization</u>. The execution, delivery and performance by the Buyer of this Agreement and the consummation of the Contemplated Transactions are within the power and authority of the Buyer and have been duly authorized by all necessary action on the part of the Buyer, and no other corporate or other action on the part of the Buyer or any other Person is necessary to authorize the execution and delivery of this Agreement by the Buyer or the consummation by the Buyer of the Contemplated Transactions. This Agreement (a) has been duly executed and delivered by the Buyer and (b) is a legal, valid and binding obligation of the Buyer, Enforceable against the Buyer in accordance with its terms, except as limited by the Enforceability Exceptions.
- 3.3 <u>Authorization of Governmental Authorities</u>. Except for compliance with the HSR Act, no action by (including any authorization, consent or approval), or in respect of, or filing with, any Governmental Authority is required for, or in connection with, the valid and lawful (a) authorization, execution, delivery and performance by the Buyer of this Agreement or (b) the consummation of the Contemplated Transactions by the Buyer.

- 3.4 <u>Noncontravention</u>. Neither the execution, delivery nor performance by the Buyer of this Agreement nor the consummation of the Contemplated Transactions will:
 - (a) violate any provision of any Legal Requirement applicable to the Buyer;
 - (b) result in a breach or violation of, or default under, or give rise to a right for any third-party to terminate or any prepayment penalty under any Contractual Obligation of the Buyer;
 - (c) require any action by (including any authorization, consent or approval) or in respect of (including notice to), any Person under any Contractual Obligation, except as contemplated by <u>Section 6.8</u>; or
 - (d) result in a breach or violation of, or default under, the Buyer's Organizational Documents.
 - 3.5 No Brokers. The Buyer has no Liability of any kind to any broker, finder or agent with respect to the Contemplated Transactions.
- 3.6 <u>Securities Law Compliance</u>. MasTec has made all filings with the SEC that it is required to make under the 1933 Act and the Exchange Act (collectively the "<u>Public Reports</u>"). The Public Reports were prepared and comply in all material respects with the requirements of the 1933 Act and the Exchange Act, as applicable.
- 3.7 <u>MasTec Financial Statements</u>. The financial statements of MasTec (the "<u>MasTec Financial Statements</u>") included in the Public Reports were prepared in accordance with the books and records of MasTec and its Subsidiaries and have been prepared in accordance with GAAP. The MasTec financial statements fairly present in all material respects in accordance with GAAP the financial position of MasTec and its Subsidiaries as at the respective dates thereof and the results of the operations of MasTec and its Subsidiaries for the respective periods covered thereby.
- 3.8 <u>MasTec Material Adverse Effect</u>. Since the date of the latest MasTec Financial Statement included in the Public Reports, there has been no MasTec Material Adverse Effect.
- 3.9 No Actual Knowledge of Certain Breaches. None of Jose Mas, Pablo Alvarez, Alberto de Cardenas or Mike Love has actual knowledge of any breach of the representations and warranties set forth in Section 2.6 (Financial Statements).

4. PRE-CLOSING COVENANTS

4.1 <u>Access and Investigation</u>. Between the date of this Agreement and the Closing Date, Sellers and the Companies will, and will cause each of their respective Representatives to, (a) afford Buyer and its Representatives and prospective lenders or other financing sources and their Representatives (collectively, "<u>Advisors</u>"), upon reasonable advance notice and during regular business hours full and free access to the Companies' and PPL's personnel (including, for purposes of negotiating post-Closing employment and non-competition arrangements), properties, jobsites, Contractual Obligations, customers, books and records (including all Tax

records), and other documents and data, (b) furnish Buyer and its Advisors with and upload to the Data Room copies of all such Contractual Obligations, books and records, and other existing documents and data as Buyer and its Advisors may reasonably request, and (c) furnish the Buyer and its Advisors with and upload to the Data Room such additional financial, operating, and other data and information as Buyer or its Advisors may reasonably request; provided that (i) no investigation pursuant to this Section 4.1 shall affect or be deemed to modify any representation or warranty made by the Companies or the Sellers; (ii) such access does not unreasonably interfere with the normal business operations of the Companies; (iii) that all requests for access shall be directed to Murphy, Rooney, the Companies' counsel, or such other person as the Companies may designate from time to time; and (iv) the Companies shall not be required to provide access to customers until the Companies arrange meetings with such customers; provided that Sellers and the Companies shall use their commercially reasonable efforts to arrange such meetings, respectively.

4.2 Operation of the Companies and their Businesses

- 4.2.1 Ordinary Course. Except as expressly consented to in writing by Buyer between the date of this Agreement and the Closing Date (the "Pre-Closing Period"), or contemplated hereby, the Companies and PPL shall and the Sellers shall cause the Companies to act and carry on their respective Businesses solely in the Ordinary Course of Business and shall use commercially reasonable efforts to (i) maintain and preserve their business organization, assets, Governmental Authorizations and properties, (ii) preserve their business relationships with customers, strategic partners, suppliers, distributors and others having business dealings with them, (iii) continue to perform under all Disclosed Contracts, (iv) maintain all Insurance Policies set forth on Schedule 2.26 and (v) keep available the services of their present officers, employees and consultants.
- 4.2.2 <u>Negative Covenants</u>. To the extent permitted by applicable laws, without limiting the generality of the foregoing, except (i) as expressly consented to in writing by Buyer, which consent will not be unreasonably withheld or delayed, (ii) as contemplated by this Agreement, (iii) in the Ordinary Course of Business or (iv) as set forth on <u>Schedule 4.2.2</u>, during the Pre-Closing Period,
 - (a) neither Company nor PPL shall directly or indirectly do any of the following:
 - (i) declare, set aside or pay any cash or non-cash distributions or dividends, split, combine or reclassify any Equity Securities or issue or authorize the issuance of any other Equity Securities in respect of, in lieu of or in substitution for its Equity Securities or Debt; or purchase, redeem or otherwise acquire any Equity Securities;
 - (ii) issue, deliver, sell, grant, pledge or otherwise dispose of or encumber any Equity Securities;

- (iii) amend or adopt any amendments to either Company's Organizational Documents;
- (iv) acquire by merging or consolidating with, or by purchasing all or a substantial portion of the assets or any stock of, or by any other manner, any business or any Person or division thereof, or any assets that are material, in the aggregate, to either Company;
- (v) sell, lease, license, pledge, or otherwise dispose of or otherwise encumber or subject (or allow to become subject) to any Encumbrance, other than Permitted Encumbrances, any of its material properties or Assets, including any capital asset or related capital assets with a fair market value in excess of One Hundred Thousand Dollars (\$100,000);
 - (vi) knowingly or irrevocably waive any material right of the Companies under any Disclosed Contract;
- (vii) incur any Debt, (B) issue, sell or amend any Debt, (C) Guarantee or otherwise become Liable for any Debt of another Person, (D) make any material loans, advances or capital contributions to, or Investment in, any other Person, (E) modify or cancel any material third-party Debt owed to either Company, or (F) enter into any arrangement having the economic effect of the foregoing;
- (viii) make any single capital expenditure or other expenditures with respect to PP&E in excess of Two Hundred Fifty Thousand Dollars (\$250,000) in the aggregate;
- (ix) except as required to comply with applicable Laws or the terms of any Employee Plan, (A) adopt, enter into, terminate or amend any Employee Plan, (B) increase the Compensation or fringe benefits of (other than increases in the Ordinary Course of Business, or pay any bonus not required by an existing plan, arrangement or agreement to, any officer or employee of either Company or PPL, (C) grant any Equity Securities, or (D) take any action other than in the Ordinary Course of Business to fund or in any other way secure the payment of compensation or benefits under any Employee Plan;
- (x) except as may be required to comply with applicable Tax Laws, make, revoke, amend or change any election in respect of Taxes, file any amendment to a Tax Return, settle any claim or assessment in respect of Taxes, or consent to any extension or waiver of the limitation period applicable to any claim or assessment in respect of Taxes (except with respect to income Taxes);
 - (xi) threaten, commence, pay, discharge, settle or satisfy any Action;

- (xii) terminate or fail to renew any Governmental Authorization that is required for continued operations;
- (xiii) enter into, amend or modify any collective bargaining agreement or union contract with any labor organization or union or waive, release or assign any rights or claims under any such agreement or contract;
- (xiv) accelerate or defer any material obligation or payment by or to either Company, or not pay any accounts payable or other obligation of either Company when due, unless contested in good faith with full and complete appropriate reserves provided in the relevant Company's books and records;
- (xv) decrease or defer in any material respect, any required maintenance, repair or replacement of any PP&E or the level of costs expended in connection therewith;
 - (xvi) fail to maintain insurance at levels at least comparable to current levels or otherwise in a manner inconsistent with past practice;
- (xvii) discontinue any line of business or adopt a plan of complete or partial liquidation, dissolution, merger, consolidation, recapitalization or other similar reorganization;
- (xviii) take any action that would or would reasonably be expected to (A) result in a representation or warranty of the Sellers and/or the Companies becoming untrue, (B) result in any of the conditions set forth in Section 6 or Section 7 not being satisfied, or (C) otherwise prevent or materially delay or materially impair its ability to consummate the Contemplated Transactions on the terms contemplated by this Agreement;
- (xix) at any time within the 90-day period prior to the Closing Date, effectuate a "<u>plant closing</u>" or "<u>mass layoff</u>" as those terms are defined in WARN or any similar state Law, affecting in whole or in part any site of employment, facility, operating unit or employee of either Company or PPL;
- (xx) make any change to its accounting methods, principles or practices (including with respect to Taxes) or to the Financials or to the working capital policies applicable to either Company, except as required by GAAP;
- (xxi) except for entering into any non-exclusive license agreements in the Ordinary Course of Business, transfer or grant to any third party any rights with respect to any Intellectual Property;

- (xxii) form any Subsidiary or acquire any Equity Security of any Person; or
- (xxiii) authorize or enter into an agreement to do anything prohibited by the foregoing; and
- (b) no Seller shall directly or indirectly do any of the following:
- (i) sell, lease, pledge, or otherwise transfer, dispose of or otherwise encumber or subject (or allow to become subject) to any Encumbrance any Units;
- (ii) take any action that would or would reasonably be expected to (A) result in a representation or warranty of the Sellers and/or the Companies becoming untrue, (B) result in any of the conditions set forth in <u>Section 6</u> or <u>Section 7</u> not being satisfied, or (C) otherwise prevent or materially delay or materially impair its ability to consummate the Contemplated Transactions on the terms contemplated by this Agreement; or
 - (iii) authorize or enter into an agreement to do anything prohibited by the foregoing.

4.3 Commercially Reasonable Efforts; Notification; Hart-Scott-Rodino Act Filings.

4.3.1 (a) Upon the terms and subject to the conditions set forth in this Agreement, except as otherwise provided in this Section 4.3, each of the Sellers, the Companies, MasTec and Buyer agrees to use commercially reasonable efforts to take, or cause to be taken, all actions, and to do, or cause to be done, and to assist and cooperate with the other parties in doing, all things necessary, proper or advisable to consummate and make effective, in the most expeditious manner practicable, the Contemplated Transactions, including (i) preparing and filing of all forms, registrations and notices required to be filed, obtaining of all necessary actions or nonactions, waivers, consents and approvals from Governmental Authorities and the making of all necessary registrations and filings (including filings with Governmental Authorities) and the taking of all commercially reasonable steps as may be necessary to obtain an approval or waiver from, or to avoid an Action by, any Governmental Authority, (ii) obtaining all necessary consents, approvals or waivers from third parties, (iii) defending of any Actions challenging this Agreement or the consummation of the Contemplated Transactions, including seeking to have any stay or temporary restraining order entered by any court or other Governmental Authority vacated or reversed and (iv) the execution and delivery of any additional instruments necessary to consummate the Contemplated Transactions. Notwithstanding the foregoing or any other provision of this Agreement to the contrary, in no event shall Buyer be obligated to, and the Companies shall not without the prior written consent of Buyer, agree or proffer to divest or hold separate, or enter into any licensing, business restriction or similar arrangement with respect to, any assets (whether tangible or intangible) or any portion of any business of Buyer, its Affiliates or the Companies, nor shall Buyer agree to take any action which would reasonably be expected to materially reduce the Earn-Out Payments without the prior w

4.3.2 If not accomplished prior to the execution of this Agreement, without limiting the generality of the foregoing, as soon as may be reasonably practicable, and in no event later than 10 Business Days after the date hereof, Buyer and the Companies each shall file with the FTC and the Antitrust Division of the DOJ Notification and Report Forms relating to the Contemplated Transactions as required by the HSR Act. Subject to applicable Laws and the preservation of any applicable attorney-client privilege, Buyer and the Companies each shall promptly (i) supply the other with any information that may be required in order to effectuate such filings, (ii) supply any additional information that reasonably may be required by the FTC or the DOJ, and each will use reasonable efforts to obtain a waiver of the applicable waiting period and (iii) make any further filings pursuant thereto that may be necessary, proper or advisable in connection therewith. Buyer and the Companies shall share equally all filing fees and other charges for the filing under the HSR Act and shall each be responsible for all costs and expenses for obtaining their own respective consents, approvals or waivers from third parties and for defending any actions against them challenging this Agreement or the consummation of the Contemplated Transactions.

4.4 Acquisition Proposals.

4.4.1 The Companies and Sellers shall not, nor shall they permit or authorize any of their Representatives to, directly or indirectly, initiate, solicit, encourage or otherwise facilitate (including by way of furnishing confidential information) any inquiries or the making of any proposal or offer, with respect to (i) any merger, reorganization, share exchange, business combination, recapitalization, consolidation, liquidation, dissolution or similar transaction involving either Company or PPL, (ii) any sale, lease, exchange, mortgage, pledge, transfer or purchase of a significant portion of the Assets or any Asset material to the Businesses (other than the sale of Inventory or obsolete PP&E, which has been replaced by all substitute PP&E necessary to continue to operate the Businesses in the Ordinary Course of Business) or Equity Securities of either Company or PPL, (iii) any purchase or sale of, or tender offer or exchange offer for Equity Securities of PPL or Units (any such proposal or offer being hereinafter referred to as an "Acquisition Proposal"). The Companies and the Sellers shall not, nor shall they permit or authorize any of their Representatives to, directly or indirectly, (a) engage in any negotiations concerning, or provide any confidential information or data to, or have any discussions or conversations with, any Person relating to an Acquisition Proposal, or otherwise facilitate any effort or attempt to make or implement or accept an Acquisition Proposal, or (b) enter into any letter of intent or similar document contemplating, or enter into any agreement with respect to, an Acquisition Proposal. Without limiting the foregoing, it is agreed that any violation of the restrictions set forth in this paragraph by any Representative, whether or not such Person is purporting to act on behalf of either Company and Seller or otherwise, shall be deemed to be a breach of this Section 4.4 by the Companies and Sellers.

4.4.2 The Companies and Sellers will promptly (and in any event within two (2) days) notify Buyer in writing of the existence of any proposal, discussion, negotiation or inquiry received by either Company, any Seller or any of their Representatives with respect to any Acquisition Proposal, and the Companies and Sellers will immediately communicate to Buyer the terms of any proposal, discussion, negotiation or inquiry which it or they may receive (including a copy of any such proposal) and the identity of the Person making such proposal or inquiry or engaging in such discussion or negotiation.

- 4.4.3 The Companies and Sellers will, and will cause their Representatives to, immediately cease and cause to be terminated any existing activities, discussions or negotiations with any Person(s) conducted heretofore with respect to any Acquisition Proposal. In addition, the Companies shall promptly request that each Person who has heretofore received information in connection with such Person's consideration of an Acquisition Proposal return or destroy all confidential information heretofore furnished to such Person by or on behalf of either Company or any Seller. The Companies shall not release any third party from, or waive any provision of, any confidentiality or standstill agreement to which either Company is a party. The Companies and the Sellers agree that they will take the necessary steps to promptly inform their Representatives of the obligations undertaken by the Companies and Sellers in this Section 4.4.
- 4.5 <u>Bank Accounts; Powers of Attorney</u>. As of the Closing, to the extent requested by Buyer, Sellers and the Companies shall cause Buyer's designees to be added, and the Companies' respective designees to be removed, as signatories with respect to each of the Companies' respective bank accounts and to terminate any powers of attorney.
- 4.6 Notice and Cure. Between the date hereof and the Closing Date, the Companies and Sellers will notify Buyer in writing (where appropriate and only with respect to matters occurring after the date hereof, through updates to the Disclosure Schedules) of, and contemporaneously will provide Buyer with and upload to the Data Room true, accurate and complete copies of any and all information or documents relating to, and will make commercially reasonable efforts to cure before the Closing, any event, transaction or circumstance, that existed or occurred on, prior to or after the date of this Agreement, as soon as practicable after it becomes known to either Company or any Seller, that causes or will cause any representation, warranty, covenant or agreement of either Company and/or the Sellers under this Agreement to be breached, that renders or will render untrue any representation or warranty of either Company and/or the Sellers contained in this Agreement as if the same were made on or as of the date of such event, transaction or circumstance, or that make the timely satisfaction of any condition to Closing impossible or unlikely. Both Sellers and Buyer shall have the right until the Closing Date to amend or supplement their respective Disclosure Schedules with respect to any matter. No notice (or updates to Disclosure Schedules) given pursuant to this Section 4.6 shall have any effect on the representations, warranties or agreements contained in this Agreement for purposes of determining satisfaction of any condition, whether a breach or default has occurred, or the termination or indemnification rights of the parties provided by this Agreement or otherwise. Notwithstanding the foregoing, no party hereto shall have any Liability for indemnification pursuant to this Agreement on account of a breach of a representation or warranty to the extent that (a) such matter or event is disclosed in a revised Disclosure Schedule has the right to terminate this Agreement based upon such breach but elects to proceed with the

4.7 <u>Consultation</u>. To the extent permitted by applicable Law, between the date hereof and the Closing Date, the Companies will consult with management of Buyer with a view to informing them as to the operation and management of the Companies and PPL.

4.8 Cooperation Regarding: Financing. Companies and the Sellers acknowledge that Buyer's financing for the Contemplated Transactions may be secured by security granted on Closing against certain assets of the Companies and agree that prior to the Closing, the Companies and the Sellers and their Representatives shall provide to Buyer and its Affiliates and Representatives all cooperation reasonably requested by Buyer that is necessary or reasonably required in connection with such financing (at Buyer's sole cost and expense), including the following: (i) causing the Sellers and the Companies' other senior officers to participate in meetings, presentations, due diligence sessions with prospective lenders, investors and rating agencies; (ii) providing information reasonably necessary for Buyer to prepare any pledge and security documents, currency or interest hedging agreement, and other definitive financing documents, provided that no obligation of the Companies under any such document or agreement shall be effective until the Closing; (iii) facilitating the pledging of collateral, provided that no pledge shall be effective until the Closing; (iv) furnishing on a confidential basis to Buyer and its financing sources, as promptly as practicable, such financial and other pertinent information regarding the Companies as may be reasonably requested by Buyer; (v) providing requested officer's certificates and representation letters; and (vi) providing information reasonably requested by Buyer to enable Buyer to secure solvency opinions, in each of the foregoing cases as may be necessary and customary in connection with a financing; provided that, notwithstanding anything in this Agreement to the contrary, until the Closing occurs, the Companies shall not (1) be required to pay any commitment or other similar fee, (2) have any Liability under any loan agreement or any related document or any other agreement or document related to Buyer's financing, or (3) be required to incur any other Liability in connection with the financing. Buyer and its Representatives shall be permitted to disclose information as necessary in connection with Buyer's financing subject to reasonable arrangements to protect the confidentiality of Sellers' and the Companies' confidential information. Notwithstanding anything to the contrary contained in this Agreement, MasTec shall not be obligated to effectuate any Financing and any Financing, to the extent effectuated, shall be upon terms and conditions that are acceptable to MasTec in its sole and absolute discretion.

4.9 Interim Financial Statements. During the Pre-Closing Period, the Sellers and Companies shall provide Buyer (a) within twenty (20) days after the date hereof, the consolidated balance sheet of the Companies, LandCo and PPL as of September 30, 2009 (the "Most Recent Balance Sheet,") and the related statements of income and cash flows of the Companies, LandCo and PPL for the nine months then ended (together with the Most Recent Balance Sheet, the "Interim Financials") and (b) within 30 days after the end of each calendar month, the unaudited consolidated balance sheet, income statement and statement of cash flows as of the end of, or for, such month as applicable, for the Companies and PPL, and (ii) within 30 days after the end of each fiscal quarter, the unaudited consolidated balance sheet and the unaudited statements of income and of cash flow for such fiscal quarter for the Companies and PPL (in each case of clauses (i) and (ii) prepared in accordance with Precision GAAP (subject to normal year-end adjustments, the effect of which will not individually or in the aggregate, be materially adverse and the absences of footnotes and consistent with the preparation of the Financial Statements) except as set forth in Schedule 2.6.2. Upon delivery of the foregoing financial statements to Buyer, the representations set forth in Section 2.6 shall be deemed to have been made as to such financial statements as of the date of this Agreement.

- 4.10 <u>Termination of Certain Employees</u>. Effective as of the Closing Date, PPL shall terminate the employment of all those employees named on <u>Schedule 4.10</u> (the "<u>Transferred Employees</u>"). Buyer shall not be liable to any Seller or other Person for any severance obligations, benefit obligations or any other Liabilities in connection with the termination of the Transferred Employees or other Person's employment with PPL, all of which Liabilities shall be the responsibility of PPL.
- 4.11 <u>Assistance</u>. Sellers shall assist Precision in obtaining the employment of the Transferred Employees, and Sellers shall cooperate with and shall make available to Buyer, to the extent permitted by applicable Laws, all information and documents as may be necessary to assist and coordinate Precision's employment of the Transferred Employees.
- 4.12 <u>Guarantees by Sellers</u>. Sellers have guaranteed certain Debt of PPL and the Companies (the "<u>Seller Guarantees</u>"). If the Seller Guarantees are not released at or prior to Closing, the Companies and MasTec will indemnify Sellers from any liability related to or arising out of the Seller Guarantees, and following Closing shall use commercially reasonable efforts to have the Seller Guarantees released.
- 4.13 <u>Consents for Assumption of Debt</u>. MasTec and Buyer shall use commercially reasonable efforts to assist Sellers in securing any necessary consent to the assignment of Contractual Obligations; provided that MasTec and its Affiliates, other than the Companies, shall have no obligation to assume or guarantee such Contractual Obligations.

4.14 Confidentiality.

4.14.1 The Companies and their Representatives have delivered and may prior to Closing deliver to MasTec, Buyer and their Representatives certain information about the properties, employees, finances, businesses and operations of the Companies. All information, whether written or oral, about the Companies furnished by the Companies and their Representatives is referred to in this Section 4.14 as "Evaluation Material." Evaluation Material also includes all notes, analyses, compilations, studies, interpretations or other documents prepared by MasTec, Buyer or their Representatives which contain, reflect or are based upon, in whole or in part, the information furnished to MasTec, Buyer or their Representatives. Evaluation Material does not include, however, information which (a) is or becomes generally available to the public other than as a result of a disclosure by MasTec, Buyer or their Representatives in violation of this Section 4.14, (b) is in MasTec's possession on a non-confidential basis prior to its disclosure by the Companies or their Representatives, (c) was or becomes available to MasTec, Buyer or their Affiliates on a non-confidential basis from a Person (other than the Companies or their Representatives) who MasTec or Buyer reasonably believed is not prohibited from disclosing such information to MasTec, Buyer or their Affiliates by a legal, contractual or fiduciary obligation to the Companies, (d) was independently developed by MasTec, Buyer or their Affiliates without reference to the Evaluation Material, or (e) is, upon the advice of Buyer's legal counsel, required to be disclosed by Law (as defined below).

4.14.2 Subject to Section 4.14.3, unless otherwise agreed in writing by the Sellers' Representative, Buyer agrees, except as required by Law, to keep confidential and not to disclose or reveal any Evaluation Material to any Person other than Buyer's Representatives. MasTec and Buyer agree they and their Representatives have used and, unless the Closing occurs, will use the Evaluation Material only in connection with their consideration of whether to purchase the Companies and will not otherwise use it in their business or otherwise. If this Agreement terminates, MasTec and Buyer agree to destroy all Evaluation Material in either their possession or control or the possession or control of their Representatives.

4.14.3 If Buyer or any of its Representatives is required by, Law to disclose any Evaluation Material, Buyer will, to the extent permitted by Law, provide the Sellers' Representative with prompt notice of such request or requirement in order to enable the Companies to seek an appropriate protective order or other remedy at the Companies sole cost and expense, to consult with Buyer with respect to the Companies taking steps to resist or narrow the scope of such request or legal process, or to waive compliance, in whole or in part, with the terms of this Section 4.14. If such protective order or other remedy is not obtained, or the Companies waive compliance, in whole or in part, with the terms of this Section 4.14, the Buyer and its Representative will use reasonable efforts to disclose only that portion of the Evaluation Material that is legally required to be disclosed. If Buyer and its Representatives shall have complied fully with the provisions of this Section 4.14.3, the Companies agree that such disclosure may be made by Buyer or its Representatives without any Liability to Sellers or the Companies whatsoever.

4.14.4 Unless the Closing occurs, MasTec and Buyer agree that neither they nor any of their Affiliates will, for a period of two years after the date hereof, hire or solicit for employment any Key Employee.

POST-CLOSING COVENANTS.

5.1 Confidentiality.

5.

5.1.1 Each Seller acknowledges that the success of the Companies after the Closing Date depends upon the continued preservation of the confidentiality of certain information possessed by such Seller, that the preservation of the confidentiality of such information by such Seller is an essential premise of the bargain between such Seller and the Buyer, and that the Buyer would be unwilling to enter into this Agreement in the absence of this Section 5.1.1. Accordingly, each Seller hereby agrees with the Buyer that such Seller and its Representatives will not, and that such Seller will cause its Affiliates not to, at any time on or after the Closing Date, directly or indirectly, without the prior written consent of the Buyer, disclose or use, any confidential or proprietary information involving or relating to the Businesses or the Companies, except to attorneys, accountants and authorized representatives of Sellers and to employees of Sellers in the Ordinary Course of Business, including (1) customer and supplier information, including lists of names and addresses of customers and suppliers of the Companies or their Affiliates, (2) business plans and strategies, compensation plans, compensation information, sales plans and strategies, pricing and other terms applicable to transactions between existing and prospective customers, suppliers or business associates, (3) market research and databases, sources of leads and methods of obtaining new business, and methods of purchasing,

marketing, selling, performing and pricing products and services employed by the Companies, (4) information concerning the configuration and architecture, technical data, networks, methods, practices, standards and capacities of the Companies' information systems, Company Software and Company Technology, (5) information identified as confidential and/or proprietary in internal documents of the Companies or PPL and (6) all information that would be a trade secret under any applicable Law; provided, however, that the information subject to the foregoing provisions of this sentence will not include any information generally available to, or known by, the public (other than as a result of disclosure in violation hereof); and provided, further, that the provisions of this Section 5.1.1 will not prohibit any retention of copies of records or disclosure (a) required by any applicable Legal Requirement so long as reasonable prior notice is given to the Buyer of such retention and disclosure and a reasonable opportunity is afforded to the Buyer to contest the same or (b) made in connection with the enforcement of any right or remedy relating to, or the performance of any obligation arising under, this Agreement or the Contemplated Transactions. Each Seller agrees that it will be responsible for any breach or violation of the provisions of this Section 5.1.1 by any of its Representatives.

- 5.1.2 Notwithstanding the foregoing, each Seller and each of its Representatives may disclose to any and all Persons, without limitation of any kind, the Tax treatment and Tax structure of the Contemplated Transactions and all materials of any kind (including opinions or other Tax analyses) that are provided to such Seller relating to such Tax treatment and Tax structure.
- 5.1.3 Prior to and following the Closing Date, each Seller hereby agrees that such Seller and its Representatives will, and that such Seller will cause its Affiliates to, hold in strict confidence all, and not divulge or disclose, or buy or sell any MasTec securities while in possession of, any material non-public information of any kind concerning MasTec and its direct and indirect Subsidiaries, taken as a whole.

5.2 <u>Restrictive Covenants</u>.

- 5.2.1 Noncompetition. Each Seller on his or its own behalf and on behalf of each of his or its Related Entities, covenants that from the Closing Date through the later of the seven year anniversary of the Effective Date or 24 months after such Seller's termination of employment with Precision or its Affiliates, neither he or it nor his or its Related Entities will for his, its or their own account or jointly with another, directly or indirectly, for or on behalf of any Person, as principal, agent or otherwise:
 - (a) (A) own, manage or control, or become engaged or serve as a shareholder, bondholder, creditor, officer, director, partner, member, employee, agent, consultant, advisor, contractor with, employer or representative of, or in any similar capacity, or (B) give financial, technical or other assistance to, otherwise invest or have a financial interest in, or in exchange for compensation otherwise associate with any Person, business or enterprise that competes directly or indirectly with the Companies or their Businesses (a "Competitive Business") anywhere within the continental Unites States (the "Territory"); provided, however, that each Seller and his or its Related Entities may passively hold up to 2% of the outstanding publicly-traded securities of a Person engaged in the Precision Business for investment purposes only;

- (b) recruit, induce, solicit for employment or in any manner attempt to recruit, induce or solicit any Key Employee or person employed by PPL (including any Transferred Employee or any non-union employee of either Company and any person who was a non-union employee of either Company within the twelve (12) months prior to the attempted recruitment, inducement or solicitation) whether or not such employment with PPL or either Company is (or was) pursuant to a written contract and whether or not such employment is (or was) at will;
- (c) solicit, contact or deal with: (i) any Person that is at such time, or at any time during the five-year period preceding the Closing Date or the five-year period preceding the latest Earn-Out Payment, whichever is more recent, was, a customer, supplier or business associate of either Company or PPL, or (ii) any Person from whom either Company or PPL solicited business or with whom either Company or PPL discussed in a material manner a potential business relationship at any time during the five-year period preceding the Closing Date, in each case, for the purpose of offering or providing services or products which are competitive with services or products provided by either Company;
- (d) cause or seek to cause to be terminated or adversely affected, or otherwise interfere with, any agreement or arrangement of any kind to which either Company is a party or from which it benefits; or
- (e) seek to interfere with or adversely affect the ongoing relationships between either Company, on the one hand, and its suppliers, customers and professional and business contacts, on the other.
- 5.2.2 Acknowledgement. Each Seller acknowledges that the Companies conduct business throughout the Territory. Accordingly, each Seller agrees that the Territory is reasonable to protect the legitimate business interests of the Buyer. Each Seller agrees that any reduction to the Territory would seriously undermine the efficacy of this Section 5.2 and the protections that it is intended to provide. Each Seller acknowledges and agrees that the covenants contained in this Section 5.2 are essential elements of this Agreement and that but for these covenants the Buyer would not have agreed to purchase the Units. Each Seller further expressly agrees and acknowledges that (a) the confidentiality, nonsolicitation and non-competition covenants contained in this Agreement (i) are reasonable and necessary for the protection of the Buyer with respect to the covenants' respective stated purposes, time, scope and geographic area, and are supported by adequate consideration; (ii) are necessary for the protection of the Buyer's legitimate business interests, including the trade secrets, goodwill, and relationships with customers and suppliers purchased by the Buyer pursuant to this Agreement; and (iii) are not unduly restrictive of any rights of the Sellers; and (b) each Seller has sufficient employment alternatives and sufficient assets, taking into account the Purchase Price received and to be received by such Seller, such that such Seller does not have to compete with either Company or impermissibly use

either Company's confidential, proprietary, trade secret information described in this Agreement during the term of this <u>Section 5.2</u> in order to earn a living. Except as set forth in <u>Section 5.2.7</u>, the existence of any claim or cause of action against the Buyer by any Seller, whether predicated on the Buyer's breach of this Agreement or otherwise, shall not constitute a defense to the enforcement by the Buyer of the covenants contained in this <u>Section 5.2</u>.

- 5.2.3 <u>Injunction</u>. The Sellers agree that a violation of the terms, provisions, covered obligations, duties and conditions described in this <u>Section 5.2</u> will cause irreparable damage to the Buyer for which money damages or other legal remedies would not be an adequate remedy. Accordingly, the Sellers acknowledge and hereby agree that in the event of any breach or threatened breach by any Seller of any of their covenants or obligations set forth in this <u>Section 5.2</u>, the Buyer shall be entitled, in any court in the United States or otherwise having jurisdiction, to an injunction or injunctions, without the posting of any bond, to prevent or restrain breaches or threatened breaches of this <u>Section 5.2</u>, and to specifically enforce the terms and provisions of this <u>Section 5.2</u> to prevent breaches or threatened breaches of, or to enforce compliance with, the covenants and obligations of this <u>Section 5.2</u>, and to specifically enforce the terms and provisions of this <u>Section 5.2</u> to prevent or restrain breaches or threatened breaches of this <u>Section 5.2</u>, and to specifically enforce the terms and provisions of this <u>Section 5.2</u> to prevent breaches or threatened breaches of this <u>Section 5.2</u>, and to specifically enforce the terms and provisions of this <u>Section 5.2</u> to prevent breaches or threatened breaches of, or to enforce compliance with, the covenants and obligations of each Seller under this <u>Section 5.2</u>. The Sellers further agree that (x) by seeking the remedies provided for in this <u>Section 5.2.3</u>, the Buyer shall not in any respect waive its right to seek any other form of relief that may be available under this Agreement (including monetary damages), and (y) nothing set forth in this <u>Section 5.2.3</u> shall require Buyer to institute any Action for (or limit Buyer's right to institute any Action for) specific performance under this <u>Section 5.2.3</u> or anything set forth in this <u>Section 5.2.3</u> restrict or limit any party's right to pursue any other remedies under this Agreement that may be available
- 5.2.4 <u>Severability</u>. In the event that any provision of this <u>Section 5</u> shall be held to be invalid or unenforceable for any reason whatsoever, such invalidity or unenforceability shall not affect any other provision of this Agreement and the remaining provisions hereof shall remain in full force and effect, to the extent permitted by Law, and any court of competent jurisdiction may so modify the objectionable provision as to make it valid, reasonable and enforceable. All parties hereto hereby agree that each and every restrictive covenant as set forth in <u>Section 5.1.1</u> and <u>Section 5.2.1(a)</u> through and including <u>Section 5.2.1(e)</u> is a separate and distinct restrictive covenant, designated to operate under different factual circumstances, and that the invalidity of one of said covenants shall not affect the validity and/or enforceability of the other covenants.
- 5.2.5 <u>Allocation of Purchase Price Has No Effect</u>. Each Seller acknowledges and agrees that a portion of the Purchase Price will be allocated (as set forth on <u>Schedule 1.3</u>)as consideration to the covenants set forth in this <u>Section 5.2</u>. Notwithstanding the foregoing, each Seller acknowledges that Buyer would not have entered into the Contemplated Transactions without the covenants set forth in this <u>Section 5.2</u>; and if any covenant set forth in this <u>Section 5.2</u> is breached by

any Seller, each Seller acknowledges and agrees that such allocation shall not be deemed to be a measure of the damages that would result from such a breach, and each Seller agrees that at no time shall he or it argue or in any way assert in any Action that such consideration is a measure of the damages resulting from such breach.

- 5.2.6 "Related Entity". Solely for purposes of this Section 5.2, "Related Entity" means with reference to any Person, another Person controlled by, under the control of or under common control with, that Person. A Person shall be deemed to control another Person if the controlling Person possesses, directly or indirectly, the power to direct or cause the direction of the management, policies and practices of the other Person, whether through ownership of voting securities or otherwise. Without limitation, any director, executive officer or beneficial owner of 50% or more of the equity of a Person shall, for purpose of this Agreement, be deemed to control the other Person.
- 5.2.7 <u>Default on Earn-Out Payments</u>. If Buyer fails to pay any Earn-Out Payment when due and does not cure such failure to pay within 30 days after the receipt of written notice from the Sellers' Representative of such failure to pay, then the restrictive covenants contained in <u>Section 5.2</u> will terminate and be of no force and effect with respect to any Seller no longer employed by Buyer or its Affiliates if Buyer thereafter fails to pay such unpaid Earn-Out Payment on or prior to the 90th day after the end of such 30 day period. For clarity, a dispute regarding the amount of an Earn-out Payment shall not in and of itself be a failure to pay an Earn-Out Payment when due, as any disputed portion of an Earn-Out Payment shall not be due until such time as the disputed amount is resolved in accordance with <u>Section 1.10.6</u>, and any failure to make an Earn-Out Payment as permitted by <u>Section 1.10.11</u> or <u>Section 9.8</u> shall not be a failure to pay an Earn-Out Payment when due as no such payment is required at such time.
- 5.3 <u>Cooperation Regarding Financial Statements</u>. The Sellers shall use their good faith efforts at Buyer's cost and expense to assist and cooperate with the efforts of the Buyer and its accountants and auditors to prepare and audit any financial statements (including pro forma financial statements) that the Buyer will be required to prepare, file or furnish pursuant to any applicable securities laws or exchange requirements, including the 1933 Act, the Exchange Act, the rules and regulations of the SEC promulgated thereunder, and any rules or regulations of the New York Stock Exchange or other stock exchange, or under any other applicable Laws.
- 5.4 <u>Litigation Support</u>. If and for so long as the Buyer or either Company is actively contesting or defending against any Action in connection with any fact, situation, circumstance, status, condition, activity, practice, plan, occurrence, event, incident, action, failure to act or transaction existing or occurring on or prior to the Closing Date involving either Company or Business, the Sellers will cooperate in the contest or defense and provide such testimony as may be reasonably necessary in connection with the contest or defense, at the cost and expense of the Buyer (unless and to the extent the Buyer is entitled to indemnification therefor hereunder, in which event such costs and expenses shall be borne by the Sellers).
- 5.5 <u>Access to Books and Records</u>. From and after the Closing, the Buyer shall provide the Sellers and their advisors with reasonable access (for the purpose of examining and copying at the Sellers' expense) during normal business hours and upon reasonable advance notice, to the books and records relating to the Businesses prior to the Closing (the "<u>Books and Records</u>") and,

until the 36-month anniversary of the Closing, reasonable access, during normal business hours and upon reasonable advance notice to the former financial and accounting employees of the Businesses, in each case to the extent necessary for the purpose of (a) compliance with tax requirements or (b) providing information relating to Third Party Claims reasonably required by the Sellers in order to satisfy their indemnification obligations pursuant to this Agreement; provided, however, that such access shall not unreasonably interfere with the normal business operations of the Buyer or its Affiliates or the Businesses and shall be provided pursuant to the Buyer's reasonable confidentiality and security safeguards. Unless otherwise consented to in writing by the Sellers, the Buyer shall not destroy, alter or otherwise dispose of any of the Books and Records in advance of the earlier of the date of destruction or disposition pursuant to the Buyer's document retention policies or seven years following the Closing without first offering to surrender to the Sellers such Books and Records or any portion thereof that the Buyer may intend to destroy, alter or dispose of.

- 5.6 <u>LLC Indemnification Provisions</u>. Buyer agrees to cause each Company to perform its obligations under such Company's Organizational Documents then in effect to provide its members, officers, employees and agents with all rights of indemnification thereunder and at least to the extent required by applicable Law; provided, however, that none of MasTec, Buyer and the Companies shall be required to indemnify any member, officer, employee or agent in respect of any liability or obligation of such person to MasTec or Buyer and which liability or obligation arises under this Agreement.
- 5.7 Employment and Benefit Arrangements. Buyer shall cause Precision to offer employment to the Transferred Employees, which offer shall be delivered to the Transferred Employees prior to the Closing; and Sellers shall cooperate with and shall make available to Buyer, to the extent permitted by applicable Laws, all information and documents as may be necessary to assist and coordinate Precision's employment of the Transferred Employees. Buyer shall cause Precision to offer such employees salary and bonus packages that are, in the aggregate, substantially comparable to the salary and bonus packages in place prior to the Closing. Precision shall cause the Transferred Employees to be covered under the MasTec's benefit plans, arrangements or agreements and shall provide such employees with credit for service with PPL thereunder for purposes of eligibility and vesting. The Buyer shall waive any applicable waiting periods, pre-existing conditions or actively-at-work requirements (other than those under MasTec's 401(k) plan) and shall give the Transferred Employees and their eligible dependants credit under the new coverage or benefit plans for deductibles, co-payments and out-of-pocket payments that have been paid during the year in which such coverage or other plan modification occurs. Within ten (10) days after the Closing, the Sellers shall provide Buyer with a schedule that reflects those deductibles, co-payments and out of those pocket payments. MasTec shall cause MasTec's 401(k) plan (or the 401(k) Plan of Buyer or an Affiliate of MasTec) to accept eligible rollover contributions from PPL's 401(k) plan from the Transferred Employees who become participants in MasTec's 401(k) plan (or the 401(k) plan for the the 401(k) pl

Company to retain any Transferred Employee for any period of time after the Closing Date and subject to the requirements of applicable Law, Buyer, MasTec and the Companies reserve the right, at any time after the Closing Date, to terminate such employment and to amend, modify or terminate any terms or conditions of employment, including any Employee Plan, program, policy, practice or arrangement or the compensation or working conditions of the Transferred Employees. MasTec and Buyer agree to cause Precision (i) to pay the Transferred Employees 2009 bonuses in an amount of at least \$500,000, (ii) to pay field employees 2009 bonuses in an amount equal to at least \$2,000,000, (iii) to provide Transferred Employees with aggregate accrued vacations with a value of at least \$25,000 as of their date of hire, (iv) to provide to the applicable employees 2009 profit sharing payments of at least \$200,000, and (v) to pay the stand-by-pay accrued as a current Liability in the Actual Closing Balance Sheet.

- 5.8 Warranty Repairs. With respect to all warranty repairs that are performed following the Closing Date by the Companies, but for which Sellers are responsible pursuant to Section 9.1.3 or any other provision of Section 9. Buyer agrees to cause the Companies to perform any such warranty repair, and the Losses of the Buyer Indemnified Persons with respect to each such warranty repair, shall in all cases be equal to the sum of (a) actual material cost, plus (b) actual labor cost, plus (c) actual fuel and transportation costs, and plus (d) all other costs actually incurred by the Companies, Buyer and their Affiliates in connection with such warranty repair. This Section 5.8, except as set forth in the previous sentence with respect to the determination of the amount of Losses, does not in any manner alter Sellers' obligations pursuant to Section 9 with respect to all such warranty repairs.
- 5.9 <u>Workers' Compensation Insurance</u>. Effective upon the Closing, Buyer and Precision shall cause the Zurich American Insurance Company workers' compensation insurance policy to be terminated. Sellers shall be responsible for, indemnify and hold the Companies and Buyer harmless from any amounts due Zurich American Insurance Company in respect of the termination of such insurance policy. If Precision receives a refund or rebate as a retroactive premium adjustment in respect of any amount of premium paid to Zurich American Insurance Company for such insurance policy for 2009, Precision shall immediately pay such refund or rebate to the Sellers in accordance with <u>Schedule 1.2</u>, such payment to be due within five (5) Business Days thereafter.
- 5.10 <u>Lien Releases and Termination Statements</u>. Not later than the Closing Date, Sellers shall procure the release of any liens and the filing of termination statements in respect of each of the UCC financing statements disclosed in Section C of <u>Schedule 2.9.1</u>.

6. CONDITIONS PRECEDENT TO BUYER'S OBLIGATION TO CLOSE

Buyer's obligation to purchase the Units and to take the other actions required to be taken by Buyer at the Closing is subject to the satisfaction, at or prior to the Closing, of each of the following conditions (any of which may be waived by Buyer, in whole or in part):

6.1 Sellers' and the Companies' Performance.

- 6.1.1 <u>Covenants; Etc.</u> All of the covenants and obligations that Sellers and/or the Companies are required to perform or to comply with pursuant to this Agreement at or prior to the Closing must have been performed and complied with in all material respects.
- 6.1.2 <u>Representations and Warranties</u>. The representations and warranties of the Sellers contained in this Agreement that are qualified as to materiality shall have been true and correct, and those not so qualified shall have been true and correct in all material respects on the Effective Date, except to the extent such representations and warranties relate to an earlier date (in which case such representations and warranties qualified as to materiality shall have been true and correct, as those not so qualified shall have been true and correct in all material respects, on and as of such earlier date).
 - 6.1.3 <u>Documents, Etc.</u> Each document required to be delivered by Sellers pursuant to <u>Section 1.9.2(a)</u> must have been delivered.
- 6.2 No Actions. Since the date of this Agreement, there must not have been commenced or threatened against Buyer, either Company, Sellers or against any Person affiliated with Buyer, either Company, or Sellers any Action (a) involving any challenge to, or seeking material damages or other material relief in connection with, any of the Contemplated Transactions, or (b) that may have the effect of preventing, making illegal, or otherwise materially interfering with any of the Contemplated Transactions that in either such case of (a) of (b) has not been fully resolved without additional cost or Liability to Buyer, which has not been fully paid by Sellers.
- 6.3 No Claim Regarding Units or Sale Proceeds. There must not have been made or threatened by any Person any claim asserting that such Person other than Sellers (a) is the holder or the beneficial owner of, or has the right to acquire or to obtain beneficial ownership of, any Units of, or any other Equity Securities in, either Company, or (b) is entitled to all or any portion of the Purchase Price payable for the Units that in either such case of (a) of (b) has not been fully resolved without additional cost or Liability to Buyer, which has not been fully paid by Sellers.
- 6.4 No Prohibition. No Governmental Authority shall have enacted, issued, promulgated, enforced or entered into any Law or Governmental Order and no other legal or regulatory restraint or prohibition shall be in effect, in either case, which has the effect of making the Contemplated Transactions illegal or that otherwise prohibits or requires the payment of any damages as a result of the Contemplated Transactions, and no Action in which any of the foregoing is sought shall be pending.
 - 6.5 No Material Adverse Effect. Since the date of this Agreement, no Material Adverse Effect shall have occurred.
- 6.6 <u>HSR Act</u>. Any waiting period (and any extension thereof) under the HSR Act applicable to the Contemplated Transactions shall have expired or been terminated, or approval of the relevant Governmental Authority otherwise obtained.

- 6.7 Escrow Agreement. The Escrow Agent shall have executed and delivered to Buyer the Escrow Agreement and a receipt for the Escrow Fund.
- 6.8 <u>Third-Party Consents</u>. The Companies shall have received and delivered to Buyer the third-party consents listed on <u>Schedule 2.5</u>, without additional cost or Liability to Buyer, which has not been fully paid by Sellers and MasTec shall have received any consent required under the Second Amended and Restated Loan and Security Agreement dated July 29, 2008, by and among MasTec, Inc., certain of its subsidiaries, Bank of America, N.A., as collateral and administrative agent and General Electric Capital Corporation, as syndication agent.
- 6.9 <u>Financing</u>. MasTec shall have consummated a financing providing net proceeds to MasTec of not less than \$75 million on such terms as may be acceptable to MasTec in its sole and absolute discretion (the "Financing").

7. CONDITIONS PRECEDENT TO SELLERS' OBLIGATION TO CLOSE

Sellers' obligations to sell the Units and to take the other actions required to be taken by Sellers at the Closing is subject to the satisfaction, at or prior to the Closing, of each of the following conditions (any of which may be waived by the Sellers' Representative, in whole or in part):

7.1 Buyer's Performance.

- 7.1.1 <u>Covenants; Etc.</u> All of the covenants and obligations that Buyer is required to perform or to comply with pursuant to this Agreement at or prior to the Closing must have been performed and complied with.
- 7.1.2 <u>Representations and Warranties</u>. The representations and warranties of the Buyer and MasTec contained in this Agreement that are qualified as to materiality shall have been true and correct, and those not so qualified shall have been true and correct in all material respects on the Effective Date, except to the extent such representations and warranties relate to an earlier date (in which case such representations and warranties qualified as to materiality shall have been true and correct, as those not so qualified shall have been true and correct in all material respects, on and as of such earlier date).
- 7.1.3 <u>Documents, Etc.</u> Buyer must have delivered each of the documents required to be delivered by Buyer pursuant to <u>Section 1.9.2(b)</u> and must have irrevocably initiated the payment of the <u>Section 1.9.2(b)</u> Cash Purchase Price required to be made by Buyer pursuant hereto.
- 7.2 No Prohibition. No Governmental Authority shall have enacted, issued, promulgated, enforced or entered into any Law or Governmental Order and no other legal or regulatory restraint or prohibition shall be in effect, in either case, which has the effect of making the Contemplated Transactions illegal or that otherwise prohibits or requires the payment of any damages as a result of the Contemplated Transactions or that would impair Buyer's ability to make the Earn-Out Payments, and no Action in which any of the foregoing is sought shall be pending.

- 7.3 <u>HSR Act</u>. Any waiting period (and any extension thereof) under the HSR Act applicable to the Contemplated Transactions shall have expired or been terminated, or approval of the relevant Governmental Authority otherwise obtained.
 - 7.4 No MasTec Material Adverse Effect. Since the date of this Agreement, no MasTec Material Adverse Effect shall have occurred.

8. TERMINATION

- 8.1 Termination Events. This Agreement may, by written notice given prior to or at the Closing, be terminated:
 - 8.1.1 by mutual written consent of the Buyer and the Sellers' Representative;
 - 8.1.2 on or after December 16, 2009 by either the Sellers' Representative or the Buyer if the Closing has not occurred on or before such date;
- 8.1.3 on or after the 37th day following the date of this Agreement, by either the Sellers' Representative or the Buyer if the Closing has not occurred on or before such date; provided, that the terminating party shall give the other party not less than two (2) Business Days notice of its intent to terminate this Agreement pursuant to this Section 8.1.3 and within such two (2) Business Day period will discuss with the other party or parties the basis upon which such notice was given and shall explore possible alternatives; provided, however, that the decision to terminate shall rest solely in the discretion of the notifying party; and provided, further that the right to terminate this Agreement pursuant to this Section 8.1.3 shall not be available to a party that is then in breach of any representation, warranty, covenant, agreement or obligation contained in this Agreement;
- 8.1.4 by the Sellers' Representative, upon written notice, if (i) a MasTec Material Adverse Effect has occurred, (ii) Buyer shall have breached any agreement, obligation or covenant such that the condition set forth in Section 7.1 would not be satisfied, or (iii) all of the conditions set forth in Section 6 are satisfied (excluding conditions that, by their terms, cannot be satisfied until the Closing, but which would be reasonably capable of being satisfied at Closing) and Buyer fails to satisfy its obligations to be carried out at Closing under Section 1; provided, that in the case of (ii) if the breach of Buyer's agreement, obligation or covenant is curable through the exercise of Buyer's commercially reasonable efforts, then the Sellers' Representative may not terminate this Agreement pursuant to this Section 8.1.4 for thirty (30) days after the Sellers' Representative shall have given written notice of such breach to Buyer (so long as Buyer continues to use commercially reasonable efforts to cure the breach during such period), it being understood that the Sellers' Representative may not terminate this Agreement pursuant to this Section 8.1.4 if Buyer cures such breach within such thirty (30) day period;
- 8.1.5 by Buyer, upon written notice to the Sellers' Representative, if (i) a Material Adverse Effect has occurred, (ii) Sellers or either Company shall have breached any agreement, obligation or covenant such that the condition set forth in Section 6.1 would not be satisfied, (iii) all of the conditions set forth in Section 7 are satisfied (excluding conditions that, by their terms, cannot be satisfied until the Closing, but which would be reasonably capable of being satisfied at Closing) and Sellers and the Companies fail to satisfy all of their obligations to be carried out at Closing under Section 1, or (iv) if the Interim Financials, in

Buyer's sole and absolute discretion, are unsatisfactory in any respect; <u>provided</u> that in the case of (ii) if the breach of Sellers' or the Companies' agreement, obligation or covenant is curable through the exercise of commercially reasonable efforts, then Buyer may not terminate this Agreement pursuant to this <u>Section 8.1.5</u> for thirty (30) days after Buyer shall have given written notice of such breach to the Sellers' Representative (so long as Sellers and/or the Companies continue to use commercially reasonable efforts to cure such breach during such period), it being understood that Buyer may not terminate this Agreement pursuant to this <u>Section 8.1.5</u> if Sellers or the Companies cure such breach within such thirty (30) day period; or

8.1.6 by Buyer or the Sellers if there shall be any Legal Requirement that makes consummation of the purchase of the Units illegal or otherwise prohibited, or if any Governmental Order enjoining Buyer or Sellers from consummating the purchase of the Units is entered and such order shall have become final and nonappealable; provided that the party seeking to terminate this Agreement pursuant to this provision shall have used all commercially reasonable efforts to remove or vacate such order.

8.2 Effect of Termination. If this Agreement is terminated pursuant to Section 8.1, this Agreement shall forthwith become void and of no further force and effect, except for the provisions of Section 4.14 (Confidentiality), this Section 8.2, Section 11 and Section 12 and the Buyer Confidentiality Agreement which shall survive such termination; provided, however, that nothing herein shall relieve any party from Liability for any breach of any representation, warranty, covenant or agreement contained in this Agreement. Notwithstanding anything to the contrary contained in this Agreement, except solely with respect to a breach of Section 4.14 (Confidentiality), neither MasTec nor Buyer shall have any liability or obligation (under Section 9 of this Agreement, at common law or otherwise) with respect to this Agreement or the Contemplated Transactions if the Financing shall not have been consummated. Unless the Financing has been consummated, in no event, if the Closing does not occur, whether or not this Agreement has been terminated pursuant to any provision hereof, shall MasTec, Buyer or any of their Affiliates or Related Entities, either individually or in the aggregate, be subject to any Liability whatsoever relating to or arising out of this Agreement or the Contemplated Transactions, including breaches by Buyer of any representations, warranties, covenants (other than Section 4.14 (Confidentiality)), agreements or obligations contained in this Agreement, and in no event shall Sellers, the Sellers' Representative, the Companies or any of their Affiliates or Related Entities seek equitable relief or seek to recover any money damages from MasTec, Buyer or any of their Affiliates or Related Entities, except solely with respect to a breach of Section 4.14 (Confidentiality).

9. INDEMNIFICATION.

9.1 <u>Indemnification by the Sellers</u>. Subject to the limitations set forth in this <u>Section 9</u>, from and after Closing, the Sellers, jointly and severally, will indemnify and hold harmless the Buyer and each of its Affiliates (including, following the Closing, the Companies), and the Representatives and Affiliates of each of the foregoing Persons (each, a "<u>Buyer Indemnified Person</u>"), from, against and in respect of any and all Losses whether or not involving a Third Party Claim, incurred or suffered by Buyer Indemnified Persons or any of them as a result of, arising out of or directly or indirectly relating to:

9.1.1 any breach of, or inaccuracy in, any representation or warranty made by the Sellers in Section 2;

- 9.1.2 any breach or violation of any covenant or agreement of the Sellers and/or either Company to the extent required to be performed or complied with by the Sellers (including under this <u>Section 9</u>) or the Companies in this Agreement; provided that with respect to either Company, only to the extent required to be performed or complied with by such Company at or prior to Closing;
- 9.1.3 any Liabilities (other than Liabilities under the Contractual Obligations of the Company, except for Liabilities of the Companies or Sellers of any kind or nature (i) the existence of which would conflict with or constitute a breach of a representation, warranty or agreement of Sellers or the Companies contained herein or (ii) by reason of or arising out of any default or breach occurring prior to the Effective Date (or that would be a default or breach but for the passage of time or giving of notice, or both) or any unperformed, warranty repair obligation of the Companies that exists prior to the Effective Date under such Contractual Obligations but that is unperformed as of the Effective Date whether or not such Contractual Obligation is disclosed on the Companies' Disclosure Schedules) of the Companies or the Sellers, of any nature whatsoever, whether accrued, absolute, contingent or otherwise, arising from or relating to the operations or activities of the Companies prior to the Effective Date, except to the extent any such liability is reflected as a current Liability in the Closing Balance Sheet or included as Actual Indebtedness pursuant to Section 1.8;
 - 9.1.4 except to the extent any such Liability is reflected as a current Liability in the Actual Closing Balance Sheet, any Disclosed Liabilities; or
 - 9.1.5 the matters set forth on Schedule 9.1.5.
- 9.1.6 The Sellers shall not be liable for any indemnification obligations under this Agreement with respect to, or to hold the Buyer Indemnified Persons harmless from, any claim or group of related claims, unless such claim or group of related claims involves Losses in excess of \$20,000 (a "De Minimis Claim"), nor shall any De Minimis Claim be applied to or considered for purposes of calculating the aggregate amount of the Buyer Indemnified Persons' Losses; provided that the limitations provided in this sentence shall only apply with respect to an aggregate of up to \$250,000 of De Minimis Claims and if all De Minimis Claims in the aggregate exceed \$250,000 (such excess, "Excess De Minimis Claims"), the Sellers shall, subject to the other limitations provided in this Agreement, indemnify and hold the Buyer Indemnified Persons harmless from the amount of such excess over \$250,000 and the amount of such excess shall be applied to and considered for purposes of calculating the aggregate amount of the Buyer Indemnified Persons' Losses needed to meet the Deductible and the Liability Cap. In addition, the Sellers shall not be liable for the indemnification obligations pursuant to Section 9.1.1, Section 9.1.3, Section 9.1.4 or Section 9.1.5, or to hold the Buyer Indemnified Persons harmless, until the aggregate amount of Losses, including Losses with respect to Excess De Minimis Claims, with respect to matters referred to in Section 9.1.1, Section 9.1.3, Section 9.1.4 and Section 9.1.5 exceeds One Million Dollars (\$1,000,000) excluding in calculating such amount any De Minimis Claims but not Excess De Minimis Claims (the "Deductible"), after which the Sellers will be jointly and severally responsible for all Losses, including Losses with

respect to Excess De Minimis Claims, in excess of the Deductible up to a maximum aggregate amount of Losses equal to Twenty-Two Million Five Hundred Thousand Dollars (\$22,500,000) (the "<u>Liability Cap</u>"). Notwithstanding anything herein to the contrary, (A) none of the De Minimis Claim limitation, the Deductible or the Liability Cap will apply (i) in the case of fraud or intentional misrepresentation or (ii) with respect to Sellers indemnification obligations pursuant to <u>Section 9.1.2</u>, and (B) the Liability Cap will not apply (i) with respect to any breach of <u>Sections 2.1</u> (Organization and Predecessors), <u>2.2</u> (Capitalization and Title), <u>2.3</u> (Power and Authorization), 2.5 (Noncontravention), <u>2.9.1</u> (Ownership of Assets), <u>2.17</u> (Employee Benefit Plans), <u>2.18</u> (Environmental Matters), or <u>2.29</u> (No Brokers) or (ii) with respect to Sellers' indemnification obligations pursuant to <u>Section 10</u>.

- 9.1.7 Notwithstanding anything contained herein to the contrary, for purposes of determining whether there has been a breach and the amount of any Losses that are the subject matter of a claim for indemnification or reimbursement hereunder, each representation and warranty in this Agreement and schedules and exhibits hereto shall be read without regard and without giving effect to any Materiality Qualifier contained in such representation or warranty which has the effect of making such representation and warranty less likely to be breached (as if such word or words were deleted from such representation and warranty).
- 9.2 <u>Indemnity by the Buyer and MasTec</u>. Subject to the limitations set forth in this <u>Section 9</u>, from and after Closing, the Buyer and MasTec, jointly and severally, will indemnify and hold harmless the Sellers and the Sellers' Affiliates, and the Representatives and Affiliates of each of the foregoing Persons (each, a "<u>Seller Indemnified Person</u>"), from, against and in respect of any and all Losses incurred or suffered by Seller Indemnified Persons or any of them as a result of, arising out of or relating to, directly or indirectly:
 - 9.2.1 any breach of, or inaccuracy in, any representation or warranty made by the Buyer and MasTec in Section 3; or
 - 9.2.2 any breach or violation of any covenant or agreement of the Buyer or MasTec (including under this <u>Section 9</u>) or any covenant or agreement of the Companies to the extent required to be performed or complied with by the Companies after the Closing Date, in either case in or pursuant to this Agreement.
- 9.3 <u>Time for Claims</u>. No claim may be made or suit instituted seeking indemnification pursuant to <u>Sections 9.1.1</u> or <u>9.2.1</u> for any breach of, or inaccuracy in, any representation or warranty or <u>Section 9.1.3</u> for any Liability unless a written notice describing such breach or inaccuracy in reasonable detail in light of the circumstances then known to the Indemnified Person, is provided to the Indemnifying Person:
 - 9.3.1 at any time, in the case of any breach of, or inaccuracy in, the representations and warranties set forth in <u>Sections 2.1</u> (Organization and Predecessors), <u>2.2</u> (Capitalization and Title), <u>2.3</u> (Power and Authorization), <u>2.5</u> (Noncontravention), <u>2.9.1</u> (Ownership of Assets), <u>2.29</u> (No Brokers), <u>3.1</u> (Organization), <u>3.2</u> (Power and Authorization), <u>3.4</u> (Noncontravention) or <u>3.5</u> (No Brokers);

- 9.3.2 at any time, in the case of any claim or suit based upon fraud or intentional misrepresentation;
- 9.3.3 at any time prior to the ninetieth (90th) day after the expiration of the applicable statute of limitations (taking into account any tolling periods and other extensions) in the case of any breach of, or inaccuracy in, the representations and warranties set forth in $\underline{2.17}$ (Employee Benefit Plans), or $\underline{2.18}$ (Environmental Matters); and
- 9.3.4 at any time prior to last day of the eighteenth (18th) month anniversary of the Closing Date, in the case of any breach of, or inaccuracy in, any other representation and warranty or any Loss arising from Liabilities described in Section 9.1.1, 9.1.3 or 9.2.1; provided that if, at any time prior to the expiration of the respective survival period set forth in this Section 9.3 with respect to any particular representation or warranty, any Indemnified Party delivers to any Indemnifying Party a written notice alleging the existence of an inaccuracy in or breach of such representation or warranty and asserting a claim for Losses pursuant to Section 9.1 or Section 9.2, then the representation or warranty underlying the claim asserted in such notice and all indemnity obligations under this Section 9 related thereto shall survive until such claim is finally and fully resolved in accordance with this Agreement.

Claims for indemnification pursuant to any other provision of $\underline{\text{Sections 9.1}}$ and $\underline{9.2}$ or pursuant to $\underline{\text{Section 10}}$ are not subject to the limitations set forth in this $\underline{\text{Section 9.3}}$.

9.4 Third Party Claims.

- 9.4.1 Notice of Claim. If any third party notifies an Indemnified Person with respect to any matter (a "Third Party Claim") which may give rise to an Indemnity Claim against an Indemnifying Person under this Section 9, then the Indemnified Person will promptly give written notice to the Indemnifying Person; provided, however, that no delay on the part of the Indemnified Person in notifying the Indemnifying Person will relieve the Indemnifying Person from any obligation under this Section 9 except to the extent that the Indemnifying Person is actually prejudiced by the Indemnified Party's failure to give such notice in such a timely manner.
- 9.4.2 <u>Assumption of Defense, etc.</u> The Indemnifying Person will be entitled to participate in the defense of any Third Party Claim that is the subject of a notice given by the Indemnified Person pursuant to <u>Section 9.4.1</u>. In addition, the Indemnifying Person will have the right to assume the defense of the Indemnified Person against the Third Party Claim with counsel reasonably satisfactory (taking into account the jurisdiction in which the claim is brought, the materiality of the claim and the experience levels and other qualifications of counsel available to defend such claim) to the Indemnified Person so long as (a) the Indemnifying Person gives written notice to the Indemnified Person within fifteen days after the Indemnified Person has given notice of the Third Party Claim that the Indemnifying Person will indemnify the Indemnified Person from and against the entirety of any and all Losses the Indemnified Person may suffer resulting from, arising out of, relating to, in the nature of, or caused by the Third Party Claim, (b) the Third Party Claim involves only money damages and does not seek an injunction or other equitable relief against the Indemnified Person, (c) the Indemnified Person has not been advised by counsel that an actual or potential conflict exists between the Indemnified Person and the Indemnifying Person in connection with the defense of the

Third Party Claim, (d) the Third Party Claim does not relate to or otherwise arise in connection with Taxes (provided, that with respect to Taxes, this Section 9.4.2 shall not affect the Sellers' right to defend any claim of a Governmental Authority as permitted by Section 10.3.4(b)) or any criminal or regulatory enforcement Action and (e) the Indemnifying Person conducts the defense of the Third Party Claim actively and diligently. The Indemnified Person may retain separate co-counsel at its sole cost and expense and participate in the defense of the Third Party Claim; provided, however, that the Indemnifying Person will pay the fees and expenses of separate co-counsel retained by the Indemnified Person that are incurred prior to the Indemnifying Person's assumption of control of the defense of the Third Party Claim.

- 9.4.3 <u>Limitations on Indemnifying Person</u>. The Indemnifying Person will not consent to the entry of any judgment or enter into any compromise or settlement with respect to the Third Party Claim without the prior written consent of the Indemnified Person, which consent will not be unreasonably withheld, conditioned or delayed, unless such judgment, compromise or settlement (a) provides for the payment by the Indemnifying Person of money as sole relief for the claimant, (b) results in the full and general release of all Buyer Indemnified Persons or Seller Indemnified Persons, as applicable, from all Liabilities arising or relating to, or in connection with, the Third Party Claim and (c) involves no finding or admission of any violation of Legal Requirements or the rights of any Person and has no effect on any other claims that may be made against the Indemnified Person.
- 9.4.4 Indemnified Person's Control. If the Indemnifying Person does not deliver the notice contemplated by clause (a) of Section 9.4.2 within 15 days after the Indemnified Person has given notice of the Third Party Claim, or otherwise at any time fails to conduct the defense of the Third Party Claim actively and diligently, the Indemnified Person may defend, and may consent to the entry of any judgment or enter into any compromise or settlement with respect to, the Third Party Claim with the consent of the Indemnifying Person, which consent shall not be unreasonably delayed, conditioned or withheld. If such notice and evidence is given on a timely basis and the Indemnifying Person conducts the defense of the Third Party Claim actively and diligently but any of the other conditions in Section 9.4.2 is or becomes unsatisfied, the Indemnified Person may defend, and may consent to the entry of any judgment or enter into any compromise or settlement with respect to, the Third Party Claim with the consent of the Indemnifying Person, which consent shall not be unreasonably delayed, conditioned or withheld. In the event that the Indemnified Person conducts the defense of the Third Party Claim pursuant to this Section 9.4.4, the Indemnifying Person will (a) advance the Indemnified Person promptly and periodically for the costs of defending against the Third Party Claim (including attorneys' fees and expenses) and (b) remain responsible for any and all other Losses that the Indemnified Person may incur or suffer resulting from, arising out of, relating to, in the nature of or caused by the Third Party Claim to the fullest extent provided in this Section 9.
- 9.4.5 <u>Reasonable Cooperation</u>. The party not in control of the prosecution or defense of a Third Party Claim will reasonably cooperate with the other party in the conduct of the prosecution or defense of such Third Party Claim.

- 9.5 Sellers' Representative. For purposes of this Section 9, it is understood and agreed that the Sellers' Representative shall act on behalf of all Sellers.
- 9.6 <u>Knowledge and Investigation</u>. Except as provided in <u>Section 4.6</u> or in this <u>Section 9</u>, the right of any Buyer Indemnified Person or Seller Indemnified Person to indemnification pursuant to this <u>Section 9</u> will not be affected by any investigation conducted or knowledge acquired (or capable of being acquired) at any time, whether before or after the execution and delivery of this Agreement or the Closing, with respect to the accuracy of any representation or warranty, performance of or compliance with any covenant or agreement, referred to in <u>Sections 9.1</u> or <u>9.2</u>, or otherwise.
- 9.7 <u>Remedies Cumulative</u>. The rights of each Buyer Indemnified Person and Seller Indemnified Person under this <u>Section 9</u> are cumulative, and each Buyer Indemnified Person and Seller Indemnified Person, as the case may be, will have the right in any particular circumstance, in its sole discretion, to enforce any provision of this <u>Section 9</u> without regard to the availability of a remedy under any other provision of this <u>Section 9</u>.
- 9.8 <u>Right of Setoff</u>. Upon notice to the Sellers specifying in reasonable detail the basis therefor and after the release of all amounts from the Escrow Account or, if prior to such release, if Indemnity Claims are pending for the full amount of the Escrow Account, the Buyer may set off any amount to which it may be entitled (after taking into account the limitations provided in this <u>Section 9</u>) from the Sellers against amounts otherwise payable, if any, under <u>Section 1.10</u> or <u>Section 9.1.5</u>. The exercise of such right of setoff by the Buyer in good faith, whether or not ultimately determined to be justified, will not constitute an event of default under this Agreement or any other agreement between the Buyer or any of its Affiliates and the Sellers. Neither the exercise or failure to exercise such right of setoff will constitute an election of remedies or limit the Buyer in any manner in the enforcement of any other remedies that may be available to it. Notwithstanding the foregoing, if the Buyer sets off an amount pursuant to this <u>Section 9.8</u> and it is later finally mutually determined by the parties or pursuant to <u>Section 11.15</u> that such setoff amount was in excess of the amount which the Sellers were required to indemnify the Buyer, then the Buyer shall pay interest to the Sellers on such excess amount at the Prime Rate as published by the Wall Street Journal on the date such setoff is made.
- 9.9 Companies' Indemnification Obligations. The obligation of the Companies' to indemnify Buyer and the other Buyer Indemnified Persons shall terminate in all respects upon the Closing; provided that for the avoidance of doubt the parties agree that the termination of the Companies' indemnification obligations pursuant to this Section 9.9 shall have no effect on or in any way limit Buyer's or any other Buyer Indemnified Person's rights to indemnification from the Sellers, including Buyer's and the other Buyer Indemnified Persons' rights to offset Losses that are incurred by or imposed on Buyer or any other Buyer Indemnified Person against any amounts otherwise payable pursuant to Section 1.10. In addition, the Sellers' rights to seek contribution or payment of any amount from either Company for any indemnification obligations that they are required to make to Buyer or any other Buyer Indemnified Person shall also terminate in all respects upon the Closing. If either Company suffers, incurs or otherwise becomes subject to any Losses as a result of or in connection with any inaccuracy in or breach of any representation, warranty, covenant or obligation of the Companies or Sellers hereunder, then (without limiting any of the rights of either Company, Buyer or any other Buyer Indemnified Persons as Indemnified Persons) Buyer shall also be deemed, by virtue of its ownership of the Equity Securities of the Companies, to have incurred Losses as a result of and in connection with such inaccuracy or breach but in any case the total amount both Buyer and the Companies may recover shall not exceed the amount of Losses.

9.10 <u>Time of Payment of Claims</u>. Except as otherwise set forth in this <u>Section 9</u>, any amount owing by any Person pursuant to this <u>Section 9</u> shall be paid within five (5) Business Days after determination of such amount.

9.11 Insurance and Other Sources of Reimbursement. The amount of any Losses for which indemnification is provided under this Section 9 shall be reduced by (i) the insurance proceeds received with respect to any such Losses and (ii) any other amount, if any, recovered from third parties (as a result of indemnification, contribution, guarantee or otherwise) by the Indemnified Party (or its Affiliates) with respect to any Losses less in the case of (i) and (ii) all reasonable costs of the Indemnified Party to collect such proceeds and any increase in insurance premiums resulting from such recovery (net of any reductions in such premiums which have occurred between the Closing and the date of the underlying Claim, other than reductions which have been previously fully offset pursuant hereto). For five years following Closing, Buyer will cause the Companies to maintain the insurance policies set forth on Schedule 9.11; provided, however, that Buyer may terminate such insurance policies if it or any of its Affiliates provides equivalent substitute coverage, either through individual policies or MasTec's insurance program. If any Indemnified Party shall have been indemnified in full pursuant to this Section 9 with respect to any Losses, such Indemnified Party shall, assign to such Indemnifying Party (to the extent of any indemnification payment received by the Indemnified Party) any claim which such Indemnified Party may have under any applicable insurance policy or other Contract which provides coverage for such Losses to the extent of such indemnification payment. Such Indemnified Party shall use its commercially reasonable efforts (at the expense of the Indemnifying Party) to collect under such insurance policy or other Contract; provided, however, that such commercially reasonable efforts shall not require any Indemnified Party to litigate any insurance claim or assert any insurance coverage claim that could reasonably be expected to result in the loss of, or material increase in the cost of, future coverage, or substantially reduce the coverage available as a result of such claim. If any Indemnified Party shall have received any indemnification payment from an Indemnifying Party pursuant to this Section 9 with respect to any Losses and the Indemnified Party has been indemnified in full with respect to such Losses (after application or other insurance proceeds as set forth in the first sentence of this paragraph) and has or shall subsequently have received insurance proceeds or other amounts with respect to such Losses, which when added together with all other insurance (less all reasonable costs of collection and any increase in insurance premiums resulting from such recovery) and other proceeds and indemnification amounts received is in excess of the amount of the Losses (without reduction for insurance or other proceeds), then such Indemnified Party shall promptly pay over to the Indemnifying Party such excess, but not in excess of the amount previously so paid by the Indemnifying Party.

9.12 <u>Purchase Price Adjustment</u>. Payments received by any party pursuant to this <u>Section 9</u> shall be treated by the parties as an adjustment to the Purchase Price for Tax purposes.

- 9.13 Effect of Computation of Actual Net Working Capital. Except for payments of amounts owed pursuant to Section 1.8, notwithstanding anything to the contrary contained in this Agreement, no Buyer Indemnified Person shall have any right to indemnification under Section 9.1 or Section 10 with respect to any Loss or alleged Loss to the extent of any Liability, reserve or accrual included as a Current Liability in the Actual Closing Balance Sheet or included as Actual Indebtedness pursuant to Section 1.8.
- 9.14 <u>Satisfaction of Indemnity Claims</u>. To the extent any amount is owed by Sellers to any Buyer Indemnified Person pursuant to <u>Section 9.1.1</u>, then such amount will first be satisfied from the Escrow Account in accordance with the terms of the Escrow Agreement. To the extent the Escrow Account is insufficient to satisfy all such amounts owed, any excess shall be satisfied in cash via wire transfer or same day funds check.

9.15 Limitations on Liability.

- 9.15.1 If a breach of the representations, warranties and/or covenants given by the Sellers in this Agreement is capable of remedy without Loss to Buyer, the Buyer Indemnified Parties shall only be entitled to compensation if the breach is not remedied within thirty (30) days after the date on which notice is served on the Sellers' Representative in accordance with <u>Section 9</u>.
- 9.15.2 Notwithstanding anything to the contrary set forth herein, no party shall be entitled to indemnification or reimbursement under any provision of this Agreement for any amount to the extent that such party has been indemnified or reimbursed for such amount pursuant to this <u>Section 9</u>.
- 9.15.3 Notwithstanding anything to the contrary set forth herein, if any Losses for which Sellers are obligated to indemnify Buyer pursuant to Section 9.1 arise in connection with or related to the work performed with respect to any Enbridge Project (job numbers 09-043, 09-054, 09-085, 09-094, 09-097, 09-002, 09-004, 09-010, 09-053, 09-084 and 09-010) during the period commencing July 1, 2009 and ending on the Effective Date, and any bonuses are paid by Enbridge and received by either Company with respect to such project, then Sellers' obligation to indemnify Buyer for such Losses shall be reduced (but not below zero) by an amount equal to the amount of such bonus received, *multiplied* by the percentage of the Enbridge Project completed between July 1, 2009 and the Effective Date.
- 9.16 Exclusive Remedy. Buyer acknowledges and agrees that, should the Closing occur, except in the case of fraud, Buyer Indemnified Parties' sole and exclusive remedy with respect to any and all claims relating to this Agreement or the Contemplated Transactions (but, for clarity, excluding the Company Agreements), whether based on contract, tort, strict liability, other laws or otherwise, including any breach or alleged breach of any representation, warranty, covenant or agreement shall be pursuant to the indemnification provisions set forth in this Section 9 and Section 10 and, as applicable, the injunction provisions set forth in Section 5.2.3, and hereby waives, from and after the Closing, to the fullest extent permitted under applicable Law, any and all other rights, claims and causes of action (other than claims of, or causes of action arising from, fraud) it may have against Sellers and their Affiliates arising under or based upon any Law or otherwise relating to this Agreement and the Contemplated Transactions.

9.17 Allocation of Purchase Price Is Not Agreement of Losses. Notwithstanding the fact that MasTec, Buyer and Sellers have determined that the Allocation reflects the fair market value of the respective categories of the Companies' Assets and the covenants set forth in Section 5.2.1, the parties agree that the respective allocations among such categories and covenants is not an agreement of the Losses that would be incurred by the Buyer Indemnified Persons if any Seller breaches (a) any of its representations, warranties or covenants set forth herein relating to any category of Assets, or (b) such covenant. Each of the parties agrees that at no time will such party argue or in any way assert in any Action that the respective allocations among the categories of the Companies' Assets and the covenant set forth in Section 5.2.1 reflect the parties' agreement of the measure of Losses that would result from any breach of this Agreement.

10. TAX MATTERS

- 10.1 <u>Representations and Obligations Regarding Taxes</u>. The Sellers jointly and severally represent and warrant to and agree with Buyer as follows, in each case except to the extent set forth on the relevant Companies' Disclosure Schedules (for purposes of this <u>Section 10</u>, the "Companies" includes the Companies and/or any entity that at any time has been a subsidiary of either Company, and either of the Companies may be referred to individually as a "Company"):
 - 10.1.1 Except as set forth on Schedule 10.1.1: (A) each Company has duly and timely filed all Tax Returns it was required to file; (B) all such Tax Returns were true, correct and complete in all material respects and all Taxes owed by such Company (whether or not shown on any Tax Return and whether or not any Tax Return was required) have been paid; (C) neither Company is the beneficiary of any extension of time within which to file any Tax Return; (D) each Company has maintained adequate provision for Taxes (excluding amounts deferred to take into account timing differences between book and tax) payable by such Company as of the Closing Date; (E) no claim has ever been made by a Governmental Authority in a jurisdiction where a Company does not currently file Tax Returns that such Company is or may be subject to taxation by that jurisdiction; and (F) there are no Liens on any of the assets of either Company that arose in connection with any failure (or alleged failure) to pay any Tax, except for Liens for Taxes not yet due.
 - 10.1.2 Except as set forth on Schedule 10.1.2, there is no material dispute or claim concerning any Tax Liability of either Company either (A) claimed or raised by any authority in writing or (B) to the Sellers' Knowledge based upon personal contact with any agent of such authority. Except as set forth on Schedule 10.1.2, neither Company has received from any Governmental Authority any written notice of proposed adjustment, deficiency, underpayment of Taxes or any other such notice which has not been satisfied by payment or been withdrawn, and no claims have been asserted relating to such Taxes against either Company. There is no dispute or claim concerning any Tax liability of either Company either claimed or raised by any Governmental Authority in writing.
 - 10.1.3 Except as set forth on Schedule 10.1.3, no Tax Return has been audited, or is currently the subject of audit. Each Company has made available in the Data Room correct and complete copies of all federal and foreign Tax Returns, examination reports, and statements of deficiencies assessed against, or agreed to by such Company since its formation. Neither Company has waived any statute of limitations in respect of Taxes or agreed to any extension of time with respect to any Tax assessment or deficiency.

- 10.1.4 Except as set forth on Schedule 10.1.4, neither Company is a party to any joint venture, partnership or other arrangement or contract that could be treated as a partnership for Federal income tax purposes. Neither Company has entered into any sale leaseback or leveraged lease transaction that fails to satisfy the requirements of Revenue Procedure 2001-28 (or similar provisions of foreign Law) or any safe harbor lease transaction. Neither Company has acquired nor does it own any assets that directly or indirectly secure any debt the interest on which is tax exempt under Section 103 of the Code.
- 10.1.5 Each Company has been since its formation an eligible entity as defined in Section 301.7701-3(a) of the Treasury Regulations, has been treated as a partnership for federal income tax purposes and has not elected to be taxed as an association for federal, state or local income tax purposes.
- 10.1.6 Except as set forth on Schedule 10.1.6, neither Company shall be required to include in a taxable period ending after the Closing Date taxable income attributable to income that accrued in a prior taxable period but was not recognized in any prior taxable period as a result of the installment method of accounting, the completed contract method of accounting, the long-term contract method of accounting, the cash method of accounting or Section 481 of the Code or any comparable provision of state, local, or foreign tax law or a "closing agreement" as described in Section 7121 of the Code.
 - 10.1.7 Neither Company has engaged in any "reportable transaction" as defined in Section 1.6011-4 of the Treasury Regulations.
- 10.1.8 Neither Company has received from any Governmental Authority any written notice of proposed adjustment, deficiency, underpayment of Taxes or any other such notice which has not been satisfied by payment or been withdrawn, and no claims have been asserted relating to such Taxes against either Company.
- 10.1.9 Except as set forth on <u>Schedule 10.1.9</u>, each Company and PPL has withheld and paid all Taxes required to have been withheld and to be paid prior to Closing in connection with amounts, allocable, paid or owing to any employee, independent contractor, creditor, member, partner or other third party. Neither Company has a non-accountable expense reimbursement arrangement within the meaning of Treasury Regulation Section 1.62-2(c).
 - 10.1.10 No Seller is a foreign person within the meaning of Section 1445 of the Code.

10.2 Covenants With Respect To Taxes.

10.2.1 Buyer shall (i) grant to the Sellers' Representative reasonable access to the Companies' books and records (including tax workpapers and returns and correspondence with tax authorities), including the right to take extracts therefrom and make copies thereof, to the extent that the books and records relate to the operations of the Companies during taxable

periods ending on or prior to or that include the Closing Date as may be necessary for Tax Return preparation, audit, or other financial reporting matters, and (ii) otherwise reasonably cooperate with the Sellers' Representative in connection with the preparation and filing of any Tax Return of the Company that the Sellers are responsible for and with any audit of Taxes that relates to the Businesses of the Companies prior to the Closing Date, in each case to the extent relevant to the Sellers' indemnification obligations under <u>Section 10.3</u>.

10.2.2 Buyer shall be responsible for preparing and filing, or causing the Companies to prepare and file, all Tax Returns of the Companies required to be filed for taxable periods ending after the Effective Date, or solely in the case of income Tax Returns, the Closing Date, and the Sellers shall be responsible, at their cost, for filing all Tax Returns of the Companies for taxable periods ending on or before the Effective Date ("Seller-Prepared Returns"), or solely in the case of income Tax Returns, the Closing Date. The Sellers shall furnish such Seller-Prepared Returns to the Buyer for its review, comment and approval (such approval not to be unreasonably withheld, conditioned or delayed) at least 30 days prior to the due date (or extended due date) for filing such Tax Returns, and such Tax Returns shall be prepared in accordance with past practice, except as required by Law. The Sellers shall pay all Taxes required to be paid with respect to Seller-Prepared Returns, except to the extent such Taxes have been previously paid, deposited or accrued in Actual Net Working Capital on the Actual Closing Balance Sheet. In the case of any Tax Returns other than income Tax Returns (a "Straddle Return") for taxable periods which begin before the Effective Date and end after the Effective Date (a "Straddle Period"), Buyer shall furnish such Tax Returns to the Seller's Representative for its review, comment and approval (such approval not to be unreasonably withheld, conditioned or delayed) at least 30 days prior to the due date (or extended due date) for filing such Tax Returns, and such Tax Returns shall be prepared in accordance with past practice, except as required by Law. The Sellers shall jointly and severally pay to Buyer within five (5) days after the date on which Taxes are paid with respect to a Straddle Return approved by the Sellers' Representative or as to which the refusal of Sellers' Representative to approve is unreasonable an amount equal to the portion of those Taxes that relates to the portion of the

10.2.3 For purposes of this Agreement, other than in the case of income Tax Returns, in the case of any period that begins before the Effective Date and ends after the Effective Date, any Tax based directly or indirectly on gross or net income or receipts or imposed in respect of specific transactions and any credits available with respect to any Tax, shall be allocated based on an interim closing of the books by assuming that the taxable period ended on the Effective Date, and any other Tax shall be allocated based on the number of days in the taxable period ending on the Effective Date divided by the total number of days in the taxable period. In the case of income Tax Returns, Straddle Periods shall be based on the Closing Date rather than the Effective Date.

10.2.4 No election shall be made to treat Buyer as an association for federal, state or local income tax purposes. On or before the Closing Date, MasTec shall furnish the Sellers' Representative with copies of the Organizational Documents relating to Buyer.

- 10.2.5 Buyer and MasTec hereby agree not to take any action or filing position inconsistent with treating the Escrow Fund as installment obligations under Section 453 of the Code.
- 10.2.6 Neither Buyer nor its Affiliates (including the Companies) may amend or cause the amendment of a Tax Return of the Companies, or file or amend any Tax election concerning the Companies, with respect to any period ending on or prior to the Closing Date or any period that begins before and ends after the Closing Date, without the written consent of the Sellers' Representative, which consent shall not be unreasonably withheld or delayed.
- 10.2.7 Any refund of Taxes of the Companies (including any interest with respect thereto) attributable to any period occurring on or before the Effective Date, or in the case of income Tax Returns, the Closing Date, or as allocated pursuant to Section 10.2.3 shall be the property of the Sellers, shall be paid promptly to the Sellers' Representative and, if received by Buyer or its Affiliate (including the Companies), shall be payable promptly to the Sellers' Representative.
- 10.2.8 Buyer shall not permit, and shall cause its Affiliates not to permit, the Companies to take any actions on the Closing Date that are out of the ordinary course of business of either Company, except as required or contemplated by this Agreement.
- 10.3 Indemnification for Taxes. Notwithstanding anything to the contrary in this Agreement (in particular Section 9):
- 10.3.1 The Sellers shall jointly and severally indemnify Buyer and its Affiliates, including, after the Closing, the Companies (each herein sometimes referred to as an "Buyer Indemnified Taxpayer" and together with Sellers, each an "Indemnified Taxpayer") against, and agree to jointly and severally protect, save and hold harmless each Buyer Indemnified Taxpayer from, any and all claims, damages, deficiencies and losses and all expenses, including reasonable attorneys', accountants' and experts' fees and disbursements (all herein referred to as "Tax Losses"), resulting from:
 - (a) a claim by any Governmental Authority for any Taxes, except to the extent such Taxes have been previously paid, deposited, or accrued in Actual Net Working Capital on the Actual Closing Balance Sheet, allocable to any period ending on or prior to the Effective Date, or in the case of income Taxes, the Closing Date, or allocable pursuant to Section 10.2.3 to that portion of any Straddle Period ending on the Effective Date, or in the case of income Taxes, the Closing Date, of (A) either the Companies or the Sellers and (B) any corporation that is or was a member of an Affiliated Group of which such Company was or is a member, or any liability of any of the foregoing for the Taxes of any Person, whether as a transferee or successor, by contract or otherwise;
 - (b) a claim by any Governmental Authority for any Taxes arising from the sale of the Units pursuant to this Agreement;

- (c) any breach of any representation, warranty or obligation set forth in this Section 10;
- (d) any Taxes owing in respect of item 3 on Schedule 10.1.1.

10.3.2 Tax Gross Up.

- (a) On or prior to February 28, 2010, MasTec shall pay to Sellers as additional cash consideration for the purchase of the Units (the "Gross-Up Payment") the amount determined pursuant to this Section 10.3.2. The Gross-Up Payment shall be equal to (i) the excess of (x) the federal, state and local income Tax of the Companies and the Sellers resulting from the operations of the Companies and the Contemplated Transactions computed for the period beginning on January 1, 2009 through and including the Closing Date (the "Closing Period") minus (y) the federal, state and local income Tax of the Companies and the Sellers resulting from the operations of the Companies and the Contemplated Transactions for the period commencing on January 1, 2009 through and including the Effective Date (the "Stub Period"), (for this purpose (1) including the taxable income or loss and deduction of Precision from Global Pipeline Partners, LLC ("GPP") as if the transfer of the interest in GPP were as of the end of each of the Closing Period and the Stub Period, (2) taking into account the Tax basis increases as a result of the allocation of the taxable income of the Companies for such periods in calculating the gain on the sale of the Units hereunder, (3) treating the Closing Date or the Effective Date, as the case may be, as the end of a Tax year; and (4) the Tax shall be determined in accordance with past practice and Precision GAAP, except as otherwise required by Law,) (the "Closing Period Tax"), divided by (ii) the percentage equal to the excess of (a) 100% minus (b) the sum of the maximum federal and Wisconsin capital gain tax rate for long-term capital gain applicable to an individual resident of Wisconsin applicable to receipt of the Gross-Up Payment, net of applicable federal deduction for the Wisconsin capital gain tax. On or before January 31, 2010, Sellers shall calculate or cause to be calculated the amount of the Closing Period Tax and furnish a copy of such calculation (the "Closing Period Tax Calculation") and such other workpapers as MasTec shall reasonably request to MasTec for review. In the event that the parties do not agree on the computation of the Gross-Up Payment or the Closing Period Tax, the dispute shall be resolved in the same manner as provided in <u>Section 1.8.3</u>. The Sellers agree that GPP may elect to "close the books" as of the Closing Date for purposes of Section 706(d) of the Code and, in that event, the items of income, gain, loss and deduction of GPP for the Tax year in which the Closing Date occurs shall be allocated between the Sellers and MasTec in accordance with the close the books method. MasTec agrees to reasonable cooperate with Sellers to cause GPP to adopt the closing of the books allocation method.
- (b) In the event that a Governmental Authority makes a claim that is settled, compromised or otherwise resolved in a final, non-appealable manner that has the effect of increasing the net taxable income of the Companies for the Closing Period, MasTec shall pay to Sellers for the purchase of the Units additional cash consideration computed in accordance with the provisions of this <u>Section 10.3.2</u>, including the interest, but not any penalties with respect to such additional Tax.

- (c) This Section 10.3.2 assumes that Closing occurs in 2009. If the Closing Date occurs after December 31, 2009, then the parties agree to negotiate in good faith to revise this Section 10.3.2 consistent with the parties' intent that the Sellers are to be made whole by MasTec with respect to income Taxes for the period from the Effective Date through the Closing Date, taking into account any Tax benefits from the payment of such Taxes.
- 10.3.3 Subject to the resolution of any Tax contest pursuant to Section 10.3.4, upon notice from Buyer to the Sellers' Representative or Sellers' Representative to Buyer, as the case may be, that an Indemnified Taxpayer is entitled to an indemnification payment for a Tax Loss pursuant to Section 10.3.1 or Section 10.3.2, the indemnifying party thereupon shall pay (or in the case of multiple indemnifying parties, they shall jointly and severally pay) to the Indemnified Taxpayer an amount that, net of any Taxes imposed on the Indemnified Taxpayer with respect to the payment and net of any Tax benefit actually realized by the Indemnified Taxpayer resulting from the Tax Loss, will indemnify and hold the Indemnified Taxpayer harmless from the Tax Loss. Any indemnity payment under this Section 10.3.3 shall be treated as an adjustment to the Purchase Price for Tax purposes.

10.3.4 Tax Contests.

- (a) If a claim with respect to Taxes (a "<u>Tax Claim</u>") shall be made by any Governmental Authority that, if successful, would result in the indemnification of an Indemnified Taxpayer, the Indemnified Taxpayer shall notify the Sellers' Representative or Buyer, as the case may be, in writing (and, in any event, within thirty (30) days of the receipt of notice of any such Tax Claim), but any failure to give the notice will not waive any rights of the Indemnified Taxpayer, except to the extent the indemnifying party or parties have suffered actual prejudice thereby.
- (b) The indemnifying party or parties shall have the right to defend the Indemnified Taxpayer against the claim with counsel of its or their choice so long as (A) with respect to a claim that does not relate to a Straddle Period, the indemnifying party notifies or indemnifying parties notify the Indemnified Taxpayer in writing within 15 days after the Indemnified Taxpayer has given notice of the claim that the indemnifying party will or indemnifying parties will jointly and severally indemnify the Indemnified Taxpayer from and against the entirety of any Tax Losses the Indemnified Taxpayer may suffer resulting from, arising out of, relating to, in the nature of, or caused by the claim (except to the extent that the claim relates to Taxes that have been previously accrued in Actual Net Working Capital on the Actual Closing Balance Sheet), and (B) the indemnifying party or indemnifying parties conduct the defense of the claim actively and diligently.

With respect to any Tax Claim, the indemnifying party or parties shall have the right to control and defend, with counsel of its or their choice at its or their expense, such Tax Claim (and all proceedings taken in connection therewith) and, without limiting the foregoing, may pursue or forego any and all administrative appeals, proceedings, hearings and conferences with any applicable representatives of a Governmental Authority with respect thereto, and may, either pay the Tax claimed and sue for a refund where applicable Law permits such refund suits or contest the Tax Claim in any permissible manner; provided, however, that the indemnifying party or parties will consult with the Indemnified Taxpayer in the negotiation and settlement of any Tax Claim, and the indemnifying party or parties will not, without the written consent of the Indemnified Taxpayer (which consent shall not be unreasonably delayed, conditioned or withheld), settle or compromise any Tax Claim in any manner if such settlement or compromise would have the effect of increasing the Taxes of the Indemnified Taxpayer. To the extent that a Tax Claim relates to a Straddle Period, the Sellers' Representative and the Buyer will jointly control all proceedings taken in connection with any such Tax Claim, and neither Sellers' Representative nor the Buyer or its Affiliate will settle or compromise a Tax Claim relating solely to Taxes of the Company for a Straddle Period without the other party's or parties' written consent which consent shall not be unreasonably delayed, conditioned or withheld. The Indemnified Taxpayer will cooperate with the indemnifying party or parties in contesting any Tax Claim, which cooperation will include the retention and (upon the indemnifying party's or parties' request) the provision to the indemnifying party or parties of records and information which are reasonably relevant to such Tax Claim, and making employees available on a mutually convenient basis to provide additional information or explanation of any material pr

10.3.5 Notwithstanding anything to the contrary in this Agreement, the indemnification obligations under this <u>Section 10</u> shall survive the Closing until the 90th day following the end of the applicable statutes of limitations.

10.3.6 All transfer, documentary, sales, use, stamp, registration and other similar Taxes and fees (including any penalties and interest) incurred in connection with this Agreement shall be jointly and severally paid by the Sellers when due, and the Sellers will, at their expense, file all necessary Tax Returns and other documentation with respect to all the transfer, documentary, sales, use, stamp, registration and other Taxes and fees, and, if required by applicable Law, Buyer will, and will cause its Affiliates to, join in the execution of any of those Tax Returns and other documentation.

11. MISCELLANEOUS

11.1 <u>Notices</u>. All notices, requests, demands, claims and other communications required or permitted to be delivered, given or otherwise provided under this Agreement must be in writing and must be delivered, given or otherwise provided:

(a) by hand (in which case, it will be effective upon delivery);

- (b) by facsimile (in which case, it will be effective upon receipt of confirmation of good transmission); or
- (c) by overnight delivery by a nationally recognized courier service (in which case, it will be effective on the Business Day after being deposited with such courier service);

in each case, to the address (or facsimile number) listed below:

If to the Sellers, to Sellers'

Representative:

3314 56th Street Eau Claire, WI 54703 Tel No.: (715) 874-4510

Michael Daniel Murphy

and

Steven R. Rooney 3314 56 th Street Eau Claire, WI 54703 Tel No.: (715) 874-4510

with a copy (which shall not constitute Gardere Wynne Sewell LLP

notice) to:

3000 Thanksgiving Tower

1601 Elm Street
Dallas, Texas 75201
Telephone: (214) 999-4703
Facsimile: (214) 999-3703

Attention: Larry Schoenbrun and Lawrence B. Goldstein

If to the Buyer, to: MasTec, Inc.

800 S. Douglas Road, 12th Floor Coral Gables, FL 33134 Telephone (305) 406-1892 Facsimile (305) 406-1900

Attention: Pablo A. Alvarez, Executive Vice President, Mergers and Acquisitions

with a copy (which shall not constitute MasTec, Inc.

notice) to:

800 S. Douglas Road, 12th Floor

Coral Gables, FL 33134 Telephone: (305) 406-1849 Facsimile: (305) 406-1947

Attention: Albert de Cardenas, Executive Vice President and General Counsel

with a copy (which shall not constitute Greenberg Traurig, P.A. notice) to:

1221 Brickell Avenue Miami, FL 33131

Telephone: (305) 579-0724 Facsimile: (305) 961-5724 Attention: David Barkus, Esq.

Each of the parties to this Agreement may specify a different address or facsimile number by giving notice in accordance with this Section 11.1 to each of the other parties hereto.

11.2 Sellers' Representative.

11.2.1 Appointment. "Sellers' Representative" means Murphy and Rooney; provided, that in the event of either Murphy's or Rooney's death, resignation or physical or mental incapacity to act as the Sellers' Representative, "Sellers' Representative" shall mean the survivor. Each Seller hereby irrevocably constitutes and appoints the Sellers' Representative as such Seller's attorney-in-fact and agent to act in such Seller's name, place and stead in connection with all matters arising from and under this Agreement, the Escrow Agreement, each of the Company Agreements and any other agreements, documents or instruments related to the Contemplated Transactions and acknowledges that such appointment is coupled with an interest. Sellers' Representative hereby accepts such appointment and authorization.

11.2.2 Authority. Each Seller agrees to be bound by all notices received or given by, and all agreements and determinations made by, and all documents executed and delivered by the Sellers' Representative under this Agreement; authorizes the Sellers' Representative to assert claims, make demands and commence actions on behalf of the Sellers under this Agreement, dispute or to refrain from disputing any claim made by the Sellers, negotiate and compromise any dispute that may arise under, and exercise or refrain from exercising remedies available to the Sellers under, this Agreement, and to sign any releases or other documents with respect to such dispute or remedy (and to bind the Sellers in so doing), give such instructions and do such other things and refrain from doing such things as the Sellers' Representative shall deem appropriate to carry out the provisions of this Agreement, give any and all consents and notices under this Agreement, and perform all actions, exercise all powers, receive service of process with respect to any Action under this Agreement, the Escrow Agreement, the Company Agreement and any other agreement or instrument in connection with the Contemplated Transactions, agree to, negotiate and authorize payments in connection with the Escrow Fund, the Purchase Price Adjustment, the Earn-Out Payments and any other payment pursuant to the terms of this Agreement, and fulfill all duties otherwise assigned to the Sellers' Representative in this Agreement. Each Seller hereby expressly acknowledges and agrees that the Sellers' Representative has the sole and exclusive authority to act on such Seller's behalf in respect of all matters arising under or in connection with this Agreement after execution of this

Agreement, notwithstanding any dispute or disagreement among them, and that no Seller shall have any authority to act unilaterally or independently of the Sellers' Representative in respect to any such matter. Buyer, the Companies (after the Closing) and the Escrow Agent shall be entitled to rely on any and all actions taken by the Sellers' Representative under this Agreement and the Escrow Agreement without any Liability to, or obligation to inquire of, any of the Sellers. All notices, counter notices or other instruments or designations delivered by any Seller in regard to this Agreement shall not be effective unless, but shall be effective if, signed by the Sellers' Representative, and if not, such document shall have no force or effect whatsoever, and Buyer, the Companies (after the Closing), the Escrow Agent and any other Person may proceed without regard to any such document.

- 11.2.3 <u>Change of Representative</u>. The Sellers' Representative may be changed by the Sellers upon not less than twenty (20) calendar days prior written notice to Buyer; provided, that the Sellers' Representative may not be removed unless Sellers who held a majority of the Units agree to such removal and to the identity of the substituted agent or agents. The Sellers' Representative may resign at any time upon not less than thirty (30) calendar days' prior written notice to Buyer, but in any event, not prior to the appointment of a substitute Sellers' Representative. No bond shall be required of the Sellers' Representative. Notices or communications to or from the Sellers' Representative shall constitute notice to or from the Sellers.
- 11.2.4 A decision, act, consent or instruction of the Sellers' Representative, including an amendment, extension or waiver of this Agreement, shall constitute a decision of the Sellers and shall be final, binding and conclusive upon the Sellers; and Buyer may conclusively and absolutely, rely, without any inquiry, upon any such decision, act, consent or instruction of the Sellers' Representative as being the decision, act, consent or instruction of the Sellers. Buyer is hereby relieved from any Liability to any Person, including any Seller, for any acts done by it in accordance with or reliance on such decision, act, consent or instruction of the Sellers' Representative.
- 11.2.5 All notices or other communications required to be made or delivered by Buyer to the Sellers shall be made to the Sellers' Representative for the benefit of the Sellers, and any notices so made shall discharge in full all notice requirements of Buyer to the Sellers with respect thereto. All notices or other communications required to be made or delivered by the Sellers to Buyer shall be made by the Sellers' Representative for the benefit of the Sellers, and any notices so made shall discharge in full all notice requirements of the Sellers to Buyer with respect thereto.
- 11.3 <u>Publicity</u>. No public announcement or disclosure will be made by any party with respect to the subject matter of this Agreement or the Contemplated Transactions without the prior written consent of the Buyer and the Sellers' Representative; <u>provided</u>, <u>however</u>, that the provisions of this <u>Section 11.3</u> will not prohibit (a) any disclosure required by any applicable Legal Requirements, including any disclosure necessary or desirable to provide proper disclosure under the securities laws or under any rules or regulations of any securities exchange on which the securities of such party may be listed or traded or (b) any disclosure made in connection with the enforcement of any right or remedy relating to, or the performance of any obligation arising under, this Agreement or the Contemplated Transactions.

- 11.4 <u>Succession and Assignment; No Third-Party Beneficiary</u>. Subject to the immediately following sentence, this Agreement will be binding upon and inure to the benefit of the parties hereto and their respective successors and permitted assigns, each of which such successors and permitted assigns will be deemed to be a party hereto for all purposes hereof. No party may assign, delegate or otherwise transfer either this Agreement or any of its rights, interests, or obligations hereunder (other than transfers by operation of law) without the prior written approval of the other parties; provided, however, that the Buyer may (a) assign any or all of its rights and interests hereunder to one or more of its Affiliates and (b) designate one or more of its Affiliates to perform its obligations hereunder; provided, however, that such assignment or designation shall not relieve Buyer of its obligations hereunder. Except as expressly set forth in Section 9 with respect to Indemnified Persons who are not parties to this Agreement, this Agreement is for the sole benefit of the parties and their permitted successors and assignees and nothing herein expressed or implied will give or be construed to give any Person, other than the parties and such successors and assignees, any legal or equitable rights hereunder.
- 11.5 <u>Amendments and Waivers</u>. No amendment or waiver of any provision of this Agreement will be valid and binding unless it is in writing and signed, in the case of an amendment, by the Buyer and the Sellers' Representative, or in the case of a waiver, by the party or parties against whom the waiver is to be effective. No waiver by any party of any breach or violation or, default under or inaccuracy in any representation, warranty or covenant hereunder, whether intentional or not, will be deemed to extend to any prior or subsequent breach, violation, default of, or inaccuracy in, any such representation, warranty or covenant hereunder or affect in any way any rights arising by virtue of any prior or subsequent such occurrence. No delay or omission on the part of any party in exercising any right, power or remedy under this Agreement will operate as a waiver thereof.
- 11.6 <u>Further Assurances</u>. From and after the date hereof, upon the request of the Sellers' Representative or the Buyer, each party will do, execute, acknowledge and deliver all such further acts, assurances, deeds, assignments, transfers, conveyances and other instruments and papers as may be reasonably required or appropriate to carry out the Contemplated Transactions. No Seller will take any action that is designed or intended to have the effect of discouraging any lessor, licensor, supplier, distributor or customer of either Company or other Person with whom such Company has a relationship from maintaining the same relationship with such Company after the Closing as it maintained prior to the Closing.
- 11.7 Entire Agreement. This Agreement, the Company Agreements, the Buyer Confidentiality Agreement and the other agreements and documents to be executed and delivered pursuant hereto or contemporaneously herewith constitute the entire agreement among the parties hereto with respect to the subject matter hereof and supersede any and all prior and contemporaneous discussions, negotiations, proposals, undertakings, understandings and agreements, whether written or oral, with respect thereto.

- 11.8 Schedules; Listed Documents, etc. Neither the listing nor description of any item, matter or document in any Disclosure Schedule hereto nor the furnishing or availability for review of any document will be construed to modify, qualify or disclose an exception to any representation or warranty of any party made herein or in connection herewith, except to the extent that such representation or warranty specifically refers to such Disclosure Schedule or a reasonable person would be reasonably likely to conclude that a matter disclosed on a Disclosure Schedule is also responsive to a matter required to be disclosed on another Disclosure Schedule. The inclusion of any information in any Disclosure Schedule shall not be deemed to be an admission or evidence of the materiality of such item. In addition, matters reflected in the Disclosure Schedules are not necessarily limited to matters required by this Agreement to be reflected in the Disclosure Schedules. Such additional matters are set forth for informational purposes only and do not necessarily include other matters of a similar nature. The information contained in this Agreement, in the Disclosure Schedules, and exhibits hereto is disclosed solely for purposes of this Agreement, and no information contained herein or therein shall be deemed to be an admission by any party hereto to any Person other than the parties hereto of any matter whatsoever (including any violation of Law or breach of contract).
- 11.9 <u>Counterparts</u>; <u>Execution</u>. This Agreement may be executed in any number of counterparts, each of which will be deemed an original, but all of which together will constitute but one and the same instrument. This Agreement will become effective when duly executed by each party hereto. Facsimile or other electronically scanned and transmitted signatures shall be deemed originals and shall constitute valid execution and acceptance of this Agreement by the signing/transmitting party.
- 11.10 <u>Survival</u>. The covenants and agreements and, subject to <u>Section 9.3</u>, the representations and warranties set forth in this Agreement shall survive and remain in effect after the Closing.
- 11.11 <u>Severability</u>. Any term or provision of this Agreement that is invalid or unenforceable in any situation in any jurisdiction will not affect the validity or enforceability of the remaining terms and provisions hereof or the validity or enforceability of the offending term or provision in any other situation or in any other jurisdiction. If any provision hereof would, under applicable Law, be invalid or unenforceable in any respect, then each party hereto intends that such provision will be construed by modifying or limiting it so as to be valid and enforceable to the maximum extent compatible with, and possible under, applicable Law.
- 11.12 <u>Headings</u>. The headings contained in this Agreement are for convenience purposes only and will not in any way affect the meaning or interpretation hereof.
- 11.13 <u>Construction</u>. The parties have participated jointly in the negotiation and drafting of this Agreement. In the event an ambiguity or question of intent or interpretation arises, this Agreement will be construed as if drafted jointly by the parties and no presumption or burden of proof will arise favoring or disfavoring any party by virtue of the authorship of any of the provisions of this Agreement. The parties intend that each representation, warranty and covenant contained herein will have independent significance. If any party has breached or violated, or if there is an inaccuracy in, any representation, warranty or covenant contained herein, the fact that there exists another representation, warranty or covenant relating to the same subject matter (regardless of the relative levels of specificity) which the party has not breached or violated, or in respect of which there is not an inaccuracy, will not detract from or mitigate the fact that the party has breached or violated, or there is an inaccuracy in, the first representation, warranty or covenant.

11.14 <u>Governing Law</u>. This Agreement, the rights of the parties and all Actions arising in whole or in part under or in connection herewith, will be governed by and construed in accordance with the laws of the State of Florida, without giving effect to any choice or conflict of law provision or rule that would cause the application of the laws of any other jurisdiction.

11.15 Jurisdiction; Venue; Service of Process.

- 11.15.1 <u>Jurisdiction</u>. Subject to the provisions of <u>Section 5.2.3</u>, each party, by its execution hereof, (a) hereby irrevocably submits to the exclusive jurisdiction of the state courts of the State of Florida or the United States District Courts located in the Southern District of Florida for the purpose of any Action between the parties arising in whole or in part under or in connection with this Agreement, (b) hereby waives to the extent not prohibited by applicable law, and agrees not to assert, by way of motion, as a defense or otherwise, in any such Action, any claim that it is not subject personally to the jurisdiction of the above-named courts, that its property is exempt or immune from attachment or execution, that any such Action brought in one of the above-named courts should be dismissed on grounds of *forum non conveniens*, should be transferred or removed to any court other than one of the above-named courts, or should be stayed by reason of the pendency of some other proceeding in any other court other than one of the above-named courts, or that this Agreement or the subject matter hereof may not be enforced in or by such court and (c) hereby agrees not to commence any such Action other than before one of the above-named courts. Notwithstanding the previous sentence a party may commence any Action in a court other than the above-named courts solely for the purpose of enforcing an order or judgment issued by one of the above-named courts.
- 11.15.2 <u>Venue</u>. Each party agrees that for any Action between the parties arising in whole or in part under or in connection with this Agreement, such party will bring Actions only in Miami-Dade County, Florida. Each party further waives any claim and will not assert that venue should properly lie in any other location within the selected jurisdiction.
- 11.15.3 Service of Process. Each party hereby (a) consents to service of process in any Action between the parties arising in whole or in part under or in connection with this Agreement in any manner permitted by Florida law, (b) agrees that service of process made in accordance with clause (a) or made by registered or certified mail, return receipt requested, at its address specified pursuant to Section 11.1, will constitute good and valid service of process in any such Action (including, with respect to each Seller that notice need only be given to the Sellers' Representative) and (c) waives and agrees not to assert (by way of motion, as a defense, or otherwise) in any such Action any claim that service of process made in accordance with clause (a) or (b) does not constitute good and valid service of process.
- 11.16 <u>Waiver of Jury Trial</u>. TO THE EXTENT NOT PROHIBITED BY APPLICABLE LAW THAT CANNOT BE WAIVED, THE PARTIES HEREBY WAIVE, AND COVENANT THAT THEY WILL NOT ASSERT (WHETHER AS PLAINTIFF, DEFENDANT OR OTHERWISE), ANY RIGHT TO TRIAL BY JURY IN ANY ACTION ARISING IN WHOLE OR IN PART UNDER OR IN CONNECTION WITH THIS AGREEMENT OR ANY OF THE CONTEMPLATED TRANSACTIONS,

WHETHER NOW EXISTING OR HEREAFTER ARISING, AND WHETHER SOUNDING IN CONTRACT, TORT OR OTHERWISE. THE PARTIES AGREE THAT ANY OF THEM MAY FILE A COPY OF THIS PARAGRAPH WITH ANY COURT AS WRITTEN EVIDENCE OF THE KNOWING, VOLUNTARY AND BARGAINED-FOR AGREEMENT AMONG THE PARTIES IRREVOCABLY TO WAIVE THEIR RIGHTS TO TRIAL BY JURY IN ANY ACTION WHATSOEVER BETWEEN OR AMONG THEM RELATING TO THIS AGREEMENT OR ANY OF THE CONTEMPLATED TRANSACTIONS, WHICH ACTION WILL INSTEAD BE TRIED IN A COURT OF COMPETENT JURISDICTION BY A JUDGE SITTING WITHOUT A JURY.

- 11.17 Expenses. Except as provided in Sections 1.8.3, 1.10.6, 4.3.2, 9 or 10, each party shall bear its own expenses incurred in connection with this Agreement and the Contemplated Transactions; provided that Sellers shall bear all Closing Date Seller Expenses, and no Closing Date Seller Expenses will be included in the Actual Closing Balance Sheet.
- 11.18 <u>Specific Performance</u>. Buyer shall have the right and remedy, in addition to any others that may be available, at law or in equity, to have the provisions of this Agreement specifically enforced through injunctive or other relief, without the necessity of posting a bond, it being acknowledged that each Company is a unique asset, with value to Buyer's business not readily quantifiable and any breach by Sellers and/or either Company which causes the Closing not to occur will cause irreparable injury to Buyer, the amount of which will be difficult to determine, and that money damages will not provide an adequate remedy to Buyer. The Sellers and the Companies covenant and agree that they shall not, and shall not authorize any other Person to, challenge or question the enforceability of any provision of this <u>Section 11.18</u>.
 - 11.19 Spousal Consent. Each of Mrs. Murphy and Mrs. Rooney acknowledges and agrees as follows:
 - 11.19.1 THAT SHE IS AWARE THAT SHE HAS THE RIGHT TO SEEK INDEPENDENT LEGAL COUNSEL RELATIVE TO HER RIGHTS AFFECTED BY THIS AGREEMENT PRIOR TO SIGNING THIS AGREEMENT;
 - 11.19.2 that she is aware of the fact that the "Wisconsin Marital Property Law" governs the property rights of married persons in Wisconsin. Under the Wisconsin Marital Property System, each spouse has a 50% ownership interest in property acquired during marriage resulting from the efforts of either or both spouses (specifically excluding gifted or inherited property which shall be the individual property of the receiving spouse);
 - 11.19.3 that she has read and understands this Agreement, having been given reasonable opportunity to inquire relative to the same; and
 - 11.19.4 that she consents to the terms and conditions of this Agreement and to the consummation of the Contemplated Transactions.
- 11.20 <u>Joint and Several</u>. The covenants, agreements, obligations and Liabilities of the Companies, the Sellers, the Sellers' Representative, Mrs. Murphy and Mrs. Rooney hereunder and under each agreement or other document executed in connection herewith, whether on the date hereof, the Effective Date, the Closing Date or at any other time shall be joint and several in all

respects. The covenants, agreements, obligations and Liabilities of MasTec and the Buyer hereunder and under each agreement or other document executed in connection herewith, whether on the date hereof, the Effective Date, the Closing Date or at any other time shall be joint and several in all respects. Where in this Agreement or any agreement or document referred to in the preceding sentence, provision is made for any action to be taken or not to be taken by any of the Companies, the Sellers' Representative Mrs. Murphy or Mrs. Rooney the Sellers agree to cause such Persons to take or not to take such action and shall be jointly and severally Liable for such Persons taking or not taking any such action, as the case may be. Where in this Agreement or any agreement or document referred to in the preceding sentence, provision is made for any action to be taken or not to be taken by Buyer, MasTec agrees to cause Buyer to take or not to take such action and shall be jointly and severally Liable for Buyer taking or not taking any such action, as the case may be.

11.21 Legal Representation. Each of the parties hereto acknowledges and understands that Gardere Wynne Sewell LLP is representing PPL, Steven R. Rooney, and Michael Daniel Murphy and no other party hereto. In any proceeding by or against any Seller wherein any Seller asserts or prosecutes any claim under, or otherwise seeks to enforce, this Agreement, or any other agreement executed or to be executed in connection with the Contemplated Transactions, any claim is asserted against any Seller, or this Agreement or any other agreement executed or to be executed in connection with the Contemplated Transactions is sought to be enforced against any Seller, Buyer and MasTec agree in connection with such proceeding (a) that neither they nor their counsel will move to seek disqualification of Gardere Wynne Sewell LLP, (b) to waive any right they may have to assert the attorney-client privilege against Gardere Wynne Sewell LLP with respect to any communication or information contained in Gardere Wynne Sewell LLP's possession or files and (c) to consent to the representation of the Sellers by Gardere Wynne Sewell LLP. This consent and waiver extends to Gardere Wynne Sewell LLP's representing the Sellers against Buyer and MasTec in litigation, arbitration or mediation in connection with this Agreement or the Contemplated Transactions.

11.22 <u>Disclaimer</u>. EXCEPT AS SET FORTH IN THIS AGREEMENT, SELLERS AND THE COMPANIES DO NOT MAKE ANY REPRESENTATION OR WARRANTY, EXPRESS, STATUTORY, OR IMPLIED WITH RESPECT TO THE INTERESTS, ANY COMPANY, OR ANY OF THE ASSETS, AND EXCEPT AS SET FORTH IN THIS AGREEMENT, SELLERS HEREBY EXPRESSLY DISCLAIM ANY AND ALL OTHER REPRESENTATIONS AND WARRANTIES WITH RESPECT TO THE INTERESTS, ANY COMPANY, OR ANY ASSETS, SUCH DISCLAIMER INCLUDING ANY WARRANTY OR REPRESENTATION (EXPRESS, STATUTORY OR IMPLIED) AS TO: (I) THE CONDITION OF THE ASSETS, (II) FITNESS OF THE ASSETS FOR A PARTICULAR PURPOSE, (III) THE ACCURACY OR COMPLETENESS OF ANY INFORMATION FURNISHED TO BUYER OR ITS REPRESENTATIVES, ADVISORS OR CONSULTANTS BY THE COMPANIES, SELLERS, SELLERS' AFFILIATES OR REPRESENTATIVES, INCLUDING ANY INFORMATION IN ANY "DATA ROOM" OR "VIRTUAL DATA ROOM," ANY PROJECTIONS AND ANY INTERIM FINANCIAL STATEMENTS (OTHER THAN THOSE "INTERIM FINANCIALS" ATTACHED AS <u>SCHEDULE 2.6.1</u>) AND (IV) COMPLIANCE WITH ANY REGULATIONS, LAWS OR PROTECTION OF ENVIRONMENT OR HEALTH. BUYER ACKNOWLEDGES AND AGREES TO THE FOREGOING AND THAT THE FOREGOING DISCLAIMER IS "CONSPICUOUS."

12. DEFINITIONS; CERTAIN RULES OF CONSTRUCTION.

- 12.1 <u>Definitions</u>. As used herein, the following terms will have the following meanings:
- "1933 Act" means the Securities Act of 1933.
- "Accounts Receivable" means the aggregate amount of accounts, commissions and debts, excluding retainage, payable to Precision. For all purposes hereunder, the Accounts Receivable shall be valued at their net realizable value, net of an allowance for bad debts.
- "Action" means any action or suit (whether in contract, tort or otherwise), litigation (whether at law or in equity, whether civil or criminal), controversy, assessment, arbitration, investigation, hearing, charge, complaint, demand, notice or proceeding to, from, by or before any Governmental Authority.
- "Affiliate" means with respect to any specified Person, (a) each Person directly or indirectly controlling, controlled by or under direct or indirect common control with such specified Person at such time, (b) each Person who is at such time an officer, limited liability company manager or director of, or direct or indirect beneficial holder of at least 30% of any class of the capital stock of, such specified Person, (c) each Person that is managed by a common group of executive officers, limited liability company managers and/or directors as such specified Person, (d) the Members of the Immediate Family (i) of each officer, director, limited liability company manager or holder described in clause (b) and (ii) if such specified Person is an individual, of such specified Person, and (e) each Person of which such specified Person or an Affiliate (as defined in clauses (a) through (d)) thereof will, directly or indirectly, beneficially own at least 20% of any class of equity interests at such time.
- "Affiliated Group" means any affiliated group within the meaning of section 1504(a) of the Code or any similar group defined under a similar provision of any Legal Requirement.
- "Backlog" means the amount of revenue that Precision, based on its past practices for calculating such amounts, expects to recognize from: (i) remaining work to be performed on uncompleted Contractual Obligations in progress and (ii) executed Contractual Obligations on which work has not yet begun.
- "Business Day" means any weekday other than a weekday on which banks in New York City, New York or Coral Gables, Florida are authorized or required to be closed.
 - "Businesses" means, collectively, the Precision Business and the Transport Business.
 - "Buyer's Disclosure Schedule" means the disclosure schedules attached hereto and delivered by Buyer to Sellers in connection with this Agreement.

"Buyer's Expenses" means all documented out of pocket expenses incurred by Buyer, MasTec or any of their Affiliates and Related Entities in connection with the investigation, financing, documentation, negotiation and consummation of the Contemplated Transactions and any other document, agreement or instrument contemplated by this Agreement, including, without limitation, the reasonable fees and expenses of counsel, accountants and financial advisors, any fees paid to obtain any consents required to permit MasTec to enter into this Agreement or consummate the Contemplated Transactions and fees and expenses incurred in connection with obtaining and consummating the Financing, all in an aggregate amount not to exceed \$1,000,000.00.

"CAC" means Capital Alliance Corporation.

"CERCLA" means the Comprehensive Environmental Response Compensation and Liability Act of 1980.

"Closing Date Purchase Price" means the Cash Purchase Price minus the Escrow Amount.

"Closing Date Seller Expenses" means all fees, costs and expenses (including any attorneys', accountants', financial advisor's or finder's fees) incurred by or for the benefit of the Companies or Sellers in connection with (a) the due diligence conducted in anticipation of the Contemplated Transactions, (b) the negotiation, preparation and review of this Agreement (including the Sellers' Disclosure Schedules and the Companies' Disclosure Schedules) and all agreements, certificates, opinions and other instruments and documents delivered or to be delivered in connection with the Contemplated Transactions, (c) the preparation and submission of any filing or notice required to be made or given in connection with the Contemplated Transactions and the obtaining of any consent required to be obtained in connection with the Contemplated Transactions, in each case of (a), (b), (c) and (d) to the extent such fees, costs and expenses are owing as of Closing Date

"Code" means the U.S. Internal Revenue Code of 1986, as amended.

"Companies' Disclosure Schedules" means the disclosure schedules attached hereto and delivered by the Companies to the Buyer in connection with this Agreement.

"Companies" and "Company" are defined in the Preamble.

"Company Software" means Owned Software and Licensed Software.

"Company Technology" means any and all Technology used or useful in connection with the Business and any and all Intellectual Property in any and all such Technology.

"Compensation" means, with respect to any Person, all salaries, compensation, remuneration, bonuses or benefits of any kind or character whatever, paid or provided directly or indirectly by any Company to such Person or Affiliates of such Person.

"Contaminant" means any pollutant, hazardous substance, radioactive substance, toxic substance, hazardous waste, medical waste, radioactive waste, special waste, solid waste, petroleum or petroleum-derived substance or waste, asbestos, air contaminant, polychlorinated biphenyls, or any hazardous or toxic constituent thereof and includes any substance defined in or regulated under any Environmental Law.

"Contemplated Transactions" means the transactions contemplated by this Agreement, including the sale and purchase of the Units.

"Contractual Obligation" means, with respect to any Person, any contract, agreement, deed, mortgage, lease, license, commitment, promise, undertaking, arrangement, performance bond, warranty obligation or understanding, whether written or oral and whether express or implied, or other document or instrument (including any document or instrument evidencing or otherwise relating to any Debt), to which or by which such Person is a party or otherwise subject or bound or to which or by which any property, business, operation or right of such Person is subject or bound.

"CPA Firm" means, unless otherwise agreed in writing by Sellers' Representative and Buyer, an accountant mutually satisfactory to Sellers' Representative and Buyer who satisfies each of the following requirements (unless otherwise agreed by Sellers' Representative and Buyer): (i) neither the accountant nor the firm that employs the accountant shall have performed any accounting or consulting services for any party or any Affiliate of any party at any time during the three year period prior to the date of this Agreement; (ii) the accountant is not related in any way by blood or marriage to any party or any executive officer or director of any party or any Affiliate of such party; (iii) the accountant has been a certified public accountant duly licensed to practice in the state where he or she has his or her primary office for a period of not less than ten years; and (iv) the accountant is willing to accept engagement as a CPA Firm on the terms and conditions of this Agreement.

"Debt" means, with respect to any Person, all obligations (including all obligations in respect of principal, accrued interest, penalties, fees and premiums) of such Person, whether direct or indirect, (a) for borrowed money (including overdraft facilities), (b) for Liabilities secured by any Encumbrance (other than Permitted Encumbrances) existing on property owned or acquired and subject thereto, (c) evidenced by notes, bonds, debentures or similar Contractual Obligations, (d) for the deferred purchase price of property, goods or services, including in connection with the acquisition of any business or non-competition agreement (other than trade payables or accruals incurred in the Ordinary Course of Business), (e) under capital leases (in accordance with GAAP), (f) in respect of letters of credit and bankers' acceptances, (g) for Contractual Obligations relating to interest rate protection, swap agreements, factoring, hedging and collar agreements, (h) in the nature of premiums (prepayment or otherwise) or penalties in connection with the obligations described in clauses (a) through (g) above, and (h) in the nature of Guarantees of the obligations described in clauses (a) through (g) above of any other Person.

"<u>Disclosed Liabilities</u>" means those Liabilities, matters, benefit plans, Actions, disputes, violations, citations, penalties, claims, and potential or Threatened Liabilities, Actions, disputes, violations, citations, penalties, claims (i) disclosed on <u>Schedule 1.2</u> with respect to any and all amounts owed to CAC, (ii) with respect to item 2 on <u>Schedule 2.6.1</u>, (iii) disclosed on <u>Schedule 2.7</u>, but with respect to item 2, only that portion of such item related to the period prior to the Effective Date, (iv) with respect to subsections (h),

(n), (o), (t) and (u) on Schedule 2.8, (v) those Liens (including all underlying Debt related thereto) set forth in Section C of the Encumbrances Schedule attached to Schedule 2.9.1, (vi) with respect to infringement and other claims with respect to item 13 on Schedule 2.14, (vii) with respect to all disclosures on Schedule 2.15, (viii) with respect to item 8 and 9 on Schedule 2.15.3, (ix) with respect to all payments under those agreements listed in item 3 and all liability with respect to item 6 on Schedule 2.17(a), (x) with respect to all items on Schedule 2.17(b), (xii) with respect to all items on Schedule 2.17(b), (xiii) with respect to all items on Schedule 2.17(b), (xiii) with respect to all items on Schedule 2.17(b), (xiii) with respect to all items on Schedule 2.17(b), (xiii) with respect to item 3, only those indemnification obligations for the period prior to the Effective Date and with respect to item 4, only those Environmental, Health and Safety Liabilities arising out of or related to the period prior to the Effective Date, (xvi) with respect to item 3.8. (with respect to breaches of the confidentiality provisions in disclosing such agreements hereunder), and q.2. on Schedule 2.19, (xviii) with respect to all items on Schedule 2.19.3, (xviii) with respect to all items on Schedule 2.22, but with respect to item 1 (including subitems (1) and (2) thereunder) only those unperformed warranty obligations with respect to the period prior to the Effective Date, (xix) with respect to all items disclosed on Schedule 2.24.3, (xx) with respect to all items disclosed on Schedule 2.25, (xxi) with respect to item 1 (with respect to breaches of the confidentiality provisions in disclosing the terms of such agreements) disclosed on Schedule 2.31, (xxii) with respect to all items disclosed on Schedule 10.1.2, and(xxiv) with respect to all items disclosed on Schedule 10.1.3 or otherwise disclosed in an update to a Schedule pursuant to Section 4.6. The matters disclosed in item (f)(2) of Schedule 2.8 and referred t

"Disclosure Schedules" means, collectively, the Companies' Disclosure Schedules, the Sellers' Disclosure Schedule and the Buyer's Disclosure Schedule.

"DOJ" means the United States Department of Justice.

"Effective Date" means November 1, 2009.

"Enbridge" means Enbridge Energy, Limited Partnership, Enbridge Pipelines (Southern Lights) L.L.C. and/or their respective Affiliates.

"Encumbrance" means any charge, claim, community or other marital property interest, condition, equitable interest, lien, license, option, pledge, security interest, mortgage, right of way, easement, encroachment, servitude, right of first offer or first refusal, buy/sell agreement and any other restriction or covenant with respect to, or condition governing the use, construction, voting (in the case of any security or equity interest), transfer, receipt of income or exercise of any other attribute of ownership.

"<u>Enforceability Exceptions</u>" means (a) bankruptcy, insolvency, reorganization, moratorium or other similar Laws now or hereafter in effect affecting the enforceability of creditors' rights generally, and (b) general principles of equity that may limit the availability of remedies (regardless of whether enforceability is considered in a proceeding in equity or at law).

"<u>Enforceable</u>" means, with respect to any Contractual Obligation stated to be "Enforceable" by or against any Person, that such Contractual Obligation is a legal, valid and binding obligation of such Person enforceable by or against such Person in accordance with its terms.

"Environmental, Health and Safety Liabilities" means any claim, demand, order, suit, cost, damages, expense, Liability, obligation (including any investigatory, corrective or remedial obligation), or other responsibility arising from or under Environmental Law or Occupational Safety and Health Law and consisting of or relating to: (a) any environmental, health, or safety matters or conditions (including on-site or off-site contamination, occupational safety and health, and regulation of chemical substances or products); (b) fines, penalties, judgments, awards, settlements, legal or administrative proceedings, damages, losses, claims, demands and response, investigative, remedial, or inspection costs and expenses arising under Environmental Law or Occupational Safety and Health Law for cleanup costs or corrective action, including any investigation, cleanup, removal, containment, or other remediation or response actions ("Cleanup") required by applicable Environmental Law or Occupational Safety and Health Law (whether or not such Cleanup has been required or requested by any Governmental Body or any other Person) and for any natural resource damages; or (d) any other compliance, corrective, investigative, or remedial measures required under Environmental Law or Occupational Safety and Health Law. The terms "removal," "remedial," and "response action," include the types of activities covered by CERCLA.

"Environmental Laws" means all Legal Requirements relating to or addressing the environment, health or safety, which shall include the use, handling, treatment, storage or disposal of any Contaminant, or workplace or worker safety and health.

"Equity Security" of any Person means any (i) capital stock, membership or partnership interest or other ownership interest of or in such Person, (ii) securities directly or indirectly convertible into or exchangeable for any for the foregoing; (iii) options, warrants or other rights directly or indirectly to purchase or subscribe for any of the foregoing or securities convertible into or exchangeable for any of the foregoing; or (iv) contracts, commitments, agreements, understandings, arrangements, calls or claims of any kind relating to the issuance of any of the foregoing or giving any Person the right to participate in or receive any payment based on the profits or performance of such Person (including any equity appreciation, phantom equity or similar plan or right).

"ERISA" means the federal Employee Retirement Income Security Act of 1974.

"Escrow Amount" means Fifteen Million Dollars (\$15,000,000).

"Exchange Act" means the Securities Exchange Act of 1934.

"Excluded Damages" means any incidental, consequential (including lost profits), punitive, special, indirect or exemplary damages, damages for diminution in value, or damages calculated on a multiple of earnings or similar basis.

- "Facilities" means any buildings, plants, improvements or structures located on the Real Property.
- "Force Majeure" means acts of God, strikes, equipment or transmission failure or damage reasonably beyond its control, or other cause reasonably beyond its control.
 - "FTC" means the United States Federal Trade Commission.
 - "GAAP" means generally accepted accounting principles in the United States as in effect from time to time.
- "Governmental Authority." means any United States federal, state or local or any foreign government, or political subdivision thereof, or any multinational organization or authority or any authority, agency or commission entitled to exercise any administrative, executive, judicial, legislative, police, regulatory or taxing authority or power, any court or tribunal (or any department, bureau or division thereof), or any arbitrator or arbitral body.
- "Governmental Authorization" means any approval, consent, license, Permit, waiver, or other authorization issued, granted, given, or otherwise made available by or under the authority of any Governmental Authority or pursuant to any Legal Requirement.
- "Governmental Order" means any order, writ, judgment, injunction, decree, stipulation, ruling, determination or award entered by or with any Governmental Authority in a judicial or administrative proceeding.
- "Guarantee" means, with respect to any Person, (a) any guarantee of the payment or performance of, or any contingent obligation in respect of, any Debt or other Liability of any other Person; (b) any other arrangement whereby credit is extended to any obligor (other than such Person) on the basis of any promise or undertaking of such Person (i) to pay the Debt or other Liability of such obligor, (ii) to purchase any obligation owed by such obligor, (iii) to purchase or lease assets under circumstances that are designed to enable such obligor to discharge one or more of its obligations or (iv) to maintain the capital, working capital, solvency or general financial condition of such obligor; and (c) any liability as a general partner of a partnership or as a venturer in a joint venture in respect of Debt or other obligations of such partnership or venture.
 - "Headquarters Facility" means Precision's headquarters and yard located at 3314 56th Street, Eau Claire, Wisconsin 54703.
 - "HSR Act" means the Hart-Scott-Rodino Premerger Notification Act, 15 U.S.C. 18a.
 - "Indemnified Person" means, with respect to any Indemnity Claim, the Person asserting such claim under Section 9.1 or 9.2, as the case may be.
- "Indemnifying Person" means, with respect to any Indemnity Claims, (i) the Sellers, or (ii) the Buyer under Section 9.1 or 9.2, respectively, against whom such claim is asserted.
 - "Indemnity Claim" means a claim for indemnity under Section 9.1 or 9.2.

- "Intellectual Property" means the entire right, title and interest in and to all proprietary rights of every kind and nature anywhere, including all rights and interests pertaining to or deriving from:
 - (a) patents, copyrights, mask work rights, Technology, know-how, processes, Trade Secrets, algorithms, inventions, works, proprietary data, databases, formulae, research and development data and Software;
 - (b) trademarks, trade names, service marks, service names, brands, trade dress and logos, and the goodwill and activities associated therewith;
 - (c) domain names, rights of privacy and publicity, moral rights, and proprietary rights of any kind or nature, however denominated, throughout the world in all media now known or hereafter created;
 - (d) any and all registrations, applications, recordings, licenses, common-law rights and Contractual Obligations relating to any of the foregoing; and
 - (e) all Actions and rights to sue at law or in equity for any past, present or future infringement or other impairment of any of the foregoing, including the right to receive all proceeds and damages therefrom, and all rights to obtain renewals, continuations, divisions or other extensions of legal protections pertaining thereto.
- "Inventory" means all inventory related to the Business, wherever located, including all finished goods whether held at any location or facility of the Company or in transit to the Company.
- "Investment" means (a) any direct or indirect ownership, purchase or other acquisition by a Person of any notes, obligations, instruments, Equity Securities (including joint venture interests) of any other Person; and (b) any capital contribution or similar obligation by a Person to any other Person.
- "Key Employee" means Kara Linderholm, Bobby Poteet, Jake Breunig, Jim Cunningham, Russell Fischer, Scott Schoener, Jim Mallon and all other superintendants of the Businesses.
- "<u>Law</u>" means any United States federal, national, foreign, supranational, state, provincial, local or similar statute, law, standard, resolution, promulgation, ordinance, regulation, rule, code, order, requirement or rule of law (including common law), or any similar provision having the force or effect of law.
 - "Legal Requirement" means any Law, Governmental Order or Permit.
- "<u>Liability</u>" means, with respect to any Person, any liability or obligation of such Person, of any kind, character or description, whether known or unknown, whether asserted or unasserted, whether executory, determined, determinable or otherwise, whether absolute or contingent, whether accrued or unaccrued, whether liquidated or unliquidated, whether disputed or undisputed, whether

disclosed or undisclosed, whether incurred or consequential, whether secured or unsecured, joint or several, vested or unvested, whether due or to become due, whether choate or inchoate and whether or not required under GAAP to be accrued on the financial statements of such Person and regardless of whether such debt, duty or liability is immediately due and payable, and including all costs and expenses related thereto.

"<u>Licensed Software</u>" means all Software that is owned by any third party and that is licensed to and used by either Company in the conduct of its respective Business.

"Losses" means any loss, liability, claim, damage, expense (including costs of defense and reasonable attorneys' fees under circumstances a party is entitled to incur such costs and fees pursuant to this Agreement); provided, however, the term Losses shall be limited to actual damages and shall exclude Excluded Damages, except (i) with respect to any breach of the representations and warranties set forth in Section 2.6 (Financial Statements) as to which Losses shall include damages for a diminution of value, indirect damages and other special damages, and (ii) in the case of fraud or intentional misrepresentation in which event Losses shall include all actual damages and all Excluded Damages.

"MasTec" means MasTec, Inc., a Florida corporation, of which the Buyer is a wholly-owned subsidiary.

"MasTec Material Adverse Effect" means any change in, development, event, occurrence or effect on, MasTec's businesses, operations, assets, currently contemplated contracts, condition (financial or otherwise) or results of operations which, when considered either individually or in the aggregate together with all other adverse changes or effects with respect to which such phrase is used in this Agreement, (a) is, or would reasonably be expected to be, materially adverse to MasTec's businesses, operations, assets, currently contemplated contracts, condition (financial or otherwise) or results of operations, or (b) prevents or materially delays MasTec's ability to perform its obligations hereunder, but shall exclude any prospects and shall also exclude any effect resulting or arising from: (i) any action taken by any of the Sellers or any of their Affiliates; (ii) any omission to act or action taken with the prior written consent of the Sellers' Representative (including those omissions to act or actions taken which are required by this Agreement); or (iii) any national or international political or social occurrence, including the engagement by the United States in hostilities, whether or not pursuant to the declaration of a national emergency or war, or the occurrence of any military or terrorist attack upon the United States, or any of its territories, possessions, or diplomatic or consular offices or upon any military installation, equipment or personnel of the United States, except to the extent any such event or occurrence impacts MasTec in a manner disproportionate to the impact on other companies in the industries in which it operates.

"Material Adverse Effect" means any change in, development, event, occurrence or effect on, the Businesses, operations, Assets, currently contemplated contracts (including (a) with El Paso Energy for both current and planned projects, including the Ruby Ridge Pipeline project (the "Ruby Project") and (b) with Enbridge), condition (financial or otherwise) or results of operations of the Companies which, when considered either individually or in the aggregate together with all other adverse changes or effects with respect to which such phrase is used in this Agreement, (a) is, or would reasonably be expected to be, materially adverse to the

Businesses, operations, Assets, currently contemplated contracts (including (a) with El Paso Energy for both current and planned projects, including the Ruby Project and (b) Enbridge), condition (financial or otherwise) or results of operations of the Companies, or (b) prevents or materially delays a Company's ability to perform its obligations hereunder, but shall exclude any prospects (other than with respect to Ruby and Enbridge projects) and shall also exclude any effect resulting or arising from: (i) any action taken by Buyer or any of its Affiliates; (ii) any omission to act or action taken with the prior written consent of the Buyer (including those omissions to act or actions taken which are required by this Agreement); or (iii) any national or international political or social occurrence, including the engagement by the United States in hostilities, whether or not pursuant to the declaration of a national emergency or war, or the occurrence of any military or terrorist attack upon the United States, or any of its territories, possessions, or diplomatic or consular offices or upon any military installation, equipment or personnel of the United States, except to the extent any such event or occurrence impacts the Companies in a manner disproportionate to the impact on other companies in the industry in which the Companies operate. The parties hereto acknowledge and agree that any loss or material adverse change to either one or both of the Ruby and Enbridge contracts shall be conclusively and for all purposes deemed to be a Material Adverse Effect, and each party waives any right he or it has to contest that such loss is not a Material Adverse Effect.

"Materiality Qualifier" means any qualifier referencing materiality, a Material Adverse Effect or any similar qualifier.

"Members of the Immediate Family" means, with respect to any individual, (a) such Person's spouse, (b) each parent, brother, sister or child of such Person or such Person's spouse, (c) the spouse of any Person described in clause (b) above, (d) each child of any Person described in clauses (a), (b) or (c) above, (e) each trust created solely for the benefit of one or more of the Persons described in clauses (a) through (d) above and (f) each custodian or guardian of any property of one or more of the Persons described in clauses (a) through (e) above in his capacity as such custodian or guardian.

"Net Working Capital" means as of any particular date (a) the value of all of the Companies' current Assets (including for any determination as of the Effective Date, all cash on the books of PPL as of the Effective Date), excluding (i) any Receivables not collected on or prior to the 90th day following the Closing Date, (ii) any inter-Company Receivables and (iii) the GPP Investment Asset, less (b) the amount of all of the Companies' current Liabilities (x) excluding any inter-Company Liabilities and Liabilities included in Actual Indebtedness (for purposes of this definition only, Actual Indebtedness shall be determined prior to the exclusion of the Indebtedness reduction described in the proviso of Section 1.5.1(c)) and (y) including accrued current Liabilities not yet due (including for any determination as of the Effective Date, accrued expenses of PPL for salary and benefits of its employees as of the Effective Date), all as determined in accordance with Precision GAAP; provided that, and regardless of whether such accruals were included in the Financial Statements, current Liabilities shall (1) include (i) accruals for "stand-by pay" accrued and unpaid to the Effective Date, (ii) office employee bonus accruals for 2009 in an amount equal to \$500,000, (iii) field employee bonus accruals for 2009 in an amount equal to \$2,000,000, (iv) vacation pay accruals for 2009 in an amount equal to \$25,000 and (v) accruals for profit sharing in an amount equal to \$200,000 (the amount set forth in each of the foregoing clauses (ii) through (v) shall be that percentage

of the full amount equal to the percentage of calendar year 2009 elapsed prior to the Effective Date), and (2) exclude any reserve or accruals for the following: (i) Receivables; (ii) retainage; (iii) Disclosed Liabilities; (iv) Tax Losses; (v) termination costs of Zurich American Insurance Company worker's compensation insurance policy described in Section 5.9; or (vi) the matters set forth on Schedule 9.1.5, nor shall any write-offs be made with respect to any of the items in this clause (2).

"Occupational Safety and Health Law" means any Legal Requirement, including the Occupational Safety and Health Act of 1970, as amended, and the rules and regulations promulgated thereunder, which both has been adopted and is effective prior to the Closing Date and which is designed to provide safe and healthful working conditions and to reduce occupational safety and health hazards, and any program, whether governmental or private (including those promulgated or sponsored by industry associations and insurance companies), designed to provide safe and healthful working conditions.

"Ordinary Course of Business" means an action taken by any Person in the ordinary course of such Person's business which is consistent with the past customs and practices of such Person (including past practice with respect to quantity, amount, magnitude and frequency, standard employment and payroll policies and past practice with respect to management of working capital) which is taken in the ordinary course of the normal day-to-day operations of such Person and does not require the consent of the shareholders, board of directors or members of such Person.

"Organizational Documents" means, with respect to any Person (other than an individual), (a) the certificate or articles of incorporation, formation or organization, and any joint venture, limited liability company, operating or partnership agreement and other similar documents entered into or adopted at any time or filed in connection with the creation, formation or organization of such Person and (b) all by-laws, shareholders' agreements, voting agreements, rights of first refusal and similar documents, instruments or agreements relating to the organization or governance of such Person, in each case, as amended or supplemented.

"Owned Software" means all Software used by each Company in the conduct of its respective Business that is owned or purported to be owned by such Company.

"<u>Permits</u>" means, with respect to any Person, any license, franchise, permit, consent, approval, right, privilege, certificate or other similar authorization issued by, or otherwise granted by, any Governmental Authority or any other Person to which or by which such Person is subject or bound or to which or by which any property, business, operation or right of such Person is subject or bound.

"Permitted Encumbrance" means (a) statutory liens for current Taxes, special assessments or other governmental charges not yet due and payable or the amount or validity of which is being contested in good faith by appropriate proceedings and for which appropriate reserves have been established in books and records in accordance with GAAP, (b) mechanics', materialmen's, carriers', workers', repairers' and similar statutory liens arising or incurred in the Ordinary Course of Business which liens are not material in amount and do not interfered with any current or anticipated operation of either Business, (c) zoning, entitlement, building and

other land use regulations imposed by governmental agencies having jurisdiction over any Real Property which are not violated in any material respect by the current or contemplated use and operation of the Real Property, (d) restrictions on the transfer of securities arising under federal and state securities laws and (e) Encumbrances which will be terminated as of the Closing as provided in this Agreement.

"<u>Person</u>" means any individual or corporation, association, partnership, limited liability company, joint venture, joint stock or other company, business trust, trust, organization, Governmental Authority or other entity of any kind.

"PP&E" means property, plant and equipment, including trucks, trailers and other construction equipment.

"Precision Business" means Precision's pipeline construction, maintenance and related services business for hydrocarbon transmission and distribution (petroleum, natural gas and related products) throughout North America, including all services provided by Precision prior to the date hereof, including (i) cross country construction of pipelines from four (4) to forty-two (42) inches, (ii) pipeline integrity projects, (iii) looping and laterials, (iv) hydrostatic testing, (v) pipeline drying services, (vi) pipeline rehabilitation, (vii) meter and pump station construction and (viii) services and construction related to pumping and compressor facilities, gathering and collection systems and municipal distribution.

"Precision GAAP" means GAAP, applied in a manner consistent with <u>Schedule 12.1</u>, but only to the extent the items on such schedule are consistent with GAAP at the relevant time of calculation.

"Release" means the release, spill, emission, leaking, pumping, injection, deposit, disposal, discharge, dispersal, leaching or migrating into the indoor or outdoor environment of any Contaminant through, in, into or from the air, soil, surface water, groundwater or any property.

"Representative" means, with respect to any Person, any director, officer, employee, agent, consultant, advisor, or other representative of such Person, including legal counsel, accountants, and financial advisors.

"Sellers' Disclosure Schedule" means the disclosure schedules attached hereto and delivered by Sellers to Buyer in connection with this Agreement.

"Sellers' Internal Knowledge" means the knowledge of Murphy, Rooney, Kara Linderholm and Bobby Poteet each of whom will be deemed to have knowledge of all such matters as he or she would have discovered had he or she reviewed the books, records and files, of the Companies and PPL.

"Sellers' Knowledge" means the knowledge of Murphy, Rooney, Kara Linderholm and Bobby Poteet each of whom will be deemed to have knowledge of all such matters as he or she would have discovered had he or she made reasonable inquiries.

"<u>Software</u>" means computer software or firmware in any form, including object code, source code, computer instructions, commands, programs, modules, routines, procedures, rules, libraries, macros, algorithms, tools, and scripts, and all documentation of or for any of the foregoing.

"Subsidiary" means, with respect to any specified Person, any other Person of which such specified Person will, at the time, directly or indirectly through one or more Subsidiaries, (a) own at least 50% of the outstanding capital stock (or other shares of beneficial interest) entitled to vote generally, (b) hold at least 50% of the partnership, limited liability company, joint venture or similar interests or (c) be a general partner, managing member or joint venturer.

"Taxs" means (1) any federal, state, local, or foreign income, gross receipts, license, payroll, employment, excise, severance, stamp, occupation, premium, windfall profits, environmental (including taxes under Section 59A of the Code), custom duties, capital stock, franchise, profits, withholding, social security (or similar), unemployment, disability, real property, personal property, sales, use, transfer, registration, value added, alternative or add-on minimum, estimated, or other tax of any kind whatsoever, including any interest, penalty, or addition thereto, whether disputed or not, and any amounts payable pursuant to the determination or settlement of a Tax audit (provided such amounts are not duplicative of any other Taxes) and (2) liability of the Company for the payment of any amounts of the type described in clause (1) above as a result of any express or implied obligation to indemnify or otherwise assume or succeed to the liability of any other Person.

"<u>Tax Return</u>" means any return, declaration, report, claim for refund or information return or statement relating to Taxes, including any schedule or attachment thereto, and including any amendment thereof.

"<u>Technology</u>" means all inventions, works, discoveries, innovations, know-how, information (including ideas, research and development, formulas, compositions, processes and techniques, data, designs, drawings, specifications, customer and supplier lists, pricing and cost information, business and marketing plans and proposals, documentation and manuals), Software, computer hardware, integrated circuits and integrated circuit masks, electronic, electrical and mechanical equipment and all other forms of technology, including improvements, modifications, works in process, derivatives or changes, whether tangible or intangible, embodied in any form, whether or not protectable or protected by patent, copyright, mask work right, trade secret law or otherwise, and all documents and other materials recording any of the foregoing.

"Threatened" means a claim, Action, dispute or audit, with respect to which a demand or statement has been made to a Company, any Seller or their Representatives orally or in writing.

"<u>Trade Secrets</u>" means trade secrets, know how and confidential business information and any other information, however documented, that is a trade secret within the meaning of the applicable trade secret protection Laws, including the Uniform Trade Secrets Act.

"Treasury Regulations" means the regulations promulgated under the Code.

Term "Ai-i-i Pure1"	Location
"Acquisition Proposal"	Section 4.4.1
"Actual Closing Balance Sheet"	Section 1.8.4
"Actual Indebtedness"	Section 1.8.1
"Actual Net Working Capital"	Section 1.8.1
"Actual PP&E"	Section 1.8.1
"Actual Uncollected Receivables"	Section 1.8.1
"Advisors"	Section 4.1
"Agreement"	Preamble
"Allocated Purchase Price"	Section 1.3
"Allocation"	Section 1.3
"Annual Budget"	Section 1.10.8
"Approved Acquisitions"	<u>Section 1.10.3</u>
"Approved Annual Budget"	Section 1.10.8
"Assets"	Section 2.9.1
"Audited Financials"	Section 2.6.1
"Board"	Section 1.10.8
"Books and Records"	Section 5.5
"Buyer"	<u>Preamble</u>
"Buyer Indemnified Person"	Section 9.1
"Buyer Indemnified Taxpayer"	<u>Section 10.3.1</u>
"Buyer's Earn-Out Position"	<u>Section 1.10.6</u>
"Buyer's Position"	<u>Section 1.8.3</u>
"Cash Purchase Price"	<u>Section 1.5.1</u>
"Closing"	Section 1.9
"Closing Balance Sheet"	<u>Section 1.8.1</u>
"Closing Certificate"	Section 1.4
"Closing Date"	<u>Section 1.9.1</u>
"Closing Period"	<u>Section 10.3.2(a)</u>
"Closing Period Tax"	<u>Section 10.3.2(a)</u>
"Closing Period Tax Calculation"	<u>Section 10.3.2(a)</u>
"Commercial Software"	<u>Section 2.14.1</u>
"Companies" and "Company"	<u>Preamble</u>
"Company Agreements"	Section 2.3
"Company Employees"	<u>Section 2.17(a)</u>
"Competitive Business"	<u>Section 5.2.1(a)</u>
"Compliance Period"	<u>Section 1.10.11</u>
"Controlled Group"	<u>Section 2.17(c)</u>
"Customer Assets"	<u>Section 2.16</u>
"Data Room"	<u>Section 1.9.2(a)(xi)</u>
"Deductible"	<u>Section 9.1.6</u>
"Disclosed Contract"	<u>Section 2.19.2</u>
"Farm Out Diameta Farmana"	C-+: 1 10 C

Section 1.10.6

"Earn-Out Dispute Expenses"

Perm Perm	Location
Earn-Out Objection"	<u>Section 1.10.6</u>
Earn-Out Payments"	<u>Section 1.10.1</u>
Earn-Out Period"	<u>Section 1.10.2</u>
Earn-Out Review Period"	<u>Section 1.10.6</u>
Earn-Out Shares"	Section 2.30.1
EBITDA"	Section 1.10.3
Employee Plans"	<u>Section 2.17(a)</u>
EO Financial Statements"	Section 1.10.3
Equipment"	Section 2.13
Escrow Account"	Section 1.6
Escrow Agent"	Section 1.6
Escrow Agreement"	Section 1.6
Escrow Fund"	Section 1.5.1
Escrow Period"	Section 1.6
Estimated Closing Balance Sheet"	Section 1.4
Estimated Indebtedness"	Section 1.4.1
Estimated Net Working Capital"	Section 1.4.3
Estimated PP&E"	Section 1.4.2
Estimated PP&E Surplus"	<u>Section 1.5.1(c)</u>
Existing Operations"	Section 1.10.3
Final Closing Statement"	Section 1.8.1
Financials"	Section 2.6.1
Financing"	Section 6.9
FLSA"	Section 2.24.2
Gardere"	Section 2.8(i)
General Release"	Section 1.9.2(a)(xiii)
GPP"	Section 10.3.2(a)
GPP Investment Asset"	Schedule 9.1.5
Gross-UP Payment"	Section 10.3.2(a)
Headquarters Lease"	Section 1.9.2(a)(xii)
Hybrid Earn-Out Period"	Section 1.10.11
dentified Excluded Business"	Section 1.10.8
ndemnified Taxpayer"	Section 10.3.1
Interim Financials"	Section 4.9
IRS"	<u>Section 2.17(e)</u>
LandCo"	<u>Section 1.9.2(a)(xii)</u>
Liability Cap"	Section 9.1.6
Liability Policies"	Section 2.26
Licenses"	Section 2.14.6
MasTec Common Stock"	Section 1.10.4
MasTec Financial Statements"	Section 3.7
Most Recent Balance Sheet"	Section 4.9
Most Recent Balance Sheet Date"	Section 2.6.1
Mrs. Murphy"	<u>Preamble</u>

Term Location "Mrs. Rooney" Preamble "Multiemployer Plan" Section 2.17(a) "Murphy" Preamble "Negative EBITDA Earn-Out Period" Section 1.10.1(g) "Noncompliance Period" Section 1.10.11 "PBGC" Section 2.17(e) "PPL" **Preamble** "PPL Assets" Section 2.9.1 "Pre-Closing Period" Section 4.2.1 "Precision" **Preamble** "Precision Units" **Recitals** "Predecessor" Section 2.1.2 "Principals Employment Agreements" Section 1.9.2(a)(xv) "Public Reports" Section 3.6 "Purchase Price" Section 1.2 "Purchase Price Adjustment" Section 1.8.4(a) "Purchase Price Dispute Expenses" Section 1.8.3 "Real Property" Section 2.12.1 Section 2.12.1 "Real Property Leases" "Receivables" Section 2.10 Section 1.9.2(a)(xiv) "Registration Rights Agreement" "Related Entity" Section 5.2.6 "Restrictive Covenant Breach" Section 1.10.11 "Revised Annual Budget" Section 1.10.8 "Rooney" <u>Preamble</u> "SEC" Section 2.15.4(d) "Seller" **Preamble** "Seller Guarantees" Section 4.12 "Seller Indemnified Person" Section 9.2 "Seller-Prepared Returns" Section 10.2.2 "Sellers' Earn-Out Position" Section 1.10.6 "Sellers' Objection" Section 1.8.2 "Sellers' Position" Section 1.8.3 "Sellers' Representative" **Preamble** "Straddle Period" Section 10.2.2 "Straddle Return" Section 10.2.2 "Stub Period" Section 10.3.2(a) "Surety Bonds" Section 2.31.1 "Tax Claim" Section 10.3.4(a) "Tax Losses" Section 10.3.1 "Territory" <u>Section 5.2.1(a)</u> "Third Party Claim" Section 9.4.1 "Transferred Employees" Section 4.10 "Transport" **Preamble**

Term	Location
"Transport Business"	<u>Recitals</u>
"Transport Sellers"	<u>Preamble</u>
"Transport Units"	<u>Recitals</u>
"Uncollected Retainage"	<u>Section 1.12</u>
"Uncollected Retainage Determination Date"	<u>Section 1.12</u>
"Underlying Leases"	<u>Section 2.12.1</u>
"Units"	<u>Recitals</u>
"WARN"	<u>Section 2.24.2</u>

12.3 <u>Rules of Construction</u>. Except as otherwise explicitly specified to the contrary, (a) each reference to a Section, Exhibit or Schedule means a Section of, or Schedule or Exhibit to this Agreement, unless another agreement is specified, (b) the word "including" will be construed as "including without limitation," (c) references to a particular statute or regulation include all rules and regulations thereunder and any predecessor or successor statute, rules or regulation, in each case as amended or otherwise modified from time to time, (d) words in the singular or plural form include the plural and singular form, respectively, (e) references to a particular Person include such Person's successors and assigns to the extent not prohibited by this Agreement and (f) all pronouns and any variations thereof refer to the masculine, feminine or neuter singular or plural as the identity of the Person or Persons may require. The terms "hereof", "herein", "hereunder", "hereto" and "herewith" and words of similar import shall, unless otherwise stated, be construed to refer to this Agreement and not to any particular provision of this Agreement. The word "or" shall not be exclusive. All references herein to "dollars" or "\$" are to United States dollars. Any accounting term used in this Agreement shall have, unless otherwise specifically provided herein, the meaning customarily given such term in accordance with GAAP and all financial computations hereunder will be computed, unless otherwise specifically provided herein, in accordance with GAAP consistently applied. All references herein to any period of days shall mean the relevant number of calendar days unless otherwise specified. Whenever any action must be taken hereunder on or by a day that is not a Business Day, then such action may be validly taken on or by the next day that is a Business Day. All references herein to a "party" or "parties" are to a party or parties to this Agreement unless otherwise specified. The phrases "date of this Agreement," "date hereof"

[signature pages follow]

IN WITNESS WHEREOF, each of the undersigned has executed and delivered this Agreement, or caused this Agreement to be duly executed on its behalf by its officer thereunto duly authorized, as of the date first set forth above.

BUYER:

PRECISION ACQUISITION, LLC

By: /s/ Jose R. Mas
Name: Jose R. Mas
Title: President
GUARANTOR:

MASTEC, INC.

By: /s/ Jose R. Mas

Name: Jose R. Mas

Title: President and Chief Executive Officer

SELLERS:

PPL MANAGEMENT, INC.

By: /s/ Michael Daniel Murphy

Name: Michael D. Murphy

Title: President

/s/ Michael Daniel Murphy

Michael Daniel Murphy

/s/ Steven R. Rooney

Steven R. Rooney

COMPANIES:

PRECISION PIPELINE LLC

By: /s/ Michael Daniel Murphy

Name: Michael D. Murphy Title: Managing Member

Signature page to Interest Purchase Agreement

PRECISION TRANSPORT, LLC

By: /s/ Michael Daniel Murphy
Name: Michael Daniel Murphy

Name: Michael Daniel Mur Title: Managing Member

/s/ Angela D. Murphy

Angela D. Murphy, solely with respect to Sections 11.19 and

11.20

/s/ Karen K. Rooney

Karen K. Rooney, solely with respect to Sections 11.19 and

<u>11.20</u>

SELLERS' REPRESENTATIVE:

/s/ Michael Daniel Murphy

Michael Daniel Murphy, as Sellers' Representative

/s/ Steven R. Rooney

Steven R. Rooney, as Sellers' Representative

Signature page to Interest Purchase Agreement



CPAs, Consultants & Advisors www.larsonallen.com

CONSENT OF LARSONALLEN LLP

We hereby consent to the use of our report dated March 30, 2009, except for Note 15, as to which the date is September 5, 2009, on the consolidated financial statements of PPL Management, Inc. and Subsidiaries for the years ended December 31, 2008, 2007 and 2006 appearing in the Current Report on Form 8-K dated November 3, 2009 and to the incorporation by reference of such information into the following Registration Statements of MasTec Inc. (Form S-8 Nos. 333-139996, 333-112010, 333-105781, 333-105516, 333-38932, 333-77823, 333-47003, 333-38940 and 333-30647 and Form S-3 Nos. 333-158502, 333-142083, 333-133252, 333-46067).

Larson Allen LLP

Eau Claire, Wisconsin November 3, 2009



LarsonAllen LLP is a member of Nexia International, a worldwide network of independent accounting and consulting firms.



Contact:

J. Marc Lewis, Vice President-Investor Relations 305-406-1815 305-406-1886 fax marc.lewis@mastec.com 800 S. Douglas Road, 12th Floor Coral Gables, Florida 33134 Tel: 305-599-1800

Fax: 305-406-1960 www.mastec.com

For Immediate Release

MasTec to Acquire Precision Pipeline, LLC

Coral Gables, FL (November 3, 2009) — MasTec, Inc. (NYSE: MTZ) today announced that it has signed a definitive agreement to acquire Precision Pipeline, LLC, a leading natural gas, crude oil and refined products transmission pipeline infrastructure services provider in North America for a purchase consideration of \$150 million, subject to certain purchase price adjustments and an earnout. The transaction is contingent on financing availability on terms and conditions acceptable to MasTec, in its sole discretion, and there is no break-up fee if such financing is not available.

Based in Eau Claire, Wisconsin, Precision is a leading energy infrastructure services provider, specializing in the construction and maintenance of large diameter pipelines. Precision's experience in the long-haul, interstate pipeline industry will complement MasTec's existing energy infrastructure service offerings, which include natural gas gathering systems, processing plants and compression stations and mid-stream pipelines. Precision employs a team of highly-skilled unionized workers and tradesmen which it deploys throughout North America and utilizes a significant pool of specialized pipeline construction equipment. With the acquisition of Precision, MasTec will become one of the leading pipeline contractors in the country, capable of providing a full array of construction services to oil and gas producers, as well as mid-stream and interstate pipeline operators.

Precision has experienced significant growth over the past several years. Precision generated \$303 million of revenue and \$37 million in EBITDA in 2007 and \$507 million of revenue and \$93 million in EBITDA in 2008. For the full year of 2009, Precision estimates revenue of slightly less than \$300 million and EBITDA of about \$60 million. The reduction in earnings in 2009 as compared to 2008 is primarily due to lower revenue as the result of a softer natural gas and petroleum pipeline construction market amidst a global economic downturn. Precision had over \$500 million in backlog as of September 30, 2009. The acquisition is expected to be accretive for MasTec in 2010 by at least \$0.08 earnings per share, before amortization of acquisition- related intangibles.

Jose Mas, MasTec's President and CEO noted, "We are very pleased with the acquisition of Precision. The Company has an excellent and motivated management team that will remain in place to continue its growth. With the acquisition of Precision, we will significantly expand our capabilities in the natural gas, crude oil and refined petroleum product pipeline industries, which we believe will be solid areas of growth for years to come. Over the last few years, we have invested heavily in positioning MasTec in industries that we believe will have significant growth opportunities. This acquisition complements our recent growth in renewables, electric transmission line construction and wireless infrastructure services."



Mr. Mas concluded, "MasTec continues to generate significant cash flow from its existing operations and management is often asked about our expected uses of the cash flow. We believe the acquisition of a well managed company with a proven track record, strong management team, and solid backlog that exposes MasTec to new geographies, customers and service offerings at an attractive multiple is one of the best ways to deploy capital and increase shareholder value."

MasTec believes that U.S. energy policy goals will continue to favor clean, domestic sources of energy and the Company expects to be a leading player in that effort in both renewable energy and natural gas pipeline construction. With recent developments in drilling and completion technologies for oil and gas, particularly the new shale gas fields, MasTec expects new production fields to be developed and old fields to be expanded significantly. MasTec expects that the resulting incremental production will provide continuing construction opportunities as oil and gas producers and pipeline operators move this oil, gas and refined products to markets via pipelines. With oil prices currently close to \$80 per barrel and natural gas futures prices improving, MasTec believes that this market will continue to grow with excellent margin opportunities. As evidence of this expected growth, as of October 15, 2009, the Federal Energy Regulatory Commission had over 5,000 miles of pending major pipeline projects on file.

Bob Campbell, MasTec's Executive Vice President of Finance and CFO, noted, "Our purchase agreement includes a requirement to obtain at least \$75 million in financing and we are currently reviewing our financing options. Our current capital structure, liquidity and cash flows are all in excellent shape today which enables us to do this very attractive acquisition. MasTec's liquidity as of September 30, 2009 was \$183 million, with liquidity defined as cash plus availability on its senior credit facility. Cash flow from operations for September year-to-date 2009 was \$86 million which was double the cash flow for the same period last year."

In addition to the availability of financing, the transaction is subject to MasTec obtaining all necessary consents from MasTec's lenders under its credit facility, Hart-Scott-Rodino Act approval and normal closing conditions and timelines and MasTec expects to close the transaction sometime before the end of the fourth quarter. Definitive details of the transaction, along with historical financial information, are included in the 8-K which is being filed with the SEC.

Management will also hold a conference call to discuss this transaction on Wednesday, November 4, 2009 at 8:30 a.m. Eastern time. Accompanying slides will be posted on the investor relations section of the Company's website at www.mastec.com. The dial-in number for the conference call is (719) 325-4894 and the replay number is (719) 457-0820, with a pass code of 1432517. The replay will run for 30 days. Additionally, the call will be broadcast live over the Internet and can be accessed and replayed through the investor relations section of the Company's website.



Reconciliation of Non-GAAP Disclosures-Unaudited

(In millions)

	009 mated)	2008	2007
Net income	\$ 44	\$83.3	\$32.2
Depreciation & amortization	10	8.7	4.3
Interest expense, net	2	8.0	0.7
Taxes	4	0.0	0.0
EBITDA	\$ 60	\$92.9	\$37.1

Tables may contain slight summation differences due to rounding.

This press release contains forward-looking statements within the meaning of the Private Securities Litigation Reform Act. These statements are based on our current expectations and are subject to risks, uncertainties, and other factors, some of which are beyond our control, that are difficult to predict, and could cause actual results to differ materially from those expressed or forecasted in the forward-looking statements. Important factors that could cause actual results to differ materially from those in forward-looking statements include; our ability to obtain Hart-Scott Rodino Act approval for our Precision acquisition, or termination of the applicable waiting period; our ability to consummate the Precision acquisition on a timely basis or at all; our ability to retain qualified personnel and key management, integrate Precision with MasTec within the expected timeframes and achieve the revenue, cost savings and earnings levels from the acquisition at or above the levels projected; our ability to maintain and grow the customer relationship with Precision's two principal customers and/or replace such contracts or otherwise obtain new business; delays associated with any of Precision's projects; the demand for oil and natural gas; the timing and extent of fluctuations in geographic, weather, equipment and operational factors affecting the oil and gas industry; the impact of any Precision liabilities that are unknown to us; our dependence on a limited number of customers; the ability of our customers, including our largest customers, to terminate or reduce the amount of work, or in some cases prices paid for services on short or no notice under our contracts; the impact of Precision's unionized workforce on our operations, including labor availability and relations; liabilities associated with Precision's multiemployer union pension plans, including underfunding liabilities; further or continued economic downturns, reduced capital expenditures, reduced financing availability, customer consolidation and technological and regulatory changes in the industries we serve; market conditions, technical and regulatory changes that affect our customers' industries; our ability to retain qualified personnel and key management from acquired businesses, enforce any noncompetition agreements, integrate acquired businesses within expected timeframes and achieve the revenue, cost savings and earnings levels from such acquisitions at or above the levels projected; the impact of the American Recovery and Reinvestment Act of 2009 and any similar local or state regulations affecting renewable energy, transmission, broadband and related projects and expenditures; the effect of state and federal regulatory initiatives, including costs of compliance with existing and future environmental requirements; our ability to attract and retain qualified managers and skilled employees; increases in fuel, maintenance, materials, labor and other costs; any liquidity issues related to our securities held for sale; any adverse determination of any claim, lawsuit or proceeding; the highly competitive nature of our industry; the adequacy of our insurance, legal and other reserves and allowances for doubtful accounts; any exposure related to our divested state Department of Transportation projects and assets; restrictions imposed by our credit facility, senior notes and any future loans or securities; any dilution or stock price volatility which shareholders may experience in connection with shares we may issue as consideration for earn-out obligations in connection with past or future acquisitions, including in connection with our acquisition of Precision, or conversions of convertible notes or other stock issuances; the outcome of our plans for future operations, growth, and services, including backlog and acquisitions; and the other factors referenced in the reports we furnish to and file with the SEC. We do not undertake any obligation to update forward-looking statements.

PPL MANAGEMENT, INC. AND SUBSIDIARIES CONSOLIDATED FINANCIAL STATEMENTS

YEARS ENDED DECEMBER 31, 2008 AND 2007 AND 2006

PPL MANAGEMENT, INC. AND SUBSIDIARIES TABLE OF CONTENTS YEARS ENDED DECEMBER 31, 2008 AND 2007 AND 2006

INDI	EPENDENT AUDITORS' REPORT	1
FINA	ANCIAL STATEMENTS	
	CONSOLIDATED BALANCE SHEETS	2
	CONSOLIDATED STATEMENTS OF OPERATIONS	4
	CONSOLIDATED STATEMENTS OF EQUITY	5
	CONSOLIDATED STATEMENTS OF CASH FLOWS	6
	NOTES TO CONSOLIDATED FINANCIAL STATEMENTS	7

INDEPENDENT AUDITORS' REPORT

Board of Directors and Stockholders PPL Management, Inc. and Subsidiaries Eau Claire, Wisconsin

We have audited the accompanying consolidated balance sheets of PPL Management, Inc. and Subsidiaries as of December 31, 2008 and 2007, and the related consolidated statements of operations, equity and cash flows for each of the three years in the period ended December 31, 2008. These consolidated financial statements are the responsibility of the Company's management. Our responsibility is to express an opinion on these consolidated financial statements based on our audits.

We conducted our audits in accordance with auditing standards generally accepted in the United States of America. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the consolidated financial statements are free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the consolidated financial statements. An audit also includes assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall consolidated financial statement presentation. We believe that our audits provide a reasonable basis for our opinion.

In our opinion, the consolidated financial statements referred to above present fairly, in all material respects, the financial position of PPL Management, Inc. and Subsidiaries as of December 31, 2008 and 2007, and the results of their operations and their cash flows for each of the three years in the period ended December 31, 2008 in conformity with accounting principles generally accepted in the United States of America.

LarsonAllen LLP

Eau Claire, Wisconsin March 30, 2009, except for Note 15 as to which the date is September 5, 2009

PPL MANAGEMENT, INC. AND SUBSIDIARIES CONSOLIDATED BALANCE SHEETS DECEMBER 31, 2008 AND 2007

	2008	2007
ASSETS		
CURRENT ASSETS		
Cash and Cash Equivalents	\$ 25,777,455	\$16,885,602
Accounts Receivable:		
Current Billings on Contracts	9,503,680	24,546,404
Retainages on Contracts	969,442	5,625,619
Costs and Estimated Earnings in Excess of Billings on Uncompleted Contracts	3,931,497	-
Other Receivables	9,315	-
Prepaid Expenses	3,862,761	1,003,538
Total Current Assets	44,054,150	48,061,163
INVESTMENTS AND OTHER ASSETS		
Equity in Joint Venture	1,732,294	1,524,726
PROPERTY AND EQUIPMENT		
Land	662,525	652,455
Equipment	58,124,298	30,543,617
Vehicles	9,495,424	6,398,072
Office equipment	1,118,248	740,145
Building	1,078,463	980,357
Total	70,478,958	39,314,646
Less Accumulated Depreciation	14,293,511	5,741,317
Net Property and Equipment	56,185,447	33,573,329
Total Assets	\$101,971,891	\$83,159,218

PPL MANAGEMENT, INC. AND SUBSIDIARIES CONSOLIDATED BALANCE SHEETS (CONTINUED) DECEMBER 31, 2008 AND 2007

	2008	2007
LIABILITIES AND EQUITY		
CURRENT LIABILITIES		
Current Maturities of Long-Term Debt	\$ 13,804,888	\$ 6,838,565
Accounts Payable	11,245,763	7,442,513
Billings in Excess of Costs and Estimated Earnings on Uncompleted Contracts	1,324,718	11,711,406
Accrued Expenses:		
Compensation	74,298	537,070
Taxes, Other than Income Taxes	215,364	598,764
Profit Sharing	205,897	106,774
Union Benefits	440	76,668
Workers Compensation and Other	4,693,202	505,171
Total Current Liabilities	31,564,570	27,816,931
LONG-TERM LIABILITIES		
Long-Term Debt (Less Current Maturities)	23,433,946	15,836,524
Total Long-Term Liabilities	23,433,946	15,836,524
Total Liabilities	54,998,516	43,653,455
CONTROLLING INTEREST IN EQUITY		
Members' Equity	42,001,107	36,913,391
NON-CONTROLLING INTEREST IN EQUITY	4,972,268	2,592,372
Total Equity	46,973,375	39,505,763
Total Liabilities and Equity	\$101,971,891	\$83,159,218

PPL MANAGEMENT, INC. AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF OPERATIONS YEARS ENDED DECEMBER 31, 2008, 2007 AND 2006

	2008 AMOUNT	2007 AMOUNT	2006 AMOUNT
CONTRACT REVENUES EARNED	\$ 506,959,422	\$ 302,540,819	\$ 49,669,334
CONTRACT COSTS	415,718,219	264,974,530	42,294,520
CONTRACT GROSS PROFIT	91,241,203	37,566,289	7,374,814
OPERATING EXPENSES	4,611,806	3,072,771	1,351,186
INCOME FROM OPERATIONS	86,629,397	34,493,518	6,023,628
OTHER INCOME (EXPENSE)			
Interest Income	626,078	422,912	9,098
Miscellaneous Income	34,228	16,658	6,145
Gain (Loss) on Sale of Assets	(12,095)	8,356	(31,976)
Interest Expense	(1,464,275)	(1,096,680)	(352,285)
Income Tax	(38,368)	(3,450)	(12,272)
Miscellaneous Expense	(55,794)	-	-
Total Other Expense	(910,226)	(652,204)	(381,290)
NET INCOME BEFORE NON-CONTROLLING INTEREST	85,719,171	33,841,314	5,642,338
Non-controlling Interest in Net Income	2,408,033	1,629,182	125,216
NET INCOME	\$ 83,311,138	\$ 32,212,132	\$ 5,517,122

PPL MANAGEMENT, INC. AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF EQUITY DECEMBER 31, 2008, 2007, AND 2006

	Controlling Interest	Non-controlling Interest	Total Equity
BALANCE, JANUARY 1, 2006	\$ 5,568,613	\$ 113,645	\$ 5,682,258
Net Income	5,517,122	125,216	5,642,338
Distributions	(397,048)	(8,103)	(405,151)
Contributions	-	2,745	2,745
Transfer of Vehicles	(814,143)	814,143	
BALANCE, DECEMBER 31, 2006	9,874,544	1,047,646	10,922,190
Net Income	32,212,132	1,629,182	33,841,314
Distributions	(5,173,285)	(105,577)	(5,278,862)
Contributions		21,121	21,121
BALANCE, DECEMBER 31, 2007	36,913,391	2,592,372	39,505,763
Net Income	83,311,138	2,408,033	85,719,171
Distributions	(77,433,261)	(818,298)	(78,251,559)
Transfer of Vehicles	(790,161)	790,161	
BALANCE, DECEMBER 31, 2008	\$ 42,001,107	\$4,972,268	\$ 46,973,375

PPL MANAGEMENT, INC. AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF CASH FLOWS DECEMBER 31, 2008, 2007, AND 2006

CASH FLOWS FROM OPERATING ACTIVITIES	2008	2007	2006
Net Income	\$ 83,311,138	\$ 32,212,132	\$ 5,517,122
Adjustments to Reconcile Net Income to Net Cash	Ψ 00,011,100	¥ 02,212,102	Ψ 0,011,122
Provided by Operating Activities:			
Non-controlling Interest in Net Income	2,408,033	1,629,182	125,216
Depreciation	8,673,818	4,256,979	1,036,527
(Gain) Loss on Sale of Equipment	12,095	(8,356)	31,976
Equity in Net Income from Joint Venture	(5,957,568)	(10,698,021)	(26,705)
(Increase) Decrease in:	(0,001,000)	(10,030,021)	(20,700)
Contract Accounts Receivable	19,689,586	(18,309,783)	(3,900,502)
Prepaid Expenses	(2,859,223)	(978,284)	48,699
Costs and Estimated Earnings in Excess of Billings on Uncompleted Contracts	(3,931,497)	877,027	(266,397)
Increase (Decrease) in:	(0,551,457)	011,021	(200,331)
Accounts Payable	3,803,250	5,850,964	87,316
Billings in Excess of Costs and Estimated	0,000,200	0,000,004	01,010
Earnings on Uncompleted Contracts	(10,386,688)	11,606,710	(257,423)
Accrued Expenses	3,364,754	499,020	939,733
Income Taxes Payable	-	(11,947)	11,947
Net Cash Provided by Operating Activities	98,127,698	26.925.623	3,347,509
	30,127,030	20,323,023	3,347,303
CASH FLOWS FROM INVESTING ACTIVITIES			
Investment in Joint Venture	-	(50,000)	-
Payments for Purchase of Equipment and Vehicles	(7,738,259)	(6,815,707)	(1,576,223)
Proceeds from Sale of Property Plant and Equipment	88,785	-	-
Distributions from Investment In Joint Venture	5,750,000	9,250,000	
Net Cash Provided (Used) by Investing Activities	(1,899,474)	2,384,293	(1,576,223)
CASH FLOWS FROM FINANCING ACTIVITIES			
Payments on Long-Term Debt	(11,155,755)	(5,981,367)	895,000
Proceeds from Long-Term Debt	2,070,943	-	(1,317,378)
Net Change in Line of Credit	-	(2,000,000)	(500,000)
Contributions - Non-controlling Interest	-	21,121	2,745
Distributions	(78,251,559)	(5,278,862)	(405,151)
Net Cash Used by Financing Activities	(87,336,371)	(13,239,108)	(1,324,784)
NET INCREASE IN CASH AND CASH EQUIVALENTS	8,891,853	16,070,808	446,502
Cash and Cash Equivalents - Beginning of Year	16,885,602	814,794	368,292
CASH AND CASH EQUIVALENTS - END OF YEAR	\$ 25,777,455	\$ 16,885,602	\$ 814,794
SUPPLEMENTAL DISCLOSURES			
Interest Paid	\$ 1,464,275	\$ 1,096,680	\$ 352,285
Non-cash Investing and Financing Transactions:	. ,,	. ,,	
Transfer of equipment from controlling interest to non-controlling interest	790,161	_	814,143
Property and equipment additions acquired with long-term debt	23,648,557	19,194,750	7,889,863

NOTE 1 SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES

Company's Business and Operating Cycle

PPL Management, Inc. is the holding company for Precision Pipeline, LLC (Pipeline). Pipeline services the oil and gas industries which own and operate interstate and intrastate pipelines in the United States as both a general contractor and subcontractor. The Company constructs large and small diameter loop and lateral lines.

The length of the Company's contracts varies, but is typically one to two years. Accordingly, assets to be realized and liabilities to be liquidated within the operating cycle are classified as current assets and liabilities.

Basis of Consolidation

In January 2008, PPL Management, Inc. acquired additional membership units of Precision Pipeline, LLC. After this acquisition PPL owned a controlling interest in Precision Pipeline, LLC (98%). Prior to 2008 PPL Management, Inc. was a variable interest entity of Precision Pipeline, LLC. The remaining two percent ownership is included in non-controlling interest in equity in the accompanying financial statements. The consolidated group included in the accompanying financial statements has not changed. The 2007 and 2006 financial statements have been reclassified to reflect this change in ownership for comparative purposes.

In December 2003, the Financial Accounting Standards Board (FASB) issued revised FASB Interpretation 46, "Consolidation of Variable Interest Entities" (FIN 46R). FIN 46R requires certain variable interest entities (VIEs) to be consolidated by the primary beneficiary of the entity if the investors in the entity do not have the characteristics of a controlling financial interest or do not have sufficient equity at risk for the entity to finance its activities without additional subordinated financial support from other parties.

The consolidated financial statements include the statements of PPL Management, Inc. (PPL), Precision Pipeline, LLC (Pipeline) and two VIEs, Precision Transport, LLC (Transport), and Precision Land Company, LLC (Land), of which Pipeline is the primary beneficiary. All transactions and balances between Pipeline, Transport, Land, and PPL have been eliminated upon consolidation.

Noncontrolling interest represents Precision Transport, LLC, Precision Land Company, LLC and 2% of Precision Pipeline, LLC.

Estimates and Assumptions

The preparation of consolidated financial statements in conformity with U.S. generally accepted accounting principles requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the consolidated financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates.

NOTE 1 SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES (CONTINUED)

Revenue and Cost Recognition

Revenues from fixed-price, modified fixed-price and unit price construction contracts are recognized on the percentage-of-completion method, only after the contract attains a 10% completion stage, measured by the percentage of costs incurred to date to estimated total costs for each contract. This method is used because management considers expended costs to be the best available measure of progress on these contracts.

Revenues from cost-plus-fee contracts are recognized on the basis of costs incurred during the period plus the fee earned, measured by the cost-to-cost method, or ratably over the term of the project, depending upon the terms of the individual contract. Because of inherent uncertainties in estimating costs and revenues, it is at least reasonably possible that the estimates used will change.

Contract costs include all direct material, subcontractors, labor costs, and equipment costs and those indirect costs related to contract performance. Operating expenses are charged to expense as incurred. Provisions for estimated losses on uncompleted contracts are made in the period in which such losses are determined. Changes in job performance, job conditions, and estimated profitability may result in revisions to costs and income and are recognized in the period in which the revenues are determined. Changes in estimated job profitability resulting from job performance, job conditions, contract penalty provisions, claims, change orders, and settlements are accounted for as changes in estimates in the current period. Profit incentives are included in revenues when their realization is reasonably assured. Claims are included in revenues when realization is probable and the amount can be reliably estimated.

The asset, "Costs and estimated earnings in excess of billings on uncompleted contracts," represents revenues recognized in excess of amounts billed. The liability, "Billings in excess of costs and estimated earnings on uncompleted contracts," represents billings in excess of revenues recognized.

Concentrations of Credit Risk

The Company performs credit evaluations of its customers and subcontractors and may require surety bonds. Liens are filed, when permissible, on construction contracts where collection problems are anticipated. As of December 31, 2008 and 2007, accounts receivable are due from gas and utility companies.

The Company's cash balances are maintained in various bank deposit accounts. At various times during the year the balances in those accounts were in excess of those insured by the Federal Deposit Insurance Corporation.

Cash and Cash Equivalents

Cash equivalents are securities held for cash management purposes having maturities of three months or less from date of purchase.

NOTE 1 SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES (CONTINUED)

Contracts Receivable

Contracts receivable from performing construction are based on contracted prices. The Company provides an allowance for doubtful collections which is based upon a review of outstanding receivables, historical collection information and existing economic conditions. Normal contracts receivable are due 30 days after the issuance of the invoice. Contract retentions are due 30 days after completion of the project and acceptance by the owner. Receivables past due more than 120 days are considered delinquent.

Delinquent receivables are written off based on individual credit valuation and specific circumstances of the customer. There are no allowances for delinquent receivables at December 31, 2008 and 2007.

Contract receivables from three customers in 2008 and two customers in 2007, represented approximately 70% and 86%, respectively, of total customer receivables for the years ended December 31, 2008 and 2007, respectively. No other customers represented greater than 10% of the total customer receivables in 2008 and 2007.

Joint Ventures

The Company accounts for joint ventures using the one-line equity method of accounting for equity in the joint venture and proportionate consolidation for revenues and expenses of the joint venture.

Property and Equipment

Property and equipment are carried at cost, less accumulated depreciation. The Company depreciates property and equipment using the straight-line method over the estimated lives of the assets. The estimated useful lives are as follows:

Equipment	3-7 Years
Vehicles	5 Years
Office Equipment	5-7 Years
Building	39 Years

In 2008 and 2006 the Company transferred trailers and other transportation related vehicles from Precision Pipeline, LLC to Precision Transportation, LLC at their carrying cost.

Long-Lived Assets

Long-lived assets to be held and used are tested for recoverability whenever events or changes in circumstances indicate that the related carrying amount may not be recoverable. When required, impairment losses on assets to be held and used are recognized based on the excess of the asset's carrying amount over the fair value of the asset. Certain long-lived assets to be disposed of by sale are reported at the lower of carrying amount or fair value less cost to sell.

NOTE 1 SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES (CONTINUED)

Income Taxes

Pipeline, Transport, and Land are Wisconsin limited liability companies and are taxed under the partnership provisions of the Internal Revenue Code and comparable state regulations. As such, the companies do not pay federal or state income taxes on their taxable income. Instead, the members report on their personal income tax returns, the companies' taxable income and tax credits.

The PPL Management has elected to be taxed as an S Corporation. Under provisions of the Internal Revenue Code and similar provisions of Wisconsin law, the Company does not pay federal or state corporate income taxes on its taxable income.

In lieu of corporation income taxes, the stockholders include their respective shares of the Company's taxable income and tax credits in their individual income tax returns.

The companies have committed to making distributions to the members and shareholders to offset income taxes incurred as a result of the pass through income.

The Company has elected to defer application of FASB Interpretation No. 48 "Accounting for Uncertainty in Income Taxes." The Company follows Financial Accounting Standard No. 5 "Accounting for Contingencies" for evaluating uncertain tax positions.

Reclassifications

Certain prior year amounts have been reclassified for comparative purposes to conform with the current year presentation.

NOTE 2 CONTRACT CONCENTRATIONS

Contract revenues from four contracts in 2008, three contracts in 2007, and three contracts in 2006 represented approximately 76%, 68%, and 40%, respectively, of total contract revenues for the years ended December 31, 2008, 2007 and 2006, respectively. No other contracts represented greater than 10% of the total contract revenues in 2008, 2007, and 2006. Accounts receivable from these contracts at December 31, 2008 and 2007 were \$1,149,475 and \$26,015,017, respectively.

NOTE 3 COSTS, ESTIMATED EARNINGS AND BILLINGS ON CONTRACTS IN PROCESS

	2008	2007
Costs Incurred on Uncompleted Projects	\$644,768,782	\$398,352,951
Estimated Gross Profit	79,083,551	33,293,453
Contract Revenues Earned	723,852,333	431,646,404
Less: Billings to Date	721,245,554	443,357,810
Total	\$ 2,606,779	\$ (11,711,406)
1000	Ψ 2,000,113	Ψ (11,711,400)

NOTE 3 COSTS, ESTIMATED EARNINGS AND BILLINGS ON CONTRACTS IN PROCESS (CONTINUED)

Reported in the accompanying consolidated balance sheets as follows:

	2008	2007
Costs and Estimated Earnings in Excess of Billings on Uncompleted Contracts	\$ 3,931,497	\$ -
Billings in Excess of Costs and Estimated Earnings on Uncompleted Contracts	(1,324,718)	(11,711,406)
Total	\$ 2,606,779	\$(11,711,406)

NOTE 4 JOINT VENTURE

On August 15, 2006, Precision Pipeline, LLC entered into a 25% interest joint venture with U.S. Pipeline, Inc., Michels Corporation, and Welded Construction, L.P. The joint venture is recorded on the equity basis and at December 31, 2008 and 2007. The following is a summary of the balance sheet, and statement of operations for the years ended December 31, 2008 and 2007:

	2008		2007	
	Joint Venture Total	Company's Share of Joint Venture	Joint Venture Total	Company's Share of Joint Venture
Total Estimated Contract	\$1,677,404,000	\$419,351,000	\$938,000,000	\$234,500,000
Total Estimated Gross Profit	\$ 85,364,000	\$ 21,341,000	\$ 50,000,000	\$ 12,500,000
Revenue	\$ 721,599,928	\$180,399,982	\$749,695,877	\$187,423,969
Expenses	697,769,654	174,442,415	706,903,793	176,725,948
Net Income	\$ 23,830,274	\$ 5,957,568	\$ 42,792,084	\$ 10,698,021
Current Assets Equipment	\$ 37,122,050	\$ 9,280,513	\$ 24,722,995	\$ 6,180,750
Total Assets	\$ 37,122,050	\$ 9,280,513	\$ 24,722,995	\$ 6,180,750
Current Liabilities	\$ 30,192,876	\$ 7,548,219	\$ 18,624,093	\$ 4,656,024
Long-Term Liabilities				
Total Liabilities	30,192,876	7,548,219	18,624,093	4,656,024
Equity in Joint Venture	6,929,174	1,732,294	6,098,902	1,524,726
Total Liabilities and Partners' Equity	\$ 37,122,050	\$ 9,280,513	\$ 24,722,995	\$ 6,180,750

There was limited activity for the joint venture in 2006.

The joint venture is accounted for using the one-line equity method on the accompanying consolidated balance sheets and the proportionate consolidation method on the accompanying consolidated statements of operations. At December 31, 2008 and 2007, Precision Pipeline, LLC had receivables from this joint venture of approximately \$1,878,000 and \$12,849,000, respectively.

NOTE 5 REVOLVING AND NON-REVOLVING CREDIT LINES

The Company has a line of credit with a bank at prime plus .5% (3.75% at December 31, 2008). Total credit facility of \$12,000,000 requiring monthly installments of interest only, due May 15, 2009, secured by a general business security agreement and guarantees of the members. There were no amounts outstanding on this line at December 31, 2008 and 2007.

The Company's revolving credit lines contain certain restrictive covenants which include maintaining certain financial ratios, restrictions on mergers and acquisitions, sales of receivables, conduct of business, sale and leaseback transactions, indebtedness and liens, guarantees of others' debt, and loans to other entities. As of December 31, 2008 and for the period then ended, the Company was in violation of one of these loan covenants. The bank has subsequently waived the violation and has amended the credit agreement for this violation.

NOTE 6 LONG-TERM NOTES PAYABLE

Payable to:	2008	2007
Line of credit with a bank. Total credit facility of \$4,000,000 requiring monthly installments of \$105,349, including interest at 5.70%, due August 2011, secured by a general business security agreement.	\$1,678,758	\$ -
Note payable to a bank in monthly installments of \$22,066, including interest at 6.99%, due December 31, 2011, secured by vehicles.	714,152	920,171
Note payable to a bank in monthly installments of \$33,552, including interest at 7.25%, due April 27, 2010, secured by vehicles.	510,236	861,906
Note payable to a bank in monthly installments of \$72,698, including interest at 5.50%, due June 15, 2011, secured by vehicles.	2,025,856	-
Note payable to a bank in monthly installments of \$2,817 plus interest at LIBOR + 1.85 (2.93% at December 31, 2008), due March 10, 2022, secured by mortgage and guarantees of members	1,232,275	1,263,120
7.00% note payable to a bank in monthly installments of \$4,753, including interest, due October, 2010, secured by vehicles.	1,082,795	1,140,733

NOTE 6 LONG-TERM NOTES PAYABLE (CONTINUED)

	2008	2007
Prime minus .25% (3.00% at December 31, 2008) Equipment line of credit with a bank. Total credit facility of \$500,000 requiring monthly installments of \$15,692, including interest, secured by a general business security agreement and guarantees of the members. Paid in full during 2008.	-	292,244
Notes payable to financing companies due in monthly installments ranging from \$3,085 to \$24,393, including interest ranging from 0% to 7.53%, due dates ranging from January 2009 to September 2012, secured by equipment and vehicles.	1,430,281	1,390,179
Notes payable to banks in monthly installments ranging from \$937 to \$14,228, including interest ranging from 5.65% to 7.50%, due dates ranging from January 2009 to May 2010, secured by equipment, vehicles, and a general business security agreement and guarantees of the members. All notes paid in full by December 31, 2008.	-	437,336
Capitalized lease obligations, at imputed interest rates ranging from 0% to 9.94%, monthly payments totaling \$614,406, secured by leased assets.	28,564,481	16,369,400
Total	37,238,834	22,675,089
Less: Current Portion	13,804,888	6,838,565
Long-Term Portion	\$23,433,946	\$15,836,524

Maturity requirements on long-term debt as of December 31, 2008 are as follows:

Year Ending December 31,	Amount
2009	\$13,804,888
2010	11,708,137
2011	4,974,345
2012	1,350,335
2013	4,363,487
Thereafter	1,037,642
Total	\$37,238,834

NOTE 7 LEASE AGREEMENTS

Office and shop facilities are leased under operating lease agreements. The office lease was an annual lease that expired June 30, 2007. The shop lease was a month-to-month rental agreement. Rent expense for these operating leases for the periods ended December 31, 2008, 2007, and 2006, was \$7,610, \$33,737, and \$56,449, respectively.

NOTE 7 LEASE AGREEMENTS (CONTINUED)

The Company rents various pieces of construction equipment under month-to-month and long-term lease agreements. The long-term equipment leases are operating and capital leases which expire in various years through 2013. In addition, the Company is required to pay maintenance and insurance costs. Rental payments on all the equipment operating leases amounted to approximately \$3,070,000, \$2,601,000, and \$496,000 for the years ended December 31, 2008, 2007, and 2006, respectively.

Capitalized leased assets consist of:

	2008	2007
Equipment and Vehicle	\$38,962,875	\$19,250,378
Less - Accumulated depreciation	6,417,147	1,897,208
Total	\$32,545,728	\$17,353,170

Minimum lease payments for capital and operating leases in future years are as follows:

	Capital Leases	Operating Leases
Year Ending December 31,		
2009	\$11,439,864	\$2,546,492
2010	10,055,544	515,320
2011	3,824,284	-
2012	1,413,777	-
2013	4,406,321	-
Total Minimum Lease Payments	\$31,139,790	\$3,061,812
Less: Interest	(2,575,309)	
Present Value of Minimum Lease Payments	\$28,564,481	

NOTE 8 QUALIFIED RETIREMENT PLAN

The Company has adopted a qualified profit sharing plan. Employee eligibility is determined by age, years of service, and the number of hours worked. Contributions to the plan are at the discretion of the board of directors, but may not exceed 25% of total eligible compensation. The Company matches employees' voluntary contributions up to 4% of compensation. The Company's contributions to the plan were \$272,848, \$150,619, and \$67,711 for the years ended December 31, 2008, 2007, and 2006, respectively. Employees are permitted to make voluntary contributions up to the maximum amount allowed by the Internal Revenue Code.

NOTE 9 BUY-SELL AGREEMENT

The members and the Company have a buy-sell agreement. In the event of a member's death, disability, or termination, the remaining members have the option to redeem the applicable shares of common stock at a price determined under the terms of the agreement.

NOTE 10 LABOR FORCE

100% of the Company's contract labor personnel are covered by collective bargaining agreements. The agreements are scheduled to expire within two years.

The Company makes contributions to various multi-employer union pension plans. The Plans cover all of the Company's union employees. The contributions are determined in accordance with the provisions of the negotiated labor contracts based on the aggregate number of hours worked. Information as to the Company's portion of the accumulated plan benefits, plan net assets and unfunded vested benefits is not determinable. In the event of withdrawal from the Plan(s), the Company may be subject to payment of a withdrawal liability. Management does not intend to take action which would subject it to such liability.

Contributions to the various unions for the various union fringe benefits, including pension, was approximately \$46,781,000, \$25,764,000, and \$4,044,382 for the years ended December 31, 2008, 2007, and 2006, respectively.

NOTE 11 VARIABLE INTEREST ENTITIES

In December 2003, the FASB issued revised FASB Interpretation 46 (FIN 46R), "Consolidation of Variable Interest Entities." FIN 46R requires that a company that holds a variable interest in an entity consolidate the entity if the company's interest in the variable interest entity (VIE) is such that the company will absorb a majority of the VIE's expected losses and/or receive a majority of the VIE's expected residual returns, if they occur. In such cases, the company is the primary beneficiary of the VIE. FIN 46 also requires additional disclosures about primary beneficiaries and other significant variable interest holders. As a result, the entities described below have been included in these financial statements.

Precision Transport, LLC is a transportation company that provides transportation services to Precision Pipeline, LLC. Precision Land Company, LLC was created to own the land and building that Precision Pipeline is using for its office and shop facilities. These two entities began operations in 2006. Precision Pipeline has guaranteed all of the financing of these entities and has the primary risk of loss related to these notes. Therefore, Precision Pipeline, LLC is considered the primary beneficiary of these entities.

NOTE 11 VARIABLE INTEREST ENTITIES (CONTINUED)

The financial information of Precision Transport Company, LLC, and Precision Land Company, LLC for the years ended December 31, 2008 and 2007 is summarized below:

	2008			
			Precision Land	
		Company, LLC	Co	ompany, LLC
Assets	\$	6,435,099	\$	1,376,926
Liabilities		3,250,244		1,232,275
Members' Equity	\$	3,184,855	\$	144,651
	_		_	
Revenues	\$	29,999,100	\$	330,929
Expenses		29,399,893		224,043
Net Income	\$	599,207	\$	106,886
			_	
		2007		
		ecision Transport		ecision Land
		Company, LLC	Co	ompany, LLC
Assets	\$	3,583,903	\$	1,300,886
Liabilities		1,782,078		1,263,121
Members' Equity	\$	1,801,825	\$	37,765
	_		_	
Revenues	\$	14,040,046	\$	175,753
Expenses		13,082,367		161,854
Net Income	\$	957,679	\$	13,899

NOTE 12 COMMITMENTS AND CONTINGENCIES

The Company is currently party to various claims and legal proceedings arising from the normal course of business. Although the outcome and the eventual liability of the Company, if any, in these matters cannot be presently determined, it is the opinion of management that the resolution of those claims not covered by insurance will not have a material adverse effect on the financial condition of the Company.

The Company has initiated a claim against a customer for approximately \$14,500,000. Management has elected to defer recognition of the revenue on this claim until it has been settled.

NOTE 13 FAIR VALUES OF FINANCIAL INSTRUMENTS

The fair values of financial instruments at December 31, 2008 and 2007 were as follows:

Cash and cash equivalents -

The carrying amount reported in the consolidated balance sheets approximates fair value based on current interest rates.

Long-term debt -

The Company's long-term debt is fixed and variable rate loans that reprice within three years; therefore, the carrying value of the Company's long-term debt approximates fair value based on current incremental borrowing rates available for similar arrangements.

NOTE 14 RELATED PARTY TRANSACTIONS

The Company had transactions with three companies that are owned or controlled by relatives of the shareholders. The following table summarizes transactions for the years ended December 31, 2008, 2007, and 2006:

	2008	2007	2006
Income from Equipment Rentals	\$ 661,833	\$ 386,510	\$ -
Purchases of subcontractors, materials, and supplies	15,238,271	12,496,263	408,774

The transactions with the related parties include the following accounts receivable and accounts payable as of December 31, 2008 and 2007:

	2008	2007
Accounts Receivable	\$517,231	\$338,196
Accounts Payable	925,025	955,420

NOTE 15 SUBSEQUENT EVENTS

The statement of operations for the year ended December 31, 2008 includes approximately \$5,642,000 from the resolution of a claim outstanding on a construction contract at December 31, 2007.

Subsequent to year end the Company resolved claims outstanding on two construction contracts at December 31, 2008. The Company will recognize revenue of \$12,138,000 in 2009 from the settlement of these claims.

NOTE 15 SUBSEQUENT EVENTS (CONTINUED)

Subsequent to year end the Company received the final approval and payment of an \$11,574,000 contract incentive from 2008. The incentive payment was contingent on the results of tests performed in 2009. The incentive payment will be recognized as revenue in 2009.

In September 2009, the Company entered into an exclusive negotiation agreement with an unrelated third party for the purchase of the membership interests of Precision Pipeline, LLC and Precision Transport, LLC.